CORRELATES OF HIV PERCEIVED RISKS AND PROTECTIVE STRATEGIES AMONG ADOLESCENTS IN RURAL MALAWI.

BY

CHRISTOPHER MANYAMBA

Submitted in Partial Fulfillment of the Requirements for the Degree of

Masters of Population Studies,

Faculty of Humanities, Development and Social Sciences,

University of KwaZulu-Natal

Durban, South Africa

July 2011

Abstract

The study examined correlates of HIV perceived risks and protective strategies among adolescents in rural Malawi i.e. whether there was an association between socio demographic characteristics of adolescents and subsequent HIV perceived risks and protective strategies. Age at sexual debut is a variable of interest in this study. The objectives were to investigate whether age at first sexual intercourse does influence adolescents' individual's risk perceptions of HIV and protective strategies, i.e. does age at first sex set a precedence in the thinking of adolescentswith regards HIV risks and protective strategies such as condom use.

Data: This study uses cross sectional data from Wave 3 of a wider study, the Malawi Diffusion and Ideational Change Project (MDICP).The MDICP is a joint collaboration between researchers from the Population Studies Centre at the University of Pennsylvania and the University of Malawi's College of Medicine and Chancellor College. In this study, a multiple regression analysis on the cross sectional data from the 2004 wave 3of the Malawi Diffusion and Ideational Change Project was done. The focus of this study is on the relationship between age at first sexual intercourse and the individual's perception of risk of HIV/AIDS and protective strategies, it examines if there is an association between the two factors.

Descriptive analysis: Descriptive statistics to obtain the frequencies of the main variables of interest such as the background characteristics of respondents (gender, marital status, mean age at sexual debut, education, economic status of household) were computed in order to better understand the population under study.

Multivariate Analysis: Three separate models were employed; two multinomial regression analysis measuring worry and likelihood of infection against socio demographic variables and a binary logistic regression measuring condom use and the socio demographic variables. Relative Risk Ratios were used to interpret the multinomial logistic regression output while odds ratios were used to interpret the logistic output.

Results: Consistent with existing literature, results indicate that early age at first sex is a predictor of HIV risk perceptions and protective strategies. Age at first sex does set a precedence in the future thinking of adolescents with regards to how they perceive HIV risks, as well as how they perceive protective strategies against HIV-particularly condom use. This

study confirms that age at first sex alongside with other socio demographic variables are significantly associated with HIV risk perceptions HIV and condom use, while other factors may matter, such as gender, region, HIV knowledge, religion or educational level.

Limitations: The information on age at first sexual encounter was based exclusively on respondent's self-reports. This data collection method often has limitations that are attributed to the tendency for people to under-report socially unacceptable behaviours (e.g. having multiple sex partners) and to over-report socially acceptable behavior. The other limitation is that the study lacks longitudinal data to allow a more in-depth analysis oftrends in HIV/AIDS-related indicators.

(Words 475)

Declaration

This dissertation represents my original work and has not been submitted in any other form to another university. Work of others where use and referenced in the text has been duly acknowledged. The research for this dissertation was performed in the School of Development Studies in the Faculty of Humanities at the University of KwaZulu-Natal, Durban. Research was undertaken under the supervision of Professor Pranitha Maharaj and Nompumelo Nzimande during the period from July 2009 to July 2011.

Signed,

Student Signature Date

Acknowledgements

I would like to express my gratitude to Professor Pranitha Maharaj and Nompumelelo Nzimande of UKZN for their continued support in my dissertation, especially in the proposal development, analysis, and write up. With their caring, and professional guidance, I feel I have been introduced to the analysis and authoring world. This experience broadened my scope and allowed me to have an insight of the academic world and the sense of scholarship, which I have always admired.

Thanks to Veronica (my beloved wife), my children Taurai, Mphatso and Tina for the psychological support they gave me in my academic carrier at UKZN. Above all I thank the Almighty God for taking me through.

List of Acronyms

AIDS	Acquired Immune Deficiency Syndrome
HIV	Human Immunodeficiency Virus
HMIS	Health Management Information Systems
MDG	Millennium Development Goals
MDICP	Malawi Diffusion and Ideational Change Project
MICS	Multiple Indicator Cluster Survey
MDHS	Malawi Demographic and Health Survey
MICS	Multiple Indicator Cluster Survey
STI	Sexually Transmitted Disease
UNICEF	United Nations Children's Fund
UNFPA	United Nations Population Fund
UNGASS	United Nations General Assembly Special Session on HIV/AIDS
WHO	World Health Organization

List of Figures

Figure 1: Theoretical Framework	17
Figure 2: Showing the map of Malawi and the Study Area	52
Figure 3: Reported Age at First Sex, Men and Women	81

List of Tables

Table 4.1:	Distribution of Selected Social and Demographic Characteristics	.78
Table 4.2:	Percentage of Ownership of Household Assets.	78
Table 4.3:	Showing Contraceptive Use by Gender	79
Table 4.4:	Indicating Selected Sexual Characteristics of the Respondents.	.80
Table 4.5:	Risk Perceptions and Worry Levels among Males and Females	.84
Table 4.6:	Showing HIV/AIDS Knowledge and Perceptions	.87
Table 4.7:	HIV Preventive Strategies among Males and Females	.91
Table 4.8	Outcome Variable: Categories and Codes	.94
Table 4.9:	Predictors: Categories and Codes	.95
Table 4.10: Re Characteristics	elative Risk Ratios Showing the Association between Socio Demographic s and Perceived Likelihood of Infection	.96
Table 4.11: RInfection and i	elative Risk Ratios Showing the Association between Worry about HIV/AII	DS 103
Table 4.2: Od and its Assoc	ds Ratios Showing the Association between Condom Use at Last Sex iated	
Correlates		108

Table of Contents

Acknowl	edgements4
List of A	cronyms5
List of Fi	gures6
List of T	ables6
CHAPTH	ER 1
1 INT	RODUCTION AND BACKGROUND10
1.1	HIV/AIDS in Sub Saharan Africa
1.2	Definitions of Adolescents11
1.3	HIV Risk Perceptions
1.4	Protective Strategies-Condom Use
1.5	Age at First Sex
1.6	The Importance and Motivation for the Study13
1.7	Aims of the Study
1.8	Research Questions
1.9	Theoretical Framework
1.9.	1 The Social Cognitive Model16
1.10	Organization of the Dissertation
2 CH	APTER 2
LITERA	TURE REVIEW20
2.1	Introduction
2.2	HIV Risk Perceptions: Worry and Likelihood of Infection
2.3	Misconceptions about HIV
2.4	HIV Protective Strategies
2.4.	Condom Use28
2.4.	2 Abstinence and Delaying Sex
2.4.	3 Marriage
2.5	Correlates of Risk Perceptions and Condom Use
2.5.	Respondent's Age
2.5.	2 Education
2.5.	3 Religion33
2.5.	4 Geographical Area/Residence33
2.5.	5 Household Wealth
2.5.	5 Marital Status

	2.5.7	7	Gender Roles	.35
	2.5.8	3	HIV Counseling and Testing	.36
	2.5.9)	Age at First Sex	.37
	2.5.1	0	Summary	.38
2.	6	Fact	ors Associated with Early Sexual Debut	. 39
	2.6.1	l	Coercion	.40
	2.6.2	2	Socio-economic status	.41
	2.6.3	3	Peer Pressure	.43
	2.6.4	ł	Family Structure	.44
	2.6.5	5	Cultural Practices	.45
	2.6.6	5	Gender Disparities	.47
	2.6.7	7	Decline of Age at Menarche	.48
2.	7	Psyc	chological Factors: Personal ties to people with HIV/AIDS	.49
3	CHA	APTE	ER 3	.51
MET	гноі	DOL	0GY	.51
3.	1	Stud	ly Area	.51
3.	2	Data	a Source and Sample	.53
	3.2.1	l	Sample Size Calculation.	.53
	3.2.2	2	Fieldwork and Data	.54
	3.2.3	3	Methods of analysis	.55
3.	3	Rese	earch Questions	.56
3.	4	Defi	nition of Variables	.56
	3.4.1	l	Outcome Variables	.57
	3.4.2	2	Covariate Factors	.58
3.	5	Met	hods of Analysis	.63
	3.5.1	l	Descriptive analysis	.64
	3.5.2	2	Multivariate Analysis	. 64
	3.5.3	3	Multinomial logistic regression	.64
	3.5.4	ł	Limitations	.68
4	CHA	APTE	ER 4	.70
ANA	ANALYSIS AND FINDINGS70			
4.	1	Intro	oduction	.70
	4.1.1	l	Socio-Demographic Characteristics	.70
	4.1.2	2	Relative Household Wealth	.74

4.1.3	3 Family Planning	75	
4.2	Sexual Characteristics	75	
4.2.1	1 Perceived Risk for HIV	78	
4.2.2	2 HIV Related Knowledge and HIV Testing	81	
4.2.3	3 HIV Prevention and Condom Use	84	
4.3	Multivariate Analysis	87	
4.3.1	1 Outcome Variables	87	
4.3.2	2 Correlates of Risk Perception and Condom Use	88	
4.4	Model 1: The Association between Likelihood of Infection and socio-demographic		
charact	teristics	89	
4.4.1	1 Correlates of No Likelihood Perception	90	
4.4.2	2 Correlates of Low Likelihood Perception	92	
4.4.3	3 Correlates of Medium Likelihood Perception	93	
4.5	Model 2: The Association between Worry about HIV/AIDS Infection and its Correlated	tes96	
4.5.1	1 Correlates of 'Not Worried at All'	98	
4.5.2	2 Worried at a Little	99	
4.6	Model 3: Association between Condom Use at Last Sex and its Correlates	101	
5 CHA	APTER 5	105	
DISCUSS	SION	105	
5.1	Correlates of Risk Perceptions and Protective Strategies	105	
CHAPTE	ER 6	111	
6 CON	NCLUSION	111	
6.1	6.1 Conclusions and Recommendations11		
6.2	Recommendations and Policy Implication	114	
7 Refe	erences	117	

CHAPTER 1

INTRODUCTION AND BACKGROUND

1.1 HIV/AIDS in Sub Saharan Africa

The social, economic, biological, psychological and cultural consequences of HIV/AIDS pandemic in the Sub Saharan African region are growing in size as well as complexity (Yeatman, 2009), and yet the effects of sexual debut on risk perceptions and protective strategies are often overlooked. The purpose of this study is to determine if age at first sex determines HIV risk perceptions and protective strategies against HIV among young people in Malawi. Many young people in Sub-Saharan Africa face the risk of contracting HIV, sexually transmitted infections (STIs) and unintended pregnancy. According to the Multiple Indicator Cluster Survey, one in three women (34.5 percent) aged 15-19 in Malawi had begun childbearing, 7.5 percent werepregnant with a first child while 27 percent had had a live birth (MICS,2006). Furthermore, the Malawi Demographic and Health Survey results indicate that 24 percent of men and 12 percent of women had a history of STIs (MDHS, 2006). Globally an estimated 33 million individuals were living with HIV/AIDS at the level in 2007, with about two-thirds of them in Sub-Saharan Africa (MDG Report, 2009).

Sub-Saharan Africa still remains the region with the highest percentage of people living with HIV (63 percent), and in 2005 an estimated 4.3 percent of 15-24 year old females and 1.5% of males in the region were HIV-positive (Biddlecom *et al.*, 2007). In Malawi, HIV prevalence among pregnant women aged 15-24 years was 14.3 percent in 2005 and 12.3 percent in 2007 (Malawi HIV and Aids Monitoring and Evaluation Report, 2007). The results of a national survey conducted in Malawi indicate that the HIV prevalence rate among adolescents aged 15-24 was 7.4 percent, with female prevalence rate twice as much (10.6 percent) as that of the males (4.1 percent) of the same age group (MDHS, 2010). The DHS Comparative Report (2009) indicates that while comprehensiveknowledge about HIV/AIDS has increased in most countries, especially in West Africa, levels of such knowledge remain unacceptably low (Sneeringer and Stacy, 2009).

Considerable proportions of men and women still lack knowledge of specific prevention methods such as abstinence, being faithful to one sexual partner, having one trusted partner, and condom use. Similarly, knowledge about how HIV is transmitted and how it can be prevented is almost universal in Malawi and yet there is a high level of sexual activity among young people (Munthali *et al.*, 2006). The long term goal for the Health Sector in Malawi is; 'To improve health status of people at all levels in a sustainable manner'. The goal is underlined with the overall policy statement which is to raise the level of health status of all Malawians by reducing the incidence of illness and occurrence of premature deaths in the population (Malawi Government HMIS Assessment Report 2009). In order to achieve this national goal there is need to reduce the HIV prevalence, especially among the adolescents. This calls for HIV intervention programmes to take into account the age at first sex, and its causal effect on the HIV risk perceptions and protective strategies among young people.

1.2 Definitions of Adolescents

Adolescents are defined in various ways. Most authors define adolescence as a period of transition from childhood to adulthood characterized by biological, cognitive, psychological and social changes (Crawford, 2003; Hoosain, 2003; cited in Munthali &Chimbiri, 2003). With regards to age, theWorld Health Organization (WHO), United Nations Population Fund (UNFPA) and UNICEF define adolescents as 'young people aged 10-19 years undergoing a period of transition in which children attain adulthood' (Crawford, 2000 cited in Munthali & Chimbiri, 2003). The Convention on the Rights of the Child defines 'children' as those aged 0-18 years, while UNFPA, WHO and UNICEF define 'youths'as those aged 15-24 years and young 'people' as those aged 10-24 years.

For the purposes of this study adolescents will encompass the youngest (years) to 24 years, and sexual intercourse is defined as the insertion of the penis into the vagina. For the purposes of this study, unless specifically stated, 'young people' and 'adolescents' refers to people between ages 10-24 and will be used interchangeably.

1.3 HIV Risk Perceptions

A number of models used to explain health behaviour state that high HIV perceived risk is associated with low levels of risk-taking behaviour (Anderson *et al.*, 2007). Clark (2004) argues that perceptions of risks, as well as accurate knowledge of ways to avoid risks, play a crucial role in HIV/AIDS prevention. Anderson *et al.*, (2007) further argues that there is a reciprocal relationship between HIV risk perception and sexual initiation; that is, sexually

inexperienced youths who perceive themselves to be at high risk will be more likely than those who perceive themselves to be at lower risk to delay first sexual intercourse, and individuals who have had sex will perceive themselves at higher risk than those who are sexually inexperienced. In addition, Kaestle*et al.*, (2005) argue that early initiation of sexual intercourse among adolescents has been linked to increased risk of sexually transmitted infections (STIs) and pregnancy during adolescence since those who initiate sex at early ages are more likely to engage in riskier sexual behaviours. Munthali *et al.*, (2006) argue that self-perceived risk of HIV can motivate adolescents to change their risky behaviours and that older adolescents might perceive themselves to have a greater chance of getting HIV because they are engaged in more risky behaviours. They further state that about 16 percent of 15–19-year old women had given birth, one-third did not want their last birth at all, and more than one-quarter wanted the birth at a later time, and about half of all adolescents were worried about getting HIV/AIDS, while about half of adolescents were worried about getting pregnant.

Empirical evidence from the DHS Comparative Report (2009) indicates that the percentage of respondents who reported themselves at high risk of contracting HIV or who already had HIV varied considerably across the countries studied (Sneering and Stacy, 2009). However, lack of published research on how those who initiate early sex are likely to perceive themselves at risk of HIV infection makes it almost difficult to integrate adolescents' HIV and reproductive health concerns in preventive programmes. This is why this study aims to investigate and establish if age at first sexual encounter among young people in Malawi does in future influence their HIV risk perceptions and protectivebehaviour.

1.4 Protective Strategies-Condom Use

Young women in developing countries are less likely than young men to use condoms during sex that poses special risks. Condom use is also much less common among young people in poorer households and among those living in rural areas (United Nations, 2011). Due to the problems in reporting condom use among young people, age at last sex was used as a measure of HIV prevention among the adolescents.

1.5 Age at First Sex

Age at first sex is the variable of interest in this study. Research evidence states that the age at which women initiate sexual intercourse marks the beginning of their exposure to reproductive risk (MDHS 2006). Research throughout Africa has highlighted the fact that the first sexual experiences are taking place in very different social settings compared to the previous generations, and that the driving forces behind this are the increase in factors that make traditional values weak, particularly in reducing the importance of virginity at marriage (Gueye *et al.*, 2001). Other researchers also agree that the timing of first intercourse is an important indicator of the onset and duration of exposure to the risk for both unplanned pregnancy and they state that those adolescents who delay sexual activity spend fewer years of their lives at risk of unplanned pregnancy and HIV/AIDS (Munthali *et al.*, 2004).

Moore et al, (2007) argue that 40 percent of new cases of HIV infection occurred worldwide among young people between the ages of 15 and 24 years in 2006, with Sub-Saharan Africa being the worst-affected region accounting for two-thirds of the HIV-positive population of which 59 percent were female. Results from a longitudinal survey of the MDICP indicate that women who had an early sexual debut (that is, by age 15) had a higher HIV prevalence (11.1 percent) than women who had a late sexual debut (7.6 percent) (Boileau *et al.*, 2009). Clark (2004) challenges on rethinking about the associations of first sex and HIV risk perception and protective strategies: this paper endeavours to do so by examining if sexual debut sets precedence on adolescents' thinking on HIV as they grow up.

1.6 The Importance and Motivation for the Study

Young people in Sub Saharan Africa are at high risk of HIV infection. Women and young people are especially vulnerable. Globally, nearly 23 per cent of all people living with HIV are under the age of 25. And young people (aged 15 to 24) account for 41 per cent of new infections among those aged 15 or older. In 2009, women represented a slight majority (about 51 per cent) of peopleliving with HIV. Because first sexual experiences may influence a young person's HIV risk and protective behaviour a better understanding of coital debut is needed. Pettifor *et al.*, (2009) states that early coital debut is associated with factors that may increase a young person's risk for HIV infection, such as forced sex and having older partners.

The relationship between perception of risk and sexual behavior is complex and poorly understood. Studies conducted in different cultures have associated HIV risk perception with a wide range of variables: number of sexual partners, knowledge of sexual partners' past sexual behavior, fear of AIDS, shame associated with having AIDS, community perception of AIDS risk, knowing someone with AIDS, discussing AIDS at home, closeness of parent-child relationships and religious affiliation (Prata et al., 2006). Studies that have examined the association between the perception of risk and sexual behavior remain inconclusive because of the difficulty of disentangling the complex relationship between the two variables (Cleland, 1995 cited in Akwaraet al., 2003). This means that continuous research is needed inorder to establish the critical factors influencing risk perception and protective strategies among adolescents. This is why this study also focuses on the relationship between socio demographic characteristics and the individual's perception of risk of HIV/AIDS and protective strategies with a particular emphasis on age at first sexual intercourse. Much as the direction of causal effect would have been more useful in making informed decisions about adolescent sexual reproductive behaviours in relation to HIV protective strategies, the study was confined to a cross sectional data that only allowed analyses examining association. Such information adds to the existing knowledge about adolescents' HIV perceptions and is very crucial in the design of young people's HIV intervention programmes if the HIV prevalence rates are to be reduced.

Furthermore, individuals' knowledge of HIV transmission and accurate assessment of their own risk are among the key factors in adoption of safer sexual practices such as condom use. Prata *et al.*, (2006) argues that adolescent behavior is of special interest for the numberof life-years saved is greatest when infections are averted in relatively young individualsand that preventing HIV infection in women of childbearing age prevents transmission from mother child. Furthermore, it may be easier to change sexual attitudes, practices and risky behaviors among the young than among older people (Prata *et al.*, 2006). Information from this study will add to the existing information which policymakers use in order to design effective policies in the fight against HIV and AIDS.

The hypothesis in this study is that the earlier the age at which first intercourse occurs, the less likely adolescents are to perceive themselves at risk of HIV infection, and the more likely they are to engage in unsafe sex. Furthermore, self-perceived risk of HIV can serve as a motivation for adolescents to change their behaviors that place them at risk of HIV (Munthali *et al.*, 2006). Since early sexual experiences inform future sexual behaviours, it is very crucial

to understand the sexual and reproductive behaviours of young people as well as the factors that protect or put them at risk of HIV infection, STIs and unwanted pregnancy (Moore *et al.*, 2006). Gausset, (2001) cited in Clark, Poulin and Kohler, (2009) reiterate the same; 'that broader social norms governing the acceptability of sexual behaviours like age of sexual debut and number of premarital and extramarital sexual partners are likely to play a larger role in determining HIV risks'. A number of studies cited in this paper have investigated the link between socio demographic characteristics including and the risk of contracting HIV but have not included the relationship between the age at first sexual intercourse and HIV risk perceptions among the young people. Munthali *et al.*, (2006) arguethat understanding themajor overall concerns of young people provides a useful perspective on how sensitive young people are likely to be to HIV as well as sexual and reproductive health information and related interventions.

Having stated the above, the aim of this paper is to examine the relationship between socio demographic characteristics including age at first sexual intercourse and perceived HIV risks perceptions and protective behaviour among young people in Malawi. The study examines whether a relationship exists between age at sexual debut and subsequent HIV perceived risks and protection strategies among adolescents. Multiple regression analyses on the cross sectional data from the 2004 Malawi Diffusion and Ideational Change Project adolescents sample who had reported engaging in sexual intercourse were run against the HIV risk perceptions and protection strategies so as to establish if there was a significant association. By examining age at sexual debut, alongside socio demographic characteristics and outcomes into the assessment of adolescent HIV risk perceptions and protective sexual behaviour, this study will contribute an important element of existing literature, as well as provide insights to HIV programme interventions targeting adolescents.

1.7 Aims of the Study

The objectives of the study are;

- i. To investigate whether age at first sexual intercourse does influence adolescents' individual perceived risk perceptions of HIV.
- ii. To investigate whether age at first sexual intercourse does influence adolescents' individual perceived protective strategies.

1.8 Research Questions

The research questions of this study are as follows;

- a. Is there an association between perceived likelihood of infection (perceived risk) and socioeconomic factors (age at first sex being the variable of interest).
- b. Is there an association between worry over contracting HIV (perceived risk) and socio economic characteristics (age at first sex being the variable of interest).
- c. Is there an association between condom use and socio economic characteristics (ageat first sex being the variable of interest

1.9 Theoretical Framework

1.9.1 The Social Cognitive Model

The Social Cognitive Theory (SCT) addresses both the psychosocial dynamics influencing health behavior and methods for promoting behavioral change. Within SCT, human behavior is explained in terms of a triadic, dynamic, and reciprocal model in which behavior, personal factors (including cognitions), and environmental influences all interact (Bandura, 1997). Eaton et al., (2003) argues that in order to understand sexual risk behaviour in Southern Africa, one needs to consider the interactive effects of factors at three levels: within the person, within his or her proximal context, and within the distal context. Personal factors include cognitions and feelings relating to sexual behaviour and HIV/AIDS, as well as thoughts about one's self (such as self-efficacy and self-esteem). In this context the proximal context comprises interpersonal relationships and the physical and organisational environment. The distal context includes cultural and structural factors within society. Culture comprises aspects such as the traditions and norms of the larger society, the social discourse within a society, shared beliefs and values, and variations in such factors across such subgroups and segments of the population. Structural factors include legal, political, economic, or organisational elements of society.

Figure 1: Framework for organizing the relationship between sexual behaviour, personal factors and the proximal and distal contexts.



Source: Eaton *et al.*, 2003: Unsafe Sexual Behavior in South African Youth; *Social Science and Medicine*(56)1: 149-165

This model states that understanding of the nature of HIV, the mechanisms of transmissionand methods of prevention is very important. In a study conducted by some researchers, fewer than 50 percent of young people understood how HIV and AIDS are related (Elkonin, 1993; Naidoo, 1994; Richter, 1996; cited in Eaton*et al.*, 2003) which is a reflection of serious gaps in knowledge and prevention. In other few interview studiesundertaken elsewhere, it appeared from respondents' spontaneous answers that their understanding of HIV/AIDS and prevention options was also sketchy (e.g. NPPHCN, 1996; Richter, 1996; Varga & Makubalo, 1996; cited in Eaton *et al.*, 2003). Within trusting relationships where contraception and sexual choices are discussed, condoms tend to be abandoned in favour of less intrusive, more effective contraceptives and that the act of leaving condoms behind symbolizes a new level of commitment within the relationship (Simpson, 1996; Wood & Foster, 1995; cited in Eaton *et al.*, 2003).

Furthermore, the importance of cultural and structural factors and the neglect of such factors in healthbehaviour research have been recognised in recent publications (Cockerham, 1997;Dressler and Oths, 1997; Eakin, 1997 cited in Eaton *et al.*, 2003). Since the model applied here is to understandinfluences on sexual behaviours specifically, a one-way process is depicted whereindividuals and their immediate environment are influenced by broader social conditions.Bandura (1977) cited in Eaton *et al.*, (2003) however argues that in general terms all these factors are potentially reciprocally determining. In other studies, it has been observedelsewhere that traditional African cultures are frequently patriarchal

and oppressivetowards women (Airhihenbuwa, 1995 cited in Eaton et al., 2003). Writers on HIV/AIDS prevention in Africa have noted that pervasive, culturally entrenched gender discrimination increases the risk of HIV infection for African women (e.g. Ng'weshemi, Boerma, Bennett, & Schapink, 1997; Webb, 1997; cited in Eaton et al. 2003). In order to explore and understand the influences on sexual behaviours specifically, depicting a process where individuals and their immediate environment are influenced by broader social conditionsthis paper will draw on this Social Cognitive Model. The social-cognitive theories havebeen found to be valid and useful, especially in Western societies. They cannot be applied to in all circumstances and to all problems. This is true for developing countries likeMalawi, where factors beyond the individual, cultural and distal have an impact that warrant special consideration. This model can be applied to adolescent HIV intervention programmes in Malawi by exploring crucial personal factors such as: the individual's capabilities to symbolize behavior, to anticipate the outcomes of behavior, to learn by observing others, to have confidence in performing a behavior (including overcoming the problems in performing the behavior), to self-determine or self-regulate behavior, and to reflect on and analyze experience (Bandura, 1997). This study explores the relationship between these personal, proximal and distal factors in relation to HIV risk perceptions and protective strategies among adolescents in Malawi.

1.10 Organization of the Dissertation

This report consists of six chapters. Chapter 1 gives an introduction and background to early sexual intercourse and the experiences that may influence a young person's HIV risk.It also highlights the global situation of HIV/AIDS and its consequences as well thesituation in Malawi with regards to the young people. The chapter explains why a betterunderstanding of coital debut is needed in HIV intervention programmes, highlightsobjectives of the study and gives a theoretical framework which can be applied in intervention programmes. Chapter 2 is a review of some of the existing literature on adolescent sexual behaviour, factors leading to early sexual debut and the associated HIV risk perceptions, as well as protective strategies adopted by the young people. Chapter 3 explains the sample, study area and how the cross sectional data will be analysed using regression methods. Chapter 4 presents the descriptive as well as themultivariate findings of the survey with tables and charts used to present the results. Chapter 5 gives an overview of the discussion on the findings of the

study. A conclusion as well as suggestion on possible recommendations basedon the findings is presented in chapter 6.

CHAPTER 2

LITERATURE REVIEW

2.1 Introduction

Malawi is representative of many other sub- Saharan African countries experiencing some of the most severe HIV/AIDS epidemics in the world. Adults aged 15 to 49 are most likely to engage in high-risk behaviour for HIV infection, and represent about 44 percent of the total population. For Sub Saharan Africa, AIDS has become the number one cause of death and accounts for more than half of all hospital admissions (JICA and Malawi MHP 1999; cited in Gerland, 2004). A high level of HIV prevalence in a country suggests a greaterprobability that an individual will acquire HIV if he or she engages in risk behaviors than ina country with low prevalence (UNFPA, 2009). In Malawi, HIV prevalence among pregnant women aged 15-24 years of age was 14.3 percent and decreased to 12.3 percent in 2007 (Malawi HIV and Aids Monitoring and Evaluation Report, 2007). It has since further decreased to 11 percent nationally, with female adolescents (15-19 years) prevalence rate higher (4.2 percent) than among males (2.7 percent), and 6.4 percent among those females aged 20-24 years compared to 4.7 percent of males of the same age group (MDHS, 2010). Someresearchers argue that AIDS awareness alone is a necessary but insufficient condition forindividual change in risk behavior (Gerland, 2004). Instead age at first sex may be apositive factor in understanding HIV risk perceptions, which then shape the young people's future protective behaviour. This chapter highlights age at first sex for young people inMalawi in relation to HIV risk perceptions as well as protective strategies and also highlights some of the factors leading to early sexual intercourse among the young people.

Firstly, some scholars argue that perception of risk is a personal risk-assessment and maynot reflect actual risk, but may reflect an individual's level of and access to knowledgeabout HIV (Prata *et al.*, 2006). Other studies argue that perception is socially constructed inthat social experiences influence the way in which people perceive superficially identicalrisks (Cross, 1992 cited in Prata *et al.*, 2006). Some scholars argue that a high perception of risk might lead to a modification of sexual behaviour, for example refusal to have sexual intercourse with a partner (Akwara *et al.*, 2003).

A number of studies have so far been carried out on the association between increased risks of unprotected sexual intercourse and early sexual initiation. Strong clinical research suggests that there is a relationship between socio demographic characteristics including age at first intercourse and an individual's health. If an adolescent engages in early age atfirst intercourse, they are likely to end up having multiple partners and also, engaging inunprotected sexual intercourse (Gueye *et al.*,2001). Further research indicates that earlyinitiation into sexual intercourse is associated with multiple partners, unsafe sex and a high risk of sexually transmitted diseases and HIV as well as unwanted pregnancy (Sieving *etal.*, 2006). Some studies have also shown that the perception of risk of HIV may be high when a new sexual relationship is formed, but that the perception of risk diminishes as the relationship progresses (Ingham &van Zessen, 1997; Fapohunda & Rutenberg, 1999; Nzioka, 2001 cited in Akwara *et al.*, 2003).

Young age tends to render especially girls unable to negotiate safe sex and they end up being pregnant or being infected with HIV and that the wider the ages gap between the partners the greater the sexual risk for the girls (Manzini, 2001). Most studies carried out on young people's sexual behaviour suggest that the younger the age at which first intercourse occurs, the greater the frequency of sexual intercourse and there is increased likelihood thatit will be unprotected and resulting in a higher risk of pregnancy and HIV infection (L'Engle *et al.*, 2006; Koenig *et al.*, 2004; Smith and Watkins, 2005). Further research shows that an individual's perception of risk is likely to be influenced by his or her ownsexual behaviour and his or her partner's sexual behaviour (Cleland and Maharaj, 2004). Astudy conducted in the US showed that early initiators had an increased likelihood ofhaving had multiple sex partners, been involved in a pregnancy, forced a partner to havesex, had frequent intercourse and had sex while drunk or high. The conclusion was that the adolescents who initiated sexual activity early engage in behaviors that place them at highrisk for negative health outcomes (O'Donnell *et al.*, 2001).

In African countries, the median age ranges between 16 and 18 years (Meekers, 1994) and the situation is not different in Malawi, where 14 percent of women aged 15–19 had sexbefore age 15 and two-thirds had sex before age 18 (Multiple Indicator Cluster Survey,2006). In a longitudinal study in South Africa in 2003-2007, the median age at first sexamong women and men who were virgins at the beginning of the period, was 18.5 and19.2 years, respectively (McGrath *et al.*,2008). Statistics on teen sexuality in the UnitedStates reveal that about 7 percent of high school students report having had sex before theage of 13 and 67

percent among younger adolescents regret their first experience and wish they had waited longer (Kim, 2008). In most cases this initiation into sexual activity occurs without any protection and result in unwanted pregnancies as evidenced by a study inMalawi where 60 percent of all respondents reported that sexual acts that took place in the three months prior to the survey were not protected at all and only 24 percent of the sex acts were protected 100 percent of the time (Munthali *et al.*, 2006). Manzini (2001) argues that once sexual activity begins, it generally does not stop, and in many parts of the world, the male partners are usually much older than the girls. Research reveals that many young men claim that they need sex in order to stay healthy and thatthere is also a pervasive belief that sexual desire is a natural force that one should not attempt to control (Meyer-Weitz *et al.*, 1998 cited in Eaton *et al.*, 2003).

On the same, Clark, Poulin, and Kohler (2009) argue that among the adolescents, sexual intercourse itself may be viewed as a signal of greater intimacy, trust, and commitment and establishing premarital sexual partnerships and gaining sexual experience is an important step toward marriage. Meekers (1994) adds that in urban areas there is less control on adolescent sexual behaviours and early sexual relations are viewed as a means of acquainting partners, and the modern education system orients the mindsets of the adolescents towards new social values and the school setting encourages interaction with the opposite sex. In Malawi it is estimated that 72 percent of men and 38 percent of women will have engaged in premarital sex before reaching their 20th birthday (Munthali, Chimbiri, and Zulu, 2004; cited in Clark, Poulin and Kohler, 2009). The trend is the same even in developed countries, where nearly40 percent of girls who begin sexual activity at ages 13 or 14 are reported to give birth outside marriage, compared to 9 percent of those who remain abstinent until their early twenties and one in four teenage girls have at least one sexually transmitted infection, and one in five will become teen mothers (Kim, 2008). Bertrand and Anhang, (2006) argue that a person's ability to avoid HIV infection depends onlypartly on their own individual knowledge and skills, but that there are other social and economic factors that are beyond the individual's control and that can put young people at higher or lower risk of infection. With regards to protective strategies, condom use during higher-risk sex is still low among young people in developing regions. On average, less than half of young men and just over a third of young women used condoms during their last high-risk sexual activity in sub- Saharan African countries (United Nations, 2011).

In summary, research has proven that the risk of HIV infection is lower among adolescents who begin sexual activity in later years (McGrath *et al.*, 2008; Munthali *et al.*, 2006; Gueye *et al.*, 2001). If indeed, age at sexual debut influences future HIV risk perceptions, as hypothesized in this study, then McGrath *et al.*,(2008) suggest that sex education and HIV education interventions can successfully delay sexual debut in developing countries. In a country like Malawi that is working towards achieving the Millennium Development Goals by meeting the health targets, and where adolescents are indulging in early unprotected sexual activities, there is a need for more in-depth research and holistic programmes for young people that are geared towards delaying sexual debut. The following section highlights some of the factors that may lead to adolescents engaging in early sexual intercourse and potential consequences such as increased likelihood of pregnancy and STIs, as well as being infected with HIV.

2.2 HIV Risk Perceptions: Worry and Likelihood of Infection

A review of literature indicates that perception of risk is a prerequisite for behavior change among adolescents, a supposition supported by empirical studies (e.g., Ajzen and Fishbein, 1980; Weinstein and Nicolich, 1993; cited in Smith and Watkins, 2005). Risk perception is defined as an individual's ability to judge whether he or she is susceptible to a condition and determine how severe or serious the outcome will be (Malawi Bridge Project, 2004; p15). There is research evidence that HIV/AIDS-related knowledge is significantly related to AIDS-related worry, perceived risk, and behavioral change (Gregson *et al.*, 1998; Klepinger *et al.*, 1993; London and Robles 2000; cited in Smith, 2003).

Perceptions of risks, as well as accurate knowledge of ways to avoid risks, play a crucial role in HIV/AIDS prevention (Clark, 2004) and are also strongly related to an increase in self-protective behavior (Lindan *et al.*, 1991; cited in Cleland and Maharaj, 2004). Furthermore, a higher perceived vulnerability and anxiety about personal risk is linked to greater intended and actual sexual behaviour change (e.g., Strebel and Perkel, 1991; Van Aswegen, 1995; Van Wijk, 1994; cited in Eaton *et al.*, 2003). Smith and Watkins, (2005) argue that there is a complicated reciprocal relationship between HIV perceived risk and behavior change. The high rate of HIV infection among adolescents in South Africa indicates that much sexual activity is unprotected by condoms (Rutenberg *et al.*, 2003).

Literature states that individuals who perceive no likelihood of HIV infection may not modify their behaviour even when their actual risk is high, while others may adopt a fatalist attitude where they decide HIV infection as inevitable and see no point in modifying their behaviour (Sherppard *et al.*, 2001). A low proportion of respondents from a study conducted in Kwa-Zulu Natal revealed that respondents consistently or occasionally using condoms with their spouse or partner (Cleland and Maharaj, 2005). Similarly, in a study in Malawi 60 percent of the sexual acts were unprotected and a significant proportion of the adolescent population were switching partners, which put them at risk of contracting HIV and other STIs and unintendedpregnancy (Munthali *et al.*, 2006). Furthermore, Forste *et al.*, (2002) argue that fear and concern about HIV motivates behavioural change, and those individuals who believe that they are already infected or would become infected with HIV/AIDS are more likely to be worried than individuals who perceive their risk of infection as negligible.

Worries about contracting HIV and self-perceived risks can serve as a motivation for adolescents to change behaviors that place them at risk to HIV. Some researchers argue that low perceived personal vulnerability is a risk factor because it reduces the motivation to take the necessary precautions (Eaton *et al.*, 2003). For example, in Malawi boys and girls who were fatalistic about their lives showed no motivation of taking protective strategies, and about 86 percent of them perceived themselves to become HIV positive within 10 years (Malawi Bridge Project, 2004), for instance, 'When you find out that you are HIV positive, you feel like you are already dead because of the insults that you receive (Malawi Bridge Project, 2004; p19)'.

Oyedokun, and Odimegwu, (2003) argue that those individuals who perceive small risk, moderate risk, great risk, and HIV-positive and ignorant individuals, are less likely to use protection compared to those who perceive no risk at all. In a study conducted in Uganda, the majority of adolescents especially the females across all age groups were very worried about the possibility of getting infected with HIV. On average, among the 15-19-year-olds, about 72 percent of the females compared to 55 percent of the males expressed this opinion and among the 12-14 year olds, the corresponding percentages were 65 percent compared to 47 percent respectively (Koenig *et al.*, 2007).

The DHS Comparative Report (2009) indicates that women and men felt that a person could reduce the risk of HIV infection by abstaining from sexual intercourse, limiting sex to

one faithful partner, and always using a condom when having sex; the socalled—A,B,C prevention methods. Furthermore, the report indicates knowledge of the A(abstaining), B (being faithful), and C (condoms) prevention methods varied considerably across countries, both among women and men (Sneeringer and Stacy, 2009). While levelsvaried widely within and across regions, knowledge was somewhat higher in east Africacompared with the other two regions. Further results also showed that knowledge levelsabout abstinence increased among men in Rwanda and Malawi, remained steady amongwomen in Rwanda, and declined among women in Malawi. Knowledge of faithfulness as aprevention method declined in Ethiopia and Malawi but increased in Rwanda and Uganda.Knowledge of condom use to prevent infection rose in all countries with two data pointsexcept Malawi, where it declined slightly among women (Sneeringer and Stacy, 2009).

Despite high HIV prevalence and high rates of sexual risk behaviors in Sub-SaharanAfrican countries, results from studies indicate that most young people in these countriesoften perceive themselves as being at low risk of HIV infection (Anderson et al., 2007). Forinstance, in South Africa it was found out that some communities perceived more risk forthe young people in their communities than the adolescents themselves, who were inclined to underestimate their own vulnerability (Rutenberg et al., 2003). Other studies reveal that perceptions of risk are unrealistically low in some groups with high rates of sexual activity and low condom use, where up to 45 percent of young men in certain samples practicedunsafe sex (e.g., Reddy, Meyer-Weitz, van den Borne, and Kok, 1999; cited in Eaton et al., 2003). In a study conducted in Malawi, where almost everyone knows about HIV/AIDS, some respondents reported that they were not at all worried about HIV, believing they werenot at risk of infection based on their knowledge and their spouse's behavior (Zulu andChepngeno, 2003; Watkins, 2003; cited in Smith and Watkins, 2005). In another studyamong adolescents in Malawi more than a third of adolescents perceived themselves atgreat risk of contracting HIV and another believed they had 'no chance at all' of contracting HIV(Munthali et al., 2006). In 1994, focus group discussions with primaryschool pupils in Malawi revealed that the youth generally perceived themselves at risk ofgetting HIV because of borrowing razor blades, visiting barber shops, sexual intercourse, injections and nonuse of condoms, 10 years down the line there was high knowledge levelsabout HIV/AIDS and STIs (Munthali et al., 2004). A study conducted three years later in 1997 showed that only 45 percent of the youth appeared to be concerned about gettingAIDS (Munthali et al., 2004). In 2004, while more than a third of adolescents

perceivedthemselves at great risk of contracting HIV, the proportion who believed they had nochance at all' of contracting HIV accounted for another third of all adolescents (Munthali*et al.*, 2006).

With regards to adolescents in marriage, recent survey results on perceptions on whetherconcerns about HIV are reflected in youths 'marital aspirations in Malawi reveal that themajorities of men and women agreed that a monogamous marriage with a spouse who wasrelatively poor, more religious, and came from their own village would lower the risk of HIV/AIDS (Clark, Poulin and Kohler, 2009). Furthermore, over 80 percent of unmarriedmen and women planned to get tested for HIV prior to getting married (Clark, Poulin andKohler, 2009). Evidence from a study in South Africa reveals a higher perception of riskwithin the community was associated with boys' increased desire to avoid a pregnancy inthe near future and this suggests that HIV is influencing boys' risk of unprotected sexand unintended pregnancy (Rutenberg *et al.*, 2003). Results from a South African study reveal that while a higher proportion of women than of men felt at risk of HIV infectionfrom their partner, women's perceived risk of HIV infection from their partner was themost powerful predictor of condom use (Cleland and Maharaj, 2005).

2.3 Misconceptions about HIV

Literature points out misperceptions exist among adolescents about how HIV is transmitted and how pregnancy occurs, which may influence adolescents' adoption of protective measures (Munthali *et al.*, 2006). In Malawi, a little more than half of all adolescents did not know whether or did not think that a woman could get pregnant the first time she had sexual intercourse or if she had sex standing up, while half thought that a woman could get pregnant if she washed herself thoroughly immediately after sex (Munthali *et al.*, 2006). Some adolescents (about 25 percent) who had heard of AIDS believed that the AIDS virus could be transmitted by sharing food or through witchcraft or supernatural means, and more than one third thought that HIV could be transmitted by mosquito bites (Munthali *et al.*, 2006). However, these myths no longer exist by now (5 years later) because of the massive campaigns on HIV awareness in Malawi. Further literature indicates that some adolescents perceive condoms as only for sex workers and promiscuous people, that using a condom means lack of love or trust, use of condoms will make people think they have HIV/AIDS, condoms make sex less enjoyable and that condoms cause loss of erection and premature ejaculation (UNFPA, PATH, WHO 2005). Further misconceptions about female condoms are that they cause itching or rash on genitals, that it is difficult to use them consistently, that they can break during sexual relations, that it is difficult to insert female condoms, they can get pushed in or pulled out of vagina, are noisy, inner ring of the female condom feels uncomfortable and that it is difficult to use them in different sexual positions (UNFPA, PATH, WHO 2005). Some studies have shown that the effectiveness of condoms at preventing pregnancy and the transmission of HIV is determined by how consistently and correctly they are used (Munthali *et al.*, 2006). Further literature adds that the more likely the use of a condom at each act of intercourse, the greater will be the reduction in HIV infection (Bracher *et al.*, 2004). For young couples, the perception of risk of HIV infection is important for adopting protective strategies, and it is possible for a woman's perceived risk of HIV infection to over-ride the man's objections (Maharaj and Cleland, 2005). However, perceptions among boys and girls make them believe that a woman does not have the right to ask her partner to use a condom (Malawi Bridge Project, 2004).

Results from DHS data shows that the percentage of women who knew that HIV cannot be transmitted by sharing food ranged from 28 percent in Benin to 89 percent in Rwanda, and among men from 42 percent in Madagascar to more than 90 percent in Rwanda and Malawi. Only three countries, Rwanda, Uganda, and Malawi, had data on misconceptions about HIV transmission risk from mosquitoes and from sharing food for two time points. The percentage who knew that HIV cannot be transmitted by sharing food increased among both women and men in all three countries. The percentage who knew that HIV cannot be transmitted by mosquitoes also increased among both women and men in two of the three countries (Rwanda and Uganda), but there was little change in Malawi (Sneeringer and Stacy, 2009). Such misconceptions could influence adolescents' adoption of protective measures and it is very crucial to fully understand whether age at sexual debut influences these misconceptions about HIV/AIDS as well as protective behaviours. Having correct beliefs about HIV/AIDS is important to avoiding infection, but surveys have shown that many people have misconceptions. Surveys conducted in Southern, West and Eastern Africa presents several of these misconceptions among women and men age 15-49. In the vast majority of countries, knowledge that a healthy-looking person could have HIV has increased among women and men from the earlier survey to the more recent (Sneeringer and Stacy, 2009).

2.4 HIV Protective Strategies

This section highlights some of the protective measures taken by the young people as a means of protecting themselves against HIV/AIDS. These include condom use, abstinence and delaying sex, as well as marriage.

2.4.1 Condom Use

Condom use during higher-risk sex is still low among young people in developing regions. On average, less than half of young men and just over a third of young women used condoms during their last high-risk sexual activity in sub-Saharan African countries (United Nations, 2011). Literature evidence that a person's self-assessed risk of HIV infection may be based on the extent to which he or she takes preventative actions, such as condom use. If this is the case, then those with low self-assessed risk of HIV infection might be those who are consistently using condoms in high-risk sexual relationships or whoare not at all engaged in high-risk behaviour (Adetunji & Meekers, 2000 cited in Sheppard*et al.*,2001). Some researchers arguethat condom use is associated with most of the demographic and behavioral characteristics such as age, education, religion etc. Previous studies have shown positive associations between education and condom use; ours shows asimilar trend, except for women with a secondary or higher education (Prata *et al.*, 2006).

Some researchers argue that one has to first identify a risk before preventative action is taken. Results from a study conducted in Ghana in 2009 shows that respondents whobelieved that they were at no or small risk of contracting HIV/AIDS, were less likely touse condoms than those who believed that they have a moderate or great risk of gettingAIDS, or who already had AIDS (Sheppard *et al.*, 2001).

There is significant evidence that sexual negotiation of any kind be it about condom use, faithfulness, or about the nature and frequency of sexual intercourse is lacking in manysexual relationships among young people (Eaton *et al.*, 2003). When young people becomes exually active, they must have the skills to practice safe sex. This means either beingfaithful to one faithful partner or consistently using a condom properly. Data fromhousehold surveys show that the proportion of young people using condoms is still quitelow even when they have sex with people who are not their regular partner (Bertrand and Anhang, 2006). Further research reveals that a person with low self-esteem is more likely

to think that condoms are offensive to their partner, to think that using condoms may maketheir partner think they are dirty, to be embarrassed about using condoms and to have anegative attitude towards condoms (Perkel et al., 1991; cited in Eaton et al., 2003).Condoms may be viewed as acceptable with casual partners, but unacceptable with moreserious partners, who may view condom use as indicating a lack of trust (Chimbiri, 2007;Civic, 1999; Muhwava, 2003; cited in Clark, Poulin and Kohler, 2009). Condom use oftendepends on the nature of the relationship (e.g., whether the partner is a boyfriend/girlfriendor a more casual partner), the relative power that the adolescent has in the relationship (e.g., whether one partner is significantly older or has given the adolescent money or gifts in exchange for sex) and whether alcohol was consumed around the time of sexual (Munthali et al., 2006). The disadvantage of condoms cited most often is loss of pleasure. Severalstudies report young men (and some women) claiming that they like sex to be 'skin on skin' (e.g., MacPhail and Campbell, 2000; NPPHCN, 1996; Reddy and Meyer-Weitz, 1997; Reddy et al., 1999; Richter, 1996; cited in Eaton et al., 2003). Further researchreveals that youth in South Africa feel that many condoms are required for many rounds ofsex; fear of condoms breaking or slipping; and awkwardness in purchasing condoms (Eatonet al., 2003). It is further argued that for those who believe the myth that condoms maydisappear into women, causing injury or death, this is obviously also a major disadvantageto condom uses (Reddy & Meyer-Weitz, 1997 cited in Eaton et al., 2003). It is interestingto note that in a study where 88 percent of sexually active respondents agreed that condomsprotect against AIDS, none actually used them (Govender et al., 1992; cited in Eaton et al., 2003). All of the above mentioned factors are also directed to misconceptions aboutcondoms.

2.4.2 Abstinence and Delaying Sex

Oyedokun and Odimegwu, (2003) argue that belief in abstinence, limiting of numbers of sexual partners, use of sterilized needles, partners taking blood test, other ways of preventing HIV/AIDS, being HIV-positive and risk perception about HIV/AIDS arecorrelated with use of protection at first sexual intercourse. Munthali *et al.*, (2006) cited in the Guttmacher Institute Report (2006) indicate that at least 7 in 10 adolescents werefamiliar with ways to avoid transmission of the AIDS virus, such as not having sex at all, being monogamous with an uninfected partner, or using a condom consistently and correctly.

Furthermore, some primary school pupils perceived themselves not being at risk of HIVinfection because they were not involved in sexual intercourse (Munthali *et al.*, 2004) whilesome studies indicate that married girls are more likely to resort to sticking to one partnerstrategy more often than did unmarried girls (Clark 2004). Some Researchers argue thatthere is further perceived disadvantage of abstinence in that it prevents people fromdemonstrating their fertility by conceiving babies (Eaton *et al.*, 2003) while others arguethat if delaying sexual intercourse in adolescence carries a lasting benefit of reduced STIsin young adulthood, this would represent a potentially huge impact for such a strategy in terms of long-term health benefits in the population (Kaestle *et al.*, 2003). Munthali *et al.*,(2006) argues (based on anecdotal evidence) that some young women use anal sex as asubstitute for vaginal sexual intercourse in order to avoid the risk of pregnancy and/or topreserve their virginity. However, while anal sex cannot result in pregnancy, it representshigh risks for STIs, including HIV, if protection is not used.

2.4.3 Marriage

Marriage is the primary indicator of the exposure of women to becoming pregnant. Morethan half of women in Malawi enter into marriage at the age of 18 years before they arephysically, mentally and economically prepared for pregnancy (Munthali et al., 2004). Insome countries most, adolescent girls, generally view sex within marriage as safer than sexoutside of marriage, which is implicitly assumed to be the only alternative (Clarke et al., 2004). Some studies indicate that some parents in Malawi, for example, deliberately encourage early marriages to shield their daughters from such perceived risks (Bracher etal., 2003 cited in Clark 2004). However some researchers argue that early marriage maypose a substantial risk with respect to HIV (Clark, 2004; Clark, Bruce, and Dude, 2006; cited in Clark, Poulin and Kohler, 2009). On the other hand, Clarke (2004) argues thatalthough married girls are less likely than single girls to have multiple partners, thisprotective behaviour may be outweighed by the risks associated with the lack of condomuse, increased frequency of sex, and high HIV prevalence among partners. DHS datashowed that married and single girls from Kenya and Zambia often reported that their partners represented their greatest source of risk (Clark 2004). Results from a study inMalawi show that both unmarried men and women were deeply divided about whether theythought marriage offered protection from infection or not. Men were slightly more likelythan women to view marriage as protective, and generally more respondents reported thatHIV risks associated with early marriage were greater and they believed that no one couldcompletely avoid infection (Clark, Poulin and Kohler, 2009). In the US, sexual activity atan early age may affect marital and economic stability later in life, and that among womenin their thirties, those who were sexually active during early adolescence are half as likelyto be in stable marriages as those who waited until their early twenties to have sex and that engaging in early sexual activity elevates the risk of teenage girls becoming pregnant and single mothers (Kim, 2008).

2.5 Correlates of Risk Perceptions and Condom Use

These may be categorized into three categories; background socio demographiccharacteristics (gender, age, province, marital status, religion, education), psychological factors (personal ties with person with HIV or died of AIDS and knowledge of AIDS), and sexual characteristics (age at first sex and condom use). These factors can predict the way adolescents perceive HIV/AIDS as well as condom use.

2.5.1 Respondent's Age

There is growing literature evidence that age of a person is a factor that may influencesexual behaviour and the level of perceived risk of HIV infection. There is often lowperception of risk among the younger ages compared to the older ages since knowledge of AIDS which is a catalyst of behavior change increases as age increases. Some researchershave found out that although the risk of HIV infection is high among young men andwomen they often do not perceive their risk to be high (Prohaska *et al.*, 1990 cited inAkwara *et al.*, 2003). In addition, a positive relationship between age and perceived riskhas been found for young people in South Africa, where older youth were more likely thanyounger youth to have had sex (Prata *et al.*, 2006). Men and women in their teens are atincreased risk of HIV infection because they often engage in unprotected sexual intercourse(Hulton *et al.*, 2000 cited in Akwara *et al.*, 2003). Furthermore some researchers argue thatsometimes there is pressure for girls to prove their fertility before marriage and those boysmay face pressure to prove manhood by impregnating a girl, or by having many sexualpartners (Meekers & Calvès, 1997; Nzioka, 2001 cited in Akwara *et al.*, 2003). Thisimplies that age of an individual is a determinant of risk perception, worry and protectivestrategies.

2.5.2 Education

Literature points out that as the level of education increased the odds of a perceived highHIV risk are increased (Sheppard et al., 2001) and that modern education may expose adolescents to the school environment that enables them to interact more with the partners of the opposite sex (Meekers, 1994). Those with no education are more likely to have thelowest odds of high HIV risk showing the importance of increased education. Someresearchers attribute this to lack of knowledge and awareness to make an informed risk-assessment (Sheppard et al., 2001). A study conducted in Ghana found out that womenwith primary education were 26 percent more likely and women with secondary or highereducation were 43 percent more likely to perceive themselves at high risk of HIV perceptioncompared to those with no education (Sheppard et al., 2001). This explains why the goal of most sex education programmes is to delay onset of sexual intercourse, and to promote consistent condom use and other forms of safer sex among those who have had sexual intercourse (Manzini, 2001). Munthali et al., (2004) argue that making comprehensive sexeducation available in schools may encourage adolescents to have sex. Although Munthaliet al.,(2004) does not mention whether the school environment affects late debut, resultsfrom a recent study conducted in rural South Africa indicate that this school environment isassociated with later first sex in men and women (McGrath et al., 2009). In Malawi, resultsfrom a national survey indicate that for women, education was related to the start of sexualact, where 29 percent of the women with no education had sex by age 15, compared with 5percent for women with at least some secondary education (MDHS, 2006). Much as thelevel of formal schooling may influence perception of HIV risk and sexual behaviour someresearchers have conflicting evidence. Caraël (1995) cited in Akwara et al., (2003) foundout that there was increased casual sexual activity among those with higher schooling. Education may also be an important protective factor, leading to a later age at first sex and higher HIV risk perceptions (Anderson et al., 2007). Recent studies have found out thathaving eight or more years of schooling was strongly associated with contraceptive use, condom use at last sex and consistent condom use in the last six months (Koenig *et al.*,2004).

2.5.3 Religion

Religions can also influence sexual behaviour through intermediate factors such as the ageat first sex, marital status, and access to information and services. It may also influenceattitudes to HIV and perception of risk (Akwara et al., 2003). There are scholarly variousarguments on whether varying levels of religious commitment between different churches, the type of messages about sex and institutional pressure within church or peer normswithin a churchbased social network have an effect in delaying first sex. Some scholarsargue that church attendance elevates the odds of having first sex among adolescents, perhaps because the church setting provides an opportunity for these males to meetpotential partners (Haas and Forste, 2002). Nzioka (1996) cited in Akwara et al., (2003) noted that religious people considered AIDS to be a disease that affected those whotransgressed against God. Consequently, those who were religious perceived their risk of HIV infection to be low. Some studies have shown that religion can also work to influence community practices and national policies. The Kenyan national AIDS programmes facedopposition from religious leaders at the onset, particularly on the issue of introducing sexeducation in schools and on condom use, which were thought to encourage promiscuity(Forsythe et al., 1996) cited in Akwara et al., (2003). Other studies have found out that religious affiliation was significantly associated with ageat first sex in women but not in men (Mc Grath et al., 2009). In a national study conductedin Malawi in 2006, results indicate that although religion may be very important to themajority of respondents, it was not an important reason given by adolescents for not havingsex (Munthali et al., 2006). However some researchers found out that church attendancewas an important determinant of delayed sexual activity, but only when a child's friendsattended the same church (Mott et al., 1996).

2.5.4 Geographical Area/Residence

Geographical area may have an influence on sexual debut. Residence, and whether a person lives in an urban or rural area, can determine the level of access to information andreproductive health services, which could influence sexual behaviour and perception of risk(Akwara *et al.*, 2003). Living in a more rural rather than a peri-urban area may beassociated with early or later age at first sex in men and women. Some researchers havefound out that for both sexes, age at first sex was significantly associated with place ofresidence (Mc Grath *et al.*, 2009). They further argue that their findings suggest that

theremay be a constellation of community-level factors that influence the timing of first sex (Mc Grath *et al.*, 2009).

Different ethnic groups with almost similar cultural practices usually live in proximity toeach other within the 3 regions of Malawi. The socio cultural contexts in these regions caninfluence people's perceptions through internalization of specific sexual beliefs, norms and practices (Akwara *et al.*, 2003). Results from a national survey indicate that women in theSouthern Region started having sex at an earlier age than women in the Central andSouthern Regions, 21 percent of women in the Southern Region had sex by age 15,compared with 10 percent in the northern and Central regions (MDHS 2006). This could bealso being attributed to the religious and cultural variations across the three regions.

2.5.5 Household Wealth

Family income may increase the likelihood of sexual activity among adolescents, suggesting that those from higher income families have more leisure time in which todevelop intimate relationships (Haas and Forste, 2002). Other researchers state that sexualactivity at an early age may also affect marital and economic stability later in life, and thatit is linked to maternal poverty (Kim, 2008). Kim (2008) found out nearly 30 percent ofmothers who began sexual activity at ages 13 or 14 lived in poverty compared to 12 percentof those who waited until their early twenties. Furthermore, among women in their thirties, those who were sexually active during early adolescence were half as likely to be in stablemarriages as those who waited until their early twenties to have sex (Kim, 2008). TheMalawi Demographic Health Survey results (2006) indicate that more women (20.3 percent) andmen (16.7 percent) from the lowest wealth quintile had sex before the age of 15, compared to less than 10 percent of men and women from the highest wealth quintile (MDHS, 2006).

2.5.6 Marital Status

Marital status is an important predictor of HIV risk perceptions and protective behaviour. It is hypothesized that married couples have a lower likelihood of engaging in risky sexualbehaviors than the unmarried couples and that unmarried respondents are more likely torely on condoms than those who are married. Recent survey results on perceptions onwhether concerns about HIV are reflected in youths' marital aspirations in Malawi revealthat the majorities of men and women agreed that a monogamous marriage with a spousewho was relatively poor, more religious, and came from their own village would lower therisk of HIV/AIDS (Clark, Poulin and Kohler, 2009).

Literature also states that whereas non-married women may have some ability to negotiatesafer sex, married women face extra challenges because of the fear of being suspected ofpromiscuity by their spouses, which may lead to unwanted consequences such as separationor even divorce. Often, married women acquiesce in unsafe sexual practices, even if theysuspect or know of their partner's extramarital relations (Blanc *et al.*, 1996). Although HIVcannot be spread through sexual intercourse in stable monogamous relationships betweenuninfected partners, among married women the presence and the nature of their partners' casual or extramarital sexual practices largely determines the risk of HIV transmission(Ahlburg *et al.*, 1997 cited in Akwara *et al.*, 2003).

In many African populations, first marriage is an important determinant of age at first sex, particularly for women (Mc Grath *et al.*, 2009). Recent studies show that women who bothdelayed sexual debut and did not marry their first partner were, once married, more likelyto experience marital disruption and to be HIV positive, and women who marry their firstpartner but who have sex at a young age however are also at considerable risk of infection(*Boileau et al.*, 2009). Results from Demographic and Health Surveys (DHS) data from sixAfrican countries conducted since 1985 indicate that males spend much longer thanfemales in the sexually active unmarried state in all countries. The authors give an implication that women would be exposed to higher risks of acquiring HIV from theirhusbands than vice versa even if after marriage neither partner had extramarital sex (Zaba*et al.*, 2004). In Malawi, the results from a national survey revealed that marital status madea difference in the likelihood of women having sex. While 19 percent of women 15-24 whohad been married had sex by age 15, the corresponding proportion for never-marriedwomen was 8 percent (MDHS, 2006).

2.5.7 Gender Roles

Gender plays a significant role in determining perceptions of risk of HIV infection. Genderbased beliefs, pressures, roles and power may influence the ability of males and females tobehave in ways that correspond to their risk perceptions. Females might realize that oncethey become sexually active, they may have less control than males in terms of condom
useand other protective behaviors, thereby increasing their chances of becoming infected withHIV. Also, young women might worry about becoming pregnant and transmitting HIV to ababy (Anderson *et al.*, 2003).

Sheppard *et al.*, (2001) states women generally have higher rates of HIV infection thanmen and this difference is more prominent among younger women. Studies have shownthat generally men have greater influence in intimate relationships and women are morevulnerable to infection and unwanted pregnancies (Gage and Njogu, 1994; MacDonald,1996 cited in Sheppard *et al.*, 2001). In a study conducted in Uganda it was found that menwere 60 percent less likely to perceive themselves at risk of HIV infection compared to females.Similarly, results from Ghana showed that men were 23 percent less likely to report themselvesat risk (Sheppard *et al.*, 2001). Other researchers argue that women may have a higherperception of risk because they lack the power needed to negotiate in sexual relationships.

Evidence from Blanc *et al.*, (1996) cited in Akwara *et al.*, (2003) suggests that about 25 percent of Ugandan men and women in their sample believed that a woman cannot refuse sexwith her partner, even if she knows that he has AIDS.

2.5.8 HIV Counseling and Testing

Literature states that HIV/AIDS testing and counseling potentially play an important role inthe construction subjective assessments of HIV status, as well as recalibration of these self-assessments (Prata *et al.*, 2006). If individuals act on these assessments, and individualswho incorrectly think they're infected are less likely to protect themselves in risk situations, then HIV testing can reveal such overestimations and thus prevent the spread of theepidemic (Prata *et al.*, 2006). On the other hand, if overestimating risk will result in morepeople protecting themselves, receiving an unexpected HIV-negative test result could leadto lowering perceived risk and a disregard for protective behavior. Evaluating the effect of HIV testing on shaping estimates of HIV infection and the subsequent effect on riskbehavior is an important future task for AIDS research (Prata *et al.*, 2006). This means that knowing one's HIVstatus plays a role in determining riskperceptions and protectivestrategies.

2.5.9 Age at First Sex

According to literature, the timing of first intercourse is an important indicator of the onsetand duration of exposure to the risk for both unplanned pregnancy, and STIs and HIV(Zaba *et al.*, 2004; Munthali *et al.*, 2006; Manzini, 2001). Research has shown thatyounger female adolescents have higher biological risks both for complications ofpregnancy and delivery and of STIs as well as HIV infection (Munthali *et al.*, 2006).However, Kaestle *et al.*, (2005) argue that earlier initiation of sexual intercourse is stronglyassociated with STIs for older adolescents but not for young adults over age 23 years.Unless the first intercourse is also the start of a mutually monogamous relationship, earlyage at first sexual intercourse is associated with a long period of exposure to sexual activity, a higher propensity to accumulate sexual partners, and increased chances of contractingsexually transmitted diseases (Dixon-Muller & Wasserheit, 1990; cited in Konings *et al.*, 1994).

Manzini (2001) argues that the high prevalence of HIV among 15-24 year olds in Kwa-Zulu Natal suggest that sex is initiated at an early age in South Africa, and other studies(Anderson et al. 2007; Pettifor, 2009; MDHS 2006) reveal that among those aged 15-24, more than half had had sex by age 18 and it is for this reason that first sexual intercourse isconsidered an HIV risk behavior. Literature also indicates that having sex with an olderpartner is likely to place adolescents at greater risk of STIs, including HIV, since olderpartners are more likely to have more sexual experience than a sexual partner of the sameage (Munthali et al., 2006). Such girls who have sexual relations with men who are more experienced and older risk their lives through illegal abortions, STIs, including HIVinfection (Munthali et al., 2004). In a South African study, girls first had sex with boyswho were about four years older than them, but the age of the partners with whom they firsthad sex ranged from 8 to 42 years (Manzini et al., 2001). Kaestle et al., (2005) also statesthat younger ages at first intercourse are associated with higher prevalence of STI incomparison with older ages. In a study conducted in Malawi, more than two thirds of respondents believed older partners carried a greater risk (Clark, Poulin and Kohler, 2009).Clark (2004) argues that the level of a girl's possible exposure to HIV via unprotected heterosexual intercourse is a function of how long she has been sexually active, the frequency with which she has had intercourse, her use of condoms or other STI-protectivemethods, number of partners and whether any of her partners are infected. Gregson et al., (2002), and Kelly et al., (2003) cited in (Clark,

2004), identified age of adolescent girls 'partners as a major risk factor, and argue that having sex with older partners substantially elevates HIV rates among adolescent girls.

Empirical evidence from DHS data of six African countries shows that there is a widediversity of patterns, trends, and sex differentials in age at first sex (USAID, 2009).Different countries have different legal ages of sexual consent, for instance in BurkinaFaso, Ghana and Malawi, the legal age of sexual consent for girls is 16 and in Uganda it is18, in each of these countries, when the girl is a virgin and a minor, irrespective of whetherit was consensual, sexual intercourse is deemed defilement and can result in prosecution of the male (Moore *et al.*, 2006). Other studies indicate that with a decline in age at first intercourse and increased reporting of coercive first sex, concerns of adverse reproductivehealth outcomes such as unintended pregnancy and sexually transmitted infections (STIs),including HIV, are important issues, particularly in the context of a high HIV prevalence(Maharaj and Munthree, 2006).

The Malawi Demographic and Health Survey (2006) indicates that the median age at firstsexual intercourse for women increased slightly since 2000 from 16.9 years to 17.3 years in2004 and by age 25, all women have had sex (MDH, 2006) and that 15 percent of womenaged 15-24 and 14 percent of men aged 15-24 had sex by age 15 while the HIV prevalencerate for adolescents aged 15-24 was 12 percent (Malawi Demographic and Health Survey,2006). Munthali *et al.*, (2006), cited in the Guttmacher Report (2006) states that more than 9 in 10 female adolescents and 8 in 10 males had sex for the first time with a boyfriend,girlfriend or spouse, and more than 7 in 10 adolescents did not use any contraceptivemethod the first time they had sex, while nearly 40 percent of females and 7 percent ofmales reported that they were not willing at all at the time of their first sexual intercourse.

2.5.10 Summary

The literature review above provides an essential foundation for this study and it can beconcluded that understanding the relationship between age at first sexual intercourse andthe perceived risks and protection strategies against HIV is therefore very crucial whendealing with adolescent HIV intervention programmes. There is not much of publishedresearch on how those who initiate early sex are likely to perceive themselves at risk of HIV and this lack of risk perception is a major factor which may lead into failure to adoptprotective measures. The literature provides strong evidence that it is important to take aholistic approach in trying to understand age at first sex and adolescent's risk perceptionsand protective strategies in the design of HIV intervention programmes for young people.

2.6 Factors Associated with Early Sexual Debut

To better understand the relationship between early sexual debut and the HIV riskperceptions among the adolescents and protective behaviour, there is need to alsounderstand the factors that lead these adolescents to indulge in such behaviour that exposes them to HIV risks and unwanted pregnancies. This chapter presents information on a range of such factors. This information will enable a better understanding of the social andeconomic reasons why young people are vulnerable to HIV and unwanted pregnancy.

Among others, socio-economic status, family, peer pressure and other social aspects of adolescents' lives that have been shown to influence their protective and risk behavioursare also discussed. According to Eaton *et al.*, (2003), the factors that promote such risky behaviours or createbarriers to safer practices can be categorised into personal factors; the proximalenvironment (including interpersonal factors, and the immediate living environment); andthe broader social context (including structural and cultural factors. The 'Protecting the Young Generation Study' in Malawi revealed that most young people in Malawi engagedin unprotected sex, often as a result of peer pressure, poverty, influence of pornographicmaterials, drug and alcohol abuse, lack of parental guidance, lack of access to condoms, non-use of family planning, and trans-generational sexual relationships resulting in HIVinfection, STIs and unwanted pregnancies (Munthali *et al.*, 2004). In South African studies, reasons cited for early unprotected sex include poverty, unemployment, overcrowding, andlow levels of education as well as lack of knowledge about HIV and AIDS (Du Plessis *etal.*, 1993; Preston-Whyte and Zondi, 1991; cited in Eaton *et al.*, 2003).

Literature also indicates that more often the girls engage in unprotected sex with olderpartners and they have little power to negotiate contraceptive use, and that the situationmay be exacerbated by some traditional practices in certain areas (Munthali *et al.*, 2004).To this effect, the Population Council is actively working towards increasing theavailability and use of the female condom, which is the only safe and effective female-initiated method that provides simultaneous protection against unintended pregnancy

andSTIs, including HIV (Population Council Annual Report, 2008). The following sections highlight some of the factors associated with early sexual intercourse among adolescents that render them at risk of HIV infection and unwanted pregnancies with regards to coercion.

2.6.1 Coercion

Coercion at first sex is an important social and public health problem that has a seriousimpact on the reproductive health and behaviours of young women, (Maharaj andMunthree, 2006). There is research evidence that it is common for the first sexual act to becoerced and that coercion may play an important role in compelling young women toengage in sexual intercourse at an early age. According to Manzini (2001), a girl's firstsexual intercourse is often unplanned and may put her at risk of STDs and HIV infection aswell as unwanted pregnancy. Moore *et al.*, (2007) add that coercive experiences at sexualdebut have been shown to be associated with other sexual risks throughout the life course. Four types of sexual coercion at sexual debut among unmarried girls namely; forced sex;pressure through money or gifts; flattery, pestering, and threatening to have sex with othergirls; and passive acceptance were revealed by a nationally representative survey of girlsaged 12-19 year old girls in Burkina Faso, Ghana, Malawi, and Uganda (Munthali *et al.*,2004).

Qualitative research also indicates that young people's heterosexual relationships in certaincommunities frequently involve sexual coercion of, and violence towards, the femalepartner. Although numbers are not generally reported, these studies describe young womenwho are physically forced or bullied into having sex. Boyfriends, who claim that a romanticrelationship must necessarily involve full penetrative sex when and how the man wants it,feel justified in using physical assault or threats of violence to coerce their girlfriends intohaving sex (Varga and Makubalo, 1996; Wood and Jewkes, 1997; Wood, Maforah, andJewkes, 1998; cited in Eaton *et al.*, 2003). This is true for Malawian adolescents, wheremales dominate over the females in influencing sexual activity and use of condoms. Otherresearchers attribute the lack of control over the terms of sex may be attributed to thenegative impact that coercion at first sex has dealt to the emotional and psychological well-being of the adolescent girls (Maharaj and Munthree, 2006). Results from Malawi showthat 38 percent of girls were 'not willing at all' at their first sexual experience followed byGhana at 30 percent, Uganda at 23 percent and Burkina Faso at 15 percent (Munthali *et al.*, 2004). Results from the DHS Comparative Report (2009) indicate that the percentage

ofyoung women forced at first sex ranged from 6 percent in Niger and Uganda to 27 percentin Malawi. Young women forced to have sex in last 12 months preceding the survey rangedfrom less than 1 percent in Niger and Uganda to 8 percent in Zambia.

Sexual coercion is associated with a higher likelihood of HIV infection and experiencing ofunintended pregnancies. Munthali et al., (2006) argue that younger male and female teensare likely to be more exposed to the risk of sexual coercion than the older adolescents andthat age is very likely to be related to knowledge of risks and means of protection. In astudy in Kwa-Zulu Natal, South Africa, results show that those who had been coerced atfirst sex were more likely to report having had an STI and having experienced unintendedpregnancy, than those who had not been coerced at first sex (Maharaj and Munthree, 2006). In another study in Cape Town, South Africa women who had experienced coercion were significantly more likely to exchange sex for material needs, have multiple sex partners, engage in high rates of unprotected vaginal intercourse, and had more STIs (Moore et al.2006). Furthermore, adolescents in Uganda who reported coerced first intercourse were significantly less likely than those who did not experience coercion at debut to be currentlyusing modern contraceptives, to have used a condom at last intercourse and to have usedcondoms consistently in the last six months and were more likely to report their most recentpregnancy as unintended and to report one or more genital tract infections (Moore et al.,2006).

2.6.2 Socio-economic status

The socioeconomic situation in which young people live can have both positive and negative impacts on their vulnerability to HIV infection and that lower socioeconomicstatus may result in lower educational attainment, which may result in gaining lessinformation and skills to protect oneself from HIV (Bertrand and Anhang, 2006). Somestudies have shown that there is a very close relationship between poverty and the spread of HIV/AIDS. Munthali *et al.*, (2004) and Adams & Marshall, (1998) cited in Eaton *et al.*, (2003) state that economic deprivation in certain households often leads girls to engage insexual intercourse for economic benefits, where older boyfriends with money will usually offer gifts and financial assistance that parents cannot afford to the adolescent girls .

Further literature from South Africa supports the fact that poverty is the major reason for he spread of HIV, whereby women subjected to economic crisis indulge in sexual relationships with men in exchange for financial support cited as follows; 'Poverty is oftenthe reason for the commodification of sex, in which women in dire economic circumstances agree to sexual relationships with men in exchange for financial support. In other instances he situation may not be so overtly an exchange, but an older boyfriend with money offersboth status and the kinds of gifts and financial assistance that parents cannot afford (Adamsand Marshall, 1998; Kelly and Parker, 2000; cited in Eaton et al., 2003: p15)'.Rutenberg et al., (2003) cite evidence from some studies on the importance of material and emotional support for the young girls by the older partners. The exchange of sex for money or gifts means that sex happens on the man's terms which usually means without a condom (Karim, 1998; Adams and Marshall, 1998; cited in Eaton et al., 2003). In Malawi adolescent girls from poor households aremore likely to engage in early sexual relations and early marriage for financialgains and some poor parents actually encourage their young daughters to providesexual favors so that they can make money to support the family. In a nationalstudy in Malawi, 66 percent reported that they accepted money or gifts in exchange forsex of female adolescents who experienced forced sex, (Munthali et al., 2004).

Further literature also indicates that the sexual domination of young women bytheir partners appears to happen more often in poor communities. These social andeconomic forces often put adolescent girls in a situation where they have to bargaintheir moral values just for survival and as a result, they often find themselves involved in nonconsensual relationships with more than one man (Munthali et al., 2004). This often results in young women engaging in sex in exchange for money or gifts usually from older men and their perspective protection from possiblefuture illness may be a lower priority than meeting immediate economic needs(Eaton et al., 2003). It should be mentioned that not only adolescents indulge in sexin exchange for financial gains; female police officers, female estate workers, andfemale primary school teachers in Malawi got involved in risky sexual relationshipswith men who could give them money to supplement their incomes as their salaries were too low to cater for all their needs (Behavioural Surveillance Survey, 2006). Finally it can be deduced that under conditions of economic hardships and social pressure, gift giving has the potential to coerce adolescents into sexual activity(Maharaj and Munthree, 2006) and that for girls sex may bring benefits in the formof gifts or financial support for girls, while for boys having many girlfriends can bean affirmation of manhood (Rutenberg et al., 2003).

2.6.3 Peer Pressure

The World Health Organisation Report (2004) states that adolescents who perceive theirfriends or peers to be sexually active are significantly more likely to engage themselves insex, as well as have multiple sexual partners. In addition, the perception that one's peers aresexually active was one of the strongest identified risk factors.Peer pressure may be a result of social interaction, where it should be expected that theproportion of network partners very worried about AIDS should be the main determinant of the probability of the respondent reporting being very worried as indicated below;If social influence dominates, on the other hand, a different pattern should be expected: when a respondent's network is sparse, variations in the proportion of network partners who are very worried should exert only small effects on therespondent's probability of being very worried, whereas these variations should beassociated quite strongly with a woman's risk perception in dense networks(Hellerringers and Kohler, 2005: p71).

Some studies indicate that both girls and boys experience considerable same-sex peerpressure to be sexually active and that for boys the pressure has to do with provingmanliness, and having many sexual partners, while for girls pressure sometimes comesfrom sexually experienced peers who exclude inexperienced girls from group discussionsbecause they are still 'children' (Buga et al., 1996; Cassimjee, 1998; NPPHCN, 1996; (Blecher et al., 1995; MacPhail and Campbell, 2000; Wood et al., 1997a; cited in Eaton etal., 2003: p14). Rutenberg et al., (2003) state that although the majority of adolescents maynot wish to become pregnant soon, they may be under a great deal of peer pressure to have a boyfriend or girlfriend, and for boys especially, to have many partners, and sexual activity with a partner confers the status of a relationship. In a study in South Africa, girls' preferences about the timing of a pregnancy were found out to be significantly associated with their peer group's perceptions of risk, which perhaps could be a proxy for their perception of their own personal risk (Rutenberg et al., 2003). It was found out that youngwomen who lived in communities where a high percentage of adolescents thought theirfriends were at high risk of acquiring HIV were significantly more likely to perceive thatbecoming pregnant in the near future would be a big problem than young women who

livedin communities where peers perceived less risk to their friends (Rutenberg *et al.*, 2003).Perkel, (1991) cited in Eaton *et al.*, (2003) state that young people with low selfesteemmay be more concerned about what their partners think of them and with avoidingdispleasure or rejection from partners than are people with more positive, selfaffirming self-concepts. Furthermore, the desire of traditionally minded young Black men in SouthAfrica to prove their virility by fathering children, and the pressure felt by girls to provetheir love and fertility by conceiving, has been well documented (Kau, 1991; Preston-Whyte and Zondi, 1991; Varga and Makubalo, 1996; Wood, Jewkes, and Maforah, 1997cited in Eaton *et al.*, 2003).

2.6.4 Family Structure

Research indicates that a key parental influence on teen sex is family structure, which referslargely to the number of parents living with, and their relationship to (biological, adoptive, step, or unrelated), the children in the families (Kim, 2008). Bertrand and Anhang (2006) argue that open communication about sexuality remains a challenge in many cultures and societies. Social science research over the decades suggests that parents can play aprotective role in delaying early teen sexual activity and reducing the risk of harmfulconsequences and such protective parental influences include the intact family structure, parents' disapproval of adolescent sex, and a strong parent-child relationship (Kim, 2008). Boys and girls may be embarrassed to discuss issues related to sex; parents may be nwilling to talk about sex or be uncomfortable doing so, both of which may result inyoung people having limited knowledge and skills about prevention. In addition, parental factors that appear to offer strong protection against the onset of early sexual activityinclude an intact family structure; parents' disapproval of adolescent sex; teens' sense ofbelonging to and satisfaction with their families; parental monitoring; and, to a lesserextent, parent-child communication about teen sex and its consequences (Kim, 2008). SouthAfrican adolescents report poor communication with parents about sexual matters and thatwhen there is poor communication within the family about sex, both supervision and lackof supervision from parents may contribute to unsafe sexual behaviour (Eaton et al., 2003). In the US, nearly all parents (90 percent) report having had a helpful conversation aboutdelaying sex and avoiding pregnancy with their teenage children, compared to 71 percentof teens who report having had such a conversation with their parents. Further research in the US reveals that teens living in intact families also tend to report fewer

44

sexual experiences and partners, and are less likely to report being infected with a sexually transmitted disease, compared to peers in non-intact families. This is primarily because teens in intact families tend to delay sexual activity (Kim, 2008).

Research results also reveal that increasing number of young people who do not grow upin a protective environment in which they have parental support, as a result of adultmortality from AIDS, may be increasingly vulnerable to infection (Bertrand and Anhang,2006) For instance, a study in rural Zimbabwe found out that among women aged 15–18 years, young people whose mother had died and young women with an infected parent hada significantly higher prevalence of HIV than other young women, and they also had moreSTI symptoms and were more likely to become pregnant (Bertrand and Anhang, 2006).In Malawi as most African countries sex is regarded as a taboo subject and is not discussed between parents and children. Munthali et al., (2004) argues that parents are not a majorsource of information because they do not normally talk to their children about sexual and reproductive health issues; this responsibility is left to other relatives, such as aunts, grandparents and other people within the community such as the traditional initiators. Asurvey in Malawi revealed that young girls and boys may imitate with their peers what theysee at home, since in most communities they share rooms with adults who have sex whilethey watch (Malawi Bridge Project, 2004). These results however leave a lot to be desired, since the evidence is not common practice in Malawi. Other studies have also shown that negative experiences in childhood, such as being physically or sexually abused, canincrease the probability of engaging in risky behaviors later in life (Munthali et al., 2006).

2.6.5 Cultural Practices

There are norms and values in communities that increase or reduce the risk of HIVinfection. The prevailing cultural and socioeconomic contexts have a strong impact on patterns of adolescent sexual and reproductive behavior. According to Munthali *et al.*, (2004), some cultural practices impact on STI/HIV/AIDS transmission and other sexualand reproductive problems on adolescents. Other researchers argue that at community levelideologies of masculinity and femininity may help to shape an individual's identity andself-concept and may thus have a strong impact on his or her role in sexual decision-making (Gage, 1998 cited in Akwara *et al.*, 2003). Akwara *et al.*, (2003) state that socialnorms may define 'a good woman' as one ignorant of sex or passive in sexual

encounters, whereas a real man 'may be defined as being sexually experienced, possibly with several partners'.

Munthali et al., (2004) state that the socialization process that young people undergo isinfluenced largely by the existence of opinion leaders, including village headmen, traditional initiators, traditional birth attendants, churches and mosques, local politicalleaders, and teachers. Furthermore, some cultural practices, such as initiation ceremonies, early sex, early marriage and funerary cleansing rites have a strong role in shaping the sexual behaviors of young people in Malawi (Munthali et al., 2004). While initiationceremonies are a common practice in most Sub Saharan Africa and are meant for character-building, they are also known to encourage early initiation of sex where many initiatescome out feeling that the initiation actually certifies them to experiment with sex since theyare now considered adults (Munthali et al., 2004). Most boys and girls in Malawi are initiated at early ages (as early as 7-9 years) and upon graduation, they are advised to 'to throw away the dust' have sex, which means to have sex with girls who have also beeninitiated within the same period (Malawi Bridge Project, 2004; Munthali et al., 2004).Some traditions in Malawi arrange for a man to have sex with initiated girls in order tocleanse them, and most of the time without any protection, for instance girls who areinitiated get to sleep with a man termed a 'fisi' (hyena¹) whose role is to initiate girls intosexual intercourse that the 'fisi-hyena' can sleep with several girls on the same night and without any protection, thereby increasing the vulnerability of girls to sexually transmittedinfections (STIs) and HIV(Munthali et al., 2004; p13)'. Munthali et al., (2004) further argue that cultural norms are widely held that women shouldbe inexperienced and naive in sexual matters and that pleasing men is the primary goal ofsex and traditionally, early marriages have been used as a form of protecting of young girlsfrom engaging in premarital sex and pregnancy in Malawi. However, Clark, Poulin andKohler (2009) argue that people who believe that marriage will offer relative safety from the deadly virus may choose to marry earlier and those who fear becoming infected by theirspouse may choose to delay marriage. There is also evidence that the norms regarding premarital sexual activity in African societies vary strongly, and in certain societies virginity is no longer highly valued, as in Uganda where premarital sexual relations areencouraged by constructing

¹ A male who is designated to sleep with girls who have just been initiated, they are termed 'hyena because they will sneak into the girls home at night.

separate dwellings for sexually active adolescents (Meekers, 1994). Certain societies in Zimbabwe greatly disapproved premarital sexual relations andgirls who have lost their virginity before marriage are required to confess, and theresponsible lover would be forced to marry them (Meekers, 1994).Such traditional practices likely to expose the adolescent girls to HIV infection, STIs and unwanted pregnancies.To overcome these cultural barriers, Meekers (1994) points out the possibility of education graditional control of adolescents. The WHO Report (2004) states that education schooling are key factors not only for reducing the risk of early sexual initiation, pregnancy, and early childbearing, but also for overcoming cultural factors that perpetuate the likelihood that adolescents will use condoms and contraception when they have sexualintercourse. Religion is another social factor in which Muslim women were less likely toindulge in early sexual intercourse than Christian women (Meekers, 1994). WHO recommends that programmes and policies that focus on improving school enrollment, retention, and performance among adolescent sexual and reproductive health outcomes (WHO, 2004).

2.6.6 Gender Disparities

This section explains men's and women's attitudes toward negotiating safer sex withhusbands, spousal communication about HIV/AIDS, and women's participation in major household decisions. Women's status relative to men, ability to negotiate safer sex, andparticipation in household decision-making can affect their vulnerability to HIV infection and their ability to take preventive action (Sneeringer and Stacy, 2009). Biologically,women are more than three times more vulnerable to HIV infection than men. Moreover,poverty, gender inequality, and socio-cultural taboos work together to increase youngwomen's vulnerability. Because sexuality is a controversial topic in most cultures, youngwomen are denied education about their bodies, reproductive health, and sexuallytransmitted infections (STIs) such as HIV. Common prevention strategies ignore the inability of young women to negotiate with their partners on abstinence, monogamy, orcondom use. Millions of young women's first intercourse is marked with violence, forceand coercion and is unlikely to occur with the protection of condoms (UNFPA, 2004).

Gender based norms do influence an individual's behaviour towards their perceived riskstowards HIV and unwanted pregnancy (Anderson *et al.*, 2007). Literature also indicatesthat in many societies, people turn a blind eye to sexual abuse against young women,

and togender-based violence more generally, both of which have important implications for HIVtransmission. Literature also indicates that there are accepted notions of masculinity andfemininity play a role in perpetuating gender disparities. For instance, in many culturalsettings, young women are supposed to be sexually innocent and may therefore be reluctantto carry or suggest using condoms. Since condoms are also associated in many contextswith illicit or extra-marital sex, married women are often powerless to request their partner wear a condom despite suspecting that he may be infected with HIV, for fear of reprisalat the implied accusation of being unfaithful (UNFPA Report, 2009). Furthermore, it isargued that when the status of women within sexual relationships is low they are at anincreased risk of contracting HIV. For example, for many girls and young women the onsetof sexual activity does not occur by choice (Bertrand and Anhang, 2006).

Results from a study conducted in South Africa show that those females who delayed theirsexual debut were those who perceived high risk of HIV if they engaged in sexualintercourse. Most adolescents are not mature enough to view HIV as a risk when they engage in their first sexual encounter and the earlier the youths engage in sexual intercoursethe higher the risk of being infected with HIV. Some researchers argue that more and moreadolescents will continue being infected in the coming decades, a trend that is likely tochange demographic characteristics of Sub Saharan populations in terms of age structure(Anderson et al., 2007). It is also argued that the increasingly disproportionate impact of HIV and AIDS on women and girls throughout the world has implicated gender inequity as a driver of the epidemic. Furthermore, the cultural construction and social organization ofgender have disempowered women and girls in many settings, making them particularlyvulnerable to HIV infection and its consequences (Bertrand and Anhang, 2006). Eaton etal., (2003) argues that research into the discourses that surround the subordination of women, reveals two main themes relating to male sexuality: biologically determined 'need', and sexual 'rights'. They claim that it is in man's nature to want many partners, andthat staying with one woman therefore goes against the essence of being a man and that some women come to believe this, too (Eaton *et al.*, 2003).

2.6.7 Decline of Age at Menarche

Literature shows that age at menarche has declined in the past decades. In this context itmay be regarded as a proxy determinant to early sexual initiation since young girls may experiment in sexual activities once they reach menarche. Zabin and Kiragu, (1998) arguethat declining

age at puberty exposes young girls to the risk of premarital sexual activityand there may be an increasing trend in early sexual onset as the age at menarche drops.Furthermore decline in age at menarche may have negative health impact on thereproductive health of the adolescents in the context of early marriage and childbearing orcontracting sexually transmitted diseases or HIV (Zabin and Kiragu, 1998). The age at menarche in Malawi ranges from 13-15 years, but can occur as early as 9 years (Munthaliand Chimbiri, 2003). In South Africa, the mean age at menarche was reported to be at 13.7(Vundule *et al.*, 2001). Probably, this early menarche can be attributable to improved healthand nutrition and it has been reported that the earlier the age at menarche, the earlier thefirst intercourse is likely to occur.

2.7 Psychological Factors: Personal ties to people with HIV/AIDS

Personal ties to people with HIV/AIDS makes the AIDS epidemic a real part ofadolescents' lives. Although there is no direct link between age at sexual debut andpersonal ties to people who are directly affected by HIV/AIDS, it is worth examining theeffect on protective behaviour. Munthali *et al.*, (2006) argue that having a personal tie to someone with HIV/AIDS could make adolescents more cautious in their own risk andprotective behaviours.

According to the AIDS risk reduction model, knowledge of AIDS is a prerequisite torecognizing risky behavior and taking action to change it, but findings regarding therelationship between knowledge and behavior have been inconsistent (Prata et al., 2006). A cross-sectional study using DHS data from Uganda, Kenya and Zambia showed thatknowing somebody with AIDS was predictive of protective sexual behavior, as wereknowledge of HIV prevention methods and correct beliefs regarding AIDS patients (Prataet al., 2006). Koenig et al, (2004) also state that knowledge of someone who had AIDS orwho had died of AIDS may increase an individual's awareness of the consequences ofHIV/AIDS and may lead to safer sexual practices (Koenig et al, 2004). A study in SouthAfrica revealed that knowing someone who had died of AIDS was associated with sexualdebut and with an elevated perceived HIV risk among females (Anderson et al., 2007).Oyedokun and Odimegwu (2003) argue that having relations who have died of HIV/AIDShave impact on the use and consistency of condom use during sexual intercourse. Munthaliet al., (2006) further states that holding negative attitudes towards people with HIV/AIDScan influence one's own willingness to acknowledge risk and to get tested for HIV.

Anderson *et al.*, (2007) state that personal experience and familiarity with HIV/AIDS maybe associated with more awareness of infection pathways, less stigma toward the diseaseand higher perceived risk of HIV infection. For example, the death of someone with AIDSmay increase the subjective perception of risk among all members of the family orcommunity even though some may be at very low risk. This literature above highlightssome of the evidence on how AIDS information plays a role in shaping adolescents' HIVperceptions and protective strategies.

CHAPTER 3

METHODOLOGY

This chapter presents research methods used in the study, including research design, research setting, study population, sampling and data collection processes as well as the analytical strategy. In order to depict a wider scope of the methodology, a description of reliability and validity of data used in the study, as well as study limitations is also included.

3.1 Study Area



Figure 2: Showing the map of Malawi and the Study Area

Source: http://www.un.org/depts/Cartographic/map/profile/malawi.

Malawi is a landlocked country located in Southern Africa. It is bordered to the north andnortheast by Tanzania, to the east, south and south west by Mozambique and to the west byZambia. The country is 901 km long and ranges in width from 80 to 161 km. It has a totalof 118, 484 square km of which 94,276 square km is land area. The remaining area ismostly composed of Lake Malawi, which is about 475 km long and runs down The country is divided into three Mozambique. Malawi'seastern boundary with administrative regions, namely Northern, Central, and Southern. There are 28 districts; six in the NorthernRegion, nine in the Central Region, and thirteen in the Southern Region. These districts vary in population, geographical and socioeconomic factors. According to the 2008 Population and Housing Census, the total population was estimated at 13,066,320. Withregards to the economy, Malawi is a low-income country with an estimated gross domesticproduct (GDP) per capita of 667 (PPP 1 US\$) (World Bank 2008). In 2005, official development assistance constituted about 27.8% of the GDP (UNDP 2007). Like mostcountries in the Sub-Saharan region, Malawi is classified in the group of low humandevelopment countries with a human development index (HDI) of 0.437 in 2005. Thecountry's Human Development Index rank during the same period was 164 out of 177countries (UNDP 2007).

The epidemiological profile is characterized by high prevalence of a communicablediseases including malaria, tuberculosis and HIV/AIDS; high incidence of maternal andchild health problems; an increasing burden of non-communicable diseases such as cancers, hypertension, diabetes, cardiovascular diseases and mental illnesses, among others. Thecountry also faces the resurgence of neglected tropical diseases. Although there has been asignificant decline in infant mortality from 76/1000 live births in 2004 to 69/1000 livebirths in 2006 and under-five mortality from 133 in 2004 to 122 in 2006 (MICS², 2006), therates are still high. The maternal mortality ratio of 807 per 100,000 live births was one of the highest in the world (MICS, 2006). In 2005, the adult HIV prevalence rate was 14.0%, in females (ages 15-24): 9.6% in males (ages 15-24) in 2005: 3.4% and number of deathsdue to AIDS in 2005 was 78, 000. The HIV and syphilis sero-survey of 2007 estimated national adult (15 - 49 years) HIV prevalence rate of 12 % (MOH³, 2007) which declined to 11 percent in 2008 (MDHS, 2010). The number of HIV persons who were alive and on Highly Active Antiretroviral Therapy increased from 61,430 at mid-2006 to158, 137 by March 2009

² Multiple Indicator Cluster Survey

³ Ministry of Health

representing 64% of those ever startingtreatment. Although the prevalence rate for Tuberculosis (TB) was not known because theprevalence survey was yet to be done, the 2007 TB annual report indicated a cure rate at78%. In 2004, despite the major gains in reducing childhood mortality, life expectancyworsened in the other age groups, due in part to the HIV and AIDS pandemic. Overall, lifeexpectancy declined from 44 years in 1990s to 38 years by 2004. The health system faces acritical shortage of human resources up to date. In 2009, the doctor to population and nurseto population ratios were 1:53,176 and 1:2,964, respectively; far below the WHOrecommended standards for developing countries of 1 doctor per 5,000 populations and 1nurse per 1,000 populations (Malawi Government, Health Information Systems AssessmentReport, 2009).

3.2 Data Source and Sample

This dissertation uses the 2004 secondary data which comes from a wider study, theMalawi Diffusion and Ideational Change Project (MDICP), which is a joint collaboration between researchers from the Population Studies Center at the University of Pennsylvania, and the University of Malawi's College of Medicine and Chancellor College. The project collects panel data in the three administrative regions of Malawi: Balaka (southern), Mchinji (central) and Rumphi (northern) and was designed to investigate the role of socialnetworks on HIV risk and related behavior. The last round of data collection was in 2010. The MDICP's first round of data collection occurred in 1998 and included a sample of 2,640 ever-married women and spouses in 125 villages across the three study sites. In 2001 the sample was augmented with new spouses resulting in 1571 women and 1097 men. Thethird wave of MDICP data collection took place in 2004 in which the project added 1,001 young women and men aged 15-28 to the 1998 cohort of ever-married respondents in eachsite in order to adjust for aging of the 1998 sample over time, which led to under-representation of the adolescent population by 2004; and to introduce nevermarriedadolescents into the MDICP sample (the 1998 sample was restricted to ever-married men and women). These adjustments made the 2004 sample representative of the rural population in each sample districts.

3.2.1 Sample Size Calculation.

The initial sampling strategy of the whole project was not designed to be representative of the national population of rural Malawi, although the sample characteristics closely match the characteristics of the rural population of the Malawi Demographic and Health Survey. The target sample for the first MDICP in the 1999 wave was 500 ever married women age15-49 in each district, plus their spouses. The third survey (2004) wave added a sample of approximately 400 adolescents age 15-29 in each district. The majority of these 400 were the ones that form part of the sample in this study. In 2004, the MDICP added an adolescent panel of men and women aged 15-29 to its existing sample of ever-marriedwomen and their spouses (Weinreb, 2006). Following the definitions UNFPA, WHO and UNICEF definitions which state that 'youths' as those aged 10-24 years of age, andfilter out those age above the upper limit of the last age group. The final total sample in this thesis was 659 respondents and comprised of 403 males and 256 females. In brief, this panel can be broadly characterized as a disproportionate stratified randomsample with stratification by sex and age (Weinreb, 2006), and took place in three roughlychronological stages as follows:

Stage 1: In order to maximize the number of adolescents from existing MDICPhouseholds, a 100 percent sample of adolescents listed in the main survey interviews wasrandomly drawn in order to allow for interesting and potentially important within-household linkages. These interviews accounted for approximately 32 percent of all adolescentinterviews ((Weinreb, 2006).

Stage 2: In order to generate the remaining 75 percent of the adolescent sample, adolescents weredrawn on the prior household rosters collected by listing teams. However, a problem wasencountered; drawing a simple random sample of adolescents would have generated adataset with high levels of multicollinearity between age and marriage due to the differencein mean age at marriage between men and women ((Weinreb, 2006).

Stage 3: In order to reach a target of approximately 350 individuals per site, or at least getcloser to it, some individuals were consequently added to the original sample lists. This wasdone by drawing them at random from the household rosters. In other words, among thesefinal additions, there was an equal probability of being selected into the sampling list.

3.2.2 Fieldwork and Data

The cross sectional data used in this study was collected in the 3 regions of Malawi fromApril to August, 2004. There were two data collection teams: the 'main survey' and the 'biomarker collection' teams where the survey team first administered the mainquestionnaire and later followed by the biomarker collection team, which typically visitedrespondents two or three days after the main survey interview and conducted HIV tests. Also, because HIV and STI tests were conducted in a laboratory in Malawi (as opposed tousing rapid results HIV test kits, as MDICP did in 2006 and 2008), test results were given to respondents between two and four months after testing at each fieldwork site (Anglewiczet al., 2009). All respondents had to consent to participation in the survey and HIV/STI tests. Because the administration of such tests required personnel trained in biomarkerspecimen collection and HIV/STI counseling, the project recruited a team of nurses toprovide counseling, collecting biomarkers, and administering a short questionnaire. Theadditional personnel and time required to complete both the main survey and the biomarkercollection necessitated two separate visits to each respondent. In addition, field logs weregenerated using Stata 9, transferred into Excel and Word files, and then given tosupervisors for the next days' fieldwork and where necessary changes to field logs could beimproved using the Access database (Anglewicz et al., 2009).

3.2.3 Methods of analysis

The aim of this study is to investigate whether background socio-demographic factors areassociated with young people's individual's risk perceptions of HIV, (i.e., respondents' evaluation of their chances of getting AIDS) and protective behavior measured by condomuse at last sex. The selected socio demographic covariates included in this study are respondents' background characteristics: age of respondent, gender, marital status, religion,province, educational level and age at first sex. Cross sectional data will be analyzed usingSTATA 11.0 software package. Separate male (N=403) and female (N=256) datasets weremerged into one dataset (N=659). This was done in order to increase the explanatory powerand enhance the interpretability of results. The analysis focuses on three outcome variables:HIV risk perception, worry over HIV and protective behaviour (condom use) in response toAIDS. Firstly, a bivariate analysis stratified according to gender will be conducted in orderto come up with the respondents' descriptive statistics indicating their backgroundcharacteristics. This will be followed by multivariate analyses which comprise twomultinomial regression analyses which are meant to test the association between likelihoodof infection

and worry in relation to the socio demographic variables. Lastly, correlates ofcondom use will be explored in a binary logistic regression analysis. The following sectiondescribes in detail the methods of analysis including definition of variables.

3.3 Research Questions

This dissertation aims to answer the following questions in order to understand the association between socio demographic characteristics and HIV risk perceptions, worryover HIV and condom use. Is there an association between likelihood of infection (perceived risk) and socioeconomic factors (age at first sex being the variable of interest).

- i. Is there an association between worry over contracting HIV (perceived risk) and socio economic characteristics (age at first sex being the variable of interest).
- ii. Is there an association between condom use and socio economic characteristics (age at first sex being the variable of interest).

The first and second questions are aimed at examining if HIV risk perceptions and worryover HIV infection are associated with or determined by socio demographic covariates. Thethird question will be used to examine if protective behaviour (condom use) among young people is associated related socio demographic characteristics. In all the models the variable of interest is the age at first sex.

3.4 Definition of Variables

The outcome variables inthis study are HIV risk perceptions which are measured bylikelihood of HIV infection, worry over HIV infection, while protective behavior will bemeasured by condom use at last sex. Two multinomial logistic regression models will berun separately in order to examine if socio demographic characteristics are associated withHIV risk perceptions and with worry over HIV infection among adolescents in Malawi.The third model will test the association between condom use at last sex and sociodemographic characteristics of the respondents and will take the form of a binary logistic regression since the responses on condom use were binary (Yes=1 and No=0).

In the first two multinomial analyses the covariates included are the respondents' background characteristics: age of respondent, gender, marital status, religion, province, educational attainment and age at first sex. There may be other crucial factors in placewhich may shape the adolescents' attitudes over perceived risks and these may act ascontrol factors in the models. Included in the two multinomial logistic regression analysisare the following potential confounding factors; AIDS-related knowledge, HIV test and personal acquaintance with a person who has died of AIDS.

Correlates included in the binary logistic regression whose outcome variable is condom useat last sex include; age of respondent, gender, marital status, religion, province, educationalattainment and age at first sex. In addition to the socio demographic correlates, the binaryregression included HIV risk perceptions; worry over HIV infection, HIV test status sincethese factors could potentially predict condom use among adolescents. This is confirmed byliterature which state that those individuals who perceive some likelihood, of HIVinfection, and HIV positive, are less likely to use protection compared to those whoperceive no risk at all (Oyedokun, and Odimegwu, 2003).

3.4.1 Outcome Variables

Risk perception measured by likelihood of HIV infection, including worry over contractingHIV, and condom use at last sex were the three outcome variables in this study. Perceptionof risk may be considered to be the first stage towards behavioural change from risk-takingto safer behavior, and perceived risk of getting AIDS could have important implications forhealth if the perceptions are rational and lead to a willingness to avoid risky behavior (Akwara *et al.*, 2003).

3.4.1.1 Perceived Likelihood of infection

The MDICP asked individuals' perceptions of being infected with HIV in the 2004, 2006and 2008 survey rounds. In 2004, the wording of the questions was: 'How likely are you of being infected with HIV now?', 'How likely are you of being infected with HIV in the future?' and, 'How likely is your partner of being infected with HIV?' The responses to allthe questions were no likelihood=0, low=1, medium=2 and high=3. Due to the small totalsample size (N=659) and low response rates for these individual perception questions, this dissertation computed a single measure of perceived likelihood of infection. This was done by pooling

individualresponses for the 3 likelihood variables (current perceived likelihood of infection, future likelihood of infection and spouse likelihood of infection) into a single likelihood measure recoded as 'perceived likelihood of infection' with the same scale (0-3) and coding of responses.

3.4.1.2 Worry about HIV/AIDS

The wording of the question was: 'How worried are you to catch AIDS?' and had thefollowing responses: (not worried at all=0), (worried a little=1), and (worried a lot=2). This question will be tested to establish if socio demographic characteristics are associated with worry over HIV infection among the adolescents.

3.4.1.3 Condom use at last sex

Condoms remain the most effective protection against HIV and other sexually transmitted infections (STIs) for sexually active young adults. It is crucial to examine the association between condom use and background characteristics which could influence HIV prevalence levels among adolescents. Data from household surveys show that the proportion of youngpeople using condoms is still quite low even when they have sex with people who are not their regular partner (Bertrand and Anhang, 2006). In a 2004 national study in Malawi, apart from limiting sexual partners and abstinence, condom use was cited by 57 percent of women and 76 percent of men as the third protective strategy in avoiding AIDS (MDHS,2004). Further literature indicates that where condom use is not very common amongmarried couples or those in regular relationships and because of problems of interpretation (Dare & Cleland, 1994), information on condom use was restricted to the last sexual encounter. This dissertation used 'condom use at last sex' as a measure of protective behaviour where the question 'Did you use a condom at your last sexual encounter?' wasasked with the binary responses Yes=1 and No=0.

3.4.2 Covariate Factors

The covariates included in the multivariate framework include the respondents' background characteristics: age of respondent, gender, marital status, religion, province, educationallevel and age at first sex, AIDS knowledge and HIV test. These are the explanatory orpredictor

variables, which were used to predict the outcomes of risk perceptions andprotective behaviors of the adolescents. The following section describes the associatedcovariates. The social cognitive theory states that in order to understand sexual riskbehavior, one needs to consider the interactive effects of factors at three levels: within theperson, within his or her proximal context, and within the distal context (Eaton *et al.*,2003). The correlates selected for the analyses fall within the personal context (age, age atfirst sex, gender, religion, education, and AIDS knowledge) and the proximal context(province).

3.4.2.1 Province or Area of Residence

Some norms and values in communities in the 3 different provinces in Malawi have aneffect in reduction or increase of HIV infection, and could predict HIV risk perceptionsamong the adolescents. According to Munthali et al., (2004), some cultural practices incertain provinces can impact on STI/HIV/AIDS transmission and other sexual andreproductive problems on adolescents. In the Southern region of Malawi, boys and girlsundergo initiation ceremonies, where they are told that they are men or women, and canindulge in sexual behavior. Such prevailing cultural contexts may have a strong impact adolescents' HIV risk perceptions or worryover contracting HIV. Furthermore funeralcleansing rites in the Southern province, where the wife of the deceased has to have sexwith another man in order to remove have a strong role in shaping the sexualbehaviors of young people in Malawi spells||, (Munthali et al., 2004). In this study, the respondentswere evenly distributed per region (province), with 32.5 percent of respondents from theSouthern region, 31.8 percent from the Central region, and 35.7 percent from the North.National survey results in 2006 indicate that women in the Southern Region started havingsex at an earlier age than women in the Central and Southern Regions (MDHS 2006). This variable was coded as follows; Southern region=1, Central Region=2 and Northernregion=3.

3.4.2.2 Educational Attainment

There is a strong association between the respondent's educational level and knowledge of AIDS prevention. This means that being in school and the level of education attained play amajor role in adopting condom use for AIDS prevention and in shaping an individual's riskperception. However some researchers argue that modern education exposes adolescents to the school environment that enables them to interact more with the partners of the oppositesex, increasing their exposure to sexual activity (Meekers, 1994). Furthermore, education is positively correlated with the likelihood of having comprehensive knowledge about HIV/AIDS (MDHS, 2004), which could in turn influence the levels of worry, condom useand risk behaviour. In order to examine if education plays a role in shaping adolescents' perception risks and the way they protect themselves from contracting HIV, this study usesthe question 'What is the level of education attained|| which had responses tertiary=1,primary education=2, and secondary education=3.

3.4.2.3 Religious affiliation

Religion in Malawi is closely related to culture, and may be a predictor of HIV riskperception and protective behaviour among adolescents. The three provinces of Malawivary a lot with regards to culture, which is strongly correlated to religion and which couldhave an effect on adolescents' risk perceptions and condom use. Christian valuesemphasize on monogamy as opposed to Moslem values that allow polygamy, which in turncould influence condom use and levels of worry and self-assessment with regards to risk ofinfection. Furthermore, church or mosque attendance may be a proxy of measuring how active or passive a religious member is. Some scholars argue that church attendance elevates the odds of having first sex among adolescents, perhaps because the church settingprovides an opportunity for adolescents to meet potential partners (Haas and Forste, 2002).

Furthermore, there is conventional wisdom among Malawians that one stands a higher riskof HIV infection by marrying a non-religious partner, where most people place trust inreligious people and feel that they could not risk HIV infection as much as they could with a non-religious person. Religious affiliation was measured through the question; what church do you belong to-with 13 categories of churches including no religion option.

For the purposes of the multinomial regression the various denominational responses werecollapsed into 3 broad categories namely: Christian=1, Moslem=2, and No religion=3. Thesame various categories were recorded into binary values of Christian=0, Non-Christian=1so as to be fitted into the binary regression analysis.

3.4.2.4 Current Age

Current age is an important variable mainly because it usually an indicator of an exposureto risk behavior and a strong determinant of ever had sex. Age may make a difference in the likelihood of young people having sex, the older they become the more likely they areto be involved in sexual activity and the more they are exposed to HIV infection. Age alsoaffects condom use; where the younger the adolescents are the less likely they are going touse a condom. Munthali *et al.*, (2006) argue that younger male and female teens are likelyto be more exposed to the risk of sexual coercion than the older adolescents and that age isvery likely to be related to knowledge of risks and means of protection. Since the agevariable was continuous, it was recoded into 3 categories as follows: 10-15 years=1, 16-20=2 years, and 21-24 years=3.

3.4.2.5Age at first sex

Age at first sex is important in predicting HIV risk perception since it may be a positivefactor in understanding HIV risk perceptions, which may shape the young people's futureprotective behaviour. Percentage of young women and men aged 15-24 who have hadsexual intercourse before the age of 15 was declared an important indicator under theUNGASS⁴ Declaration of Commitment on HIV/AIDS indicator (UNAIDS, 2008). Atabout the same time this study took place, national results indicate that 15 percent of women and 14 percent of men aged 15-24 had sex by age 15, with young men's sexualdebut having occurred at a slightly earlier age than women. Furthermore, those who startedhaving sex at an early age (before age 15) had a higher HIV prevalence than those with alater sexual debut (18 percent compared with 15 percent or lower (MDHS, 2004).

Researchers argue that if an adolescent engages in early age at first intercourse, they arelikely to end up having multiple partners and also, engaging in unprotected sexualintercourse (Gueye *et al.*, 2001). Most studies carried out on young people's sexualbehaviour suggest that the younger the age at which first intercourse occurs, the greater thefrequency of sexual intercourse and there is increased likelihood that it will be unprotected and resulting in a higher risk of pregnancy and HIV infection (L'Engle *et al.*, 2006; Koenig*et al*, 2004; Smith and Watkins, 2005). This shows that age at first sex is an important predictor of risk perceptions and condom use among adolescents. Like in many othersimilar studies the age at first sex is measured through the question 'How old were you thefirst time you had sex?', and

⁴ United Nations General Assembly Special Session

the responses in this case were continuous data. For thepurposes of the analysis, the variable age at first sex was used as a continuous variable in multinomial regression analysis with the following stratifications; 'Less than 10years'=0, '11-15 years'=1,'16-20 years=2', and '20-24 years=3', and 'Refused to answer=4'. For the binary regression analysis the age groups were coded as dichotomous variables as follows; 'up to 15 years=0' and '16-24 years=1'.

3.4.2.6 Confounding Factors

HIV/AIDS exposure was assessed by asking if the respondent personally knew someonewho had died from HIV/AIDS, knowledge of AIDS information and HIV test status. Thisstudy includes these three as possible confounding factors that were observed in the data. These covariates are all important in predicting HIV risk perceptions since they reflectone's knowledge of the dangers of HIV and knowing someone who has died of AIDS mayaffect personal safety and protection, while knowing one's HIV status may determine levelsof worry and risk as well as condom use and consistency.

3.4.2.7Knowledge about AIDS

Having accurate knowledge about AIDS is a precondition for evaluating adequately one'srisk knowledge of HIV/AIDS; hence AIDS related knowledge might play an important rolein determining the effect of age at first sex on risk perceptions and protective strategies among adolescents. For those who are HIV infected, knowledge of their status allows themto better protect their sexual partners, to access treatment, and to plan for their future(MDHS, 2004). AIDS-related knowledge was measured through the question 'Have you ever seen AIDS information?', and had responses 1=Yes and 0=No.

3.4.2.8Knowledge of Someone who died of AIDS

There is research evidence that HIV/AIDS-related knowledge is significantly related toAIDSrelated worry, perceived risk, and behavioral change (Gregson *et al.*, 1998;Klepinger et al., 1993; London and Robles 2000; cited in Smith and Watkins, 2003). Someresearchers argue that having a personal tie to someone with HIV/AIDS could makeadolescents more cautious in their own risk and protective behaviors (Munthali *et al.*, 2006). Furthermore, literature states that acquaintance with someone who has died or isliving with AIDS is a proxy for the levels of AIDS morbidity and mortality in thecommunity, which is hypothesized to shift the perception of AIDS from a distant to animmediate threat (Macintyre *et al.*, 2001). Recent research results reveal that knowledge of someone who died of AIDS was correlated with an increased-HIV-risk-perception (Kenyon*et al.*, 2010). It can thus be concluded that knowing someone who has died with AIDS alsoincreases the perception of personal risk, presumably because it may remove the illusion of safety within one's social circle, hence the importance of including this variable as apredictor of risk perception and protective strategies. This personal acquaintance with aperson who has died of AIDS was measured through the question —Is there anyone who hasdied of AIDS in the past 3 years, and had responses Yes=1 and No=0.

3.4.2.9HIV Tests

Knowledge of HIV status helps HIV-negative individuals make specific decisions to reduce the risk of contracting the disease and to increase safer sex practices so they can remaindisease free. For those who are HIV infected, knowledge of their status allows them tobetter protect their sexual partners, to access treatment, and to plan for their future (MDHS,2004). With regards to condom use, literature indicates that young men who have beentested for HIV are significantly more likely than those who have not been tested to haveused a condom during their most recent sexual intercourse (Kenyon et al., 2010). For analyses, bothmultinomial and binary regression HIV test status had the correspondingresponses were No=0 and Yes-1 and was measured through the question 'Have you everbeen tested to see if you have the AIDS virus?'.

3.5 Methods of Analysis

The sample used in the study is restricted to sexually experienced young women and menaged 15 to 24 (256 females and 403 males). Smith and Watkins, (2005) state that although the sample is not strictly representative of Malawi, comparison with a nationally representative survey shows that it is close to being nationally representative of the rural population intercourse. Descriptive statistics and inferential statistical models will becomputed in order to examine if there is an association between socio demographic characteristics and HIV risk perceptions and condom use among adolescents.

Three modelswill be run topredict perceived HIVrisks and protective behaviours, model 1 and 2 employing a multinomial regression and Relative Risk Ratios (RRR) as output and model 3 employing a binary logistics regression.For the binary model, odds ratios output were used for ease of interpretation, and results arepresented as odds ratios (OR) with 95 percent confidence interval (CI).

3.5.1 Descriptive analysis

Firstly, descriptive statistics to obtain the frequencies of the background characteristics of respondents (age, marital status, mean age at sexual debut, education, economic status ofhousehold) will be computed. Secondly, variables which measure risk perception will becomputed; likelihood of infection and levels of worry as well protective strategies againstHIV. Lastly, descriptive statistics indicating knowledge of HIV/AIDS and knowledge of someone who died of AIDS variables will be computed.

3.5.2 Multivariate Analysis

Three models will be fitted in this analysis. The models will examine the relationshipsbetween socio-demographic and cultural factors and the likelihood of infection; worryabout HIV/AIDS; and condom use at last sex. The socio demographic correlates are province, gender, religious affiliation, educational attainment, age, age at first sex. Due tothe uneven distribution of responses on marital status, the 3 models have not included marital status. In all the models potential confounding factors include knowledge of someone with HIV, knowing someone who had died of AIDS, and HIV test status. Thepredictor variable of interest in this dissertation is the age at first sex while the outcomevariables are HIV risk perceptions, worry over HIV and condom use at last sex. Due to thesmall separate sample sizes for both males (N=405) and females (N=256), which yieldedstatistically insignificant results, the two datasets were merged in order to increase the explanatory power of the analysis and also to enhance interpretability of the results. Thefollowing section explains the three models that will be used in more detail.

3.5.3 Multinomial logistic regression

Multinomial logistic regression is used where a categorical dependent variable has more than two categories. The multinomial regression model allows for multiple outcomes that are nominalin nature, rather than ranked in some meaningful ways. The multinomial regression model breaks the regression up into a series of binary regressions, comparingeach group to a baseline group and output is interpreted as Relative Risk Ratios.

3.5.3.1 Model 1: Is Likelihood of HIV Infection Associated with SocioDemographic Characteristics.

In model 1, likelihood of infection consists of 4 groups: high likelihood, mediumlikelihood, low likelihood, and no likelihood at all. High likelihood is set to be the baselinegroup. Multinomial regression will assess the odds of low likelihood of infection versussocio demographic variables; no likelihood at all socio demographic variables; and mediumlikelihood versus demographic economic variables in relation to high likelihood ofinfection (base outcome). The multinomial regression model mainly has Relative Risk Ratios(RRR) thatwould predict the odds of being a low likelihood perceiver as compared to high likelihood perceiver, and the odds of being no likelihood at all perceiver as compared to highlikelihood perceiver. The general model for multinomial regression is denoted by:

$$pr(y_i = m | \mathbf{x}_i) = \frac{\exp(\mathbf{x}_i \boldsymbol{\beta}_m)}{1 + \sum_{j=2}^{J} \exp(\mathbf{x}_i \boldsymbol{\beta}_j)}$$

Models for each dependent variable (Likelihood of infection) can be written as:

Pr. $y = 1/X = 1/1 + \exp(\beta 1X) + \exp(\beta 2X)$ (first category, high likelihood) Pr. $y = 2/X = \exp(\beta 2X)/1 + \exp(\beta 1X) + \exp(\beta 3X)$ (second category, low likelihood) Pr. $y = 3/X = \exp(\beta 1X)/1 + \exp(\beta 2X) + \exp(\beta 3X)$ (third category, no likelihood at all) Pr. $y = 4/X = \exp(\beta 1X)/1 + \exp(\beta 2X) + \exp(\beta 3X)$ (Fourth category, no likelihood at all) (The first category is the reference group is *high likelihood*)

3.5.3.2 Model 2: Is Worry over HIV Infection Associated with SocioDemographic Characteristics

In model 2, worry over infection was stratified into 3 groups: worried a lot, worried a little, and not worried at all. Worried a lot is set to be the baseline group. Multinomial regressionwill assess the odds of low levels of worry over infection versus high levels of worry, andnot worried at all versus high levels of worry with respect to the socio demographic variables. The multinomial regression model would also produce Relative Risk Ratios (RRR)that would predict the odds of being a low levels of worry adolescent as compared to high levels of worry adolescent's vis-à-vis socio demographic variables. The general model for this second multinomial regression is denoted by:

$$pr(y_i = m | \mathbf{x}_i) = \frac{\exp(\mathbf{x}_i \boldsymbol{\beta}_m)}{1 + \sum_{j=2}^{J} \exp(\mathbf{x}_i \boldsymbol{\beta}_j)}$$

Models for the dependent variable (worry over infection) can be written as:

Pr. y =1/X = 1 / 1 + *exp* (
$$\beta$$
1X) + *exp* (β 2X) (first category, worried a lot)
Pr. y =2/X = *exp* (β 2X) / 1 + *exp* (β 1X) + *exp* (β 3X) (second category, worried a little)
(The first category is the reference group-worried a lot)

3.5.3.3 Model 3: Is Condom Use at Last Sex Associated with SocioDemographic Characteristics

This model used a logistic regression analysis since logistic regression is well suited fordescribing and testing hypotheses about relationships between a dependent (dichotomous)outcome variable and one or more independent (explanatory) or predictor variables. In thiscase, binary logistic regression model has been used to test the association between condomuse (dependent outcome variable) and socio- demographiccharacteristics including riskperceptions and worry over HIV. Focusing on the variable of interest, age at sexual debutas an independent variable, analysis has been done in order to examine the associationbetween condom use at last sex and demographic characteristics including HIV riskperception among young people in Malawi, i.e. examining if a change in the demographicattributes sets a precedence among young peoples' thinking on condom use.

In this model the equation will be in the form:

ln {Pi /1- Pi}= α + β 1*x*1*i* + β 2*x*2*i* ++ β kxki Where, Pi /1- Pi is the odds ratio, i=1, 2 ... n *Pi* represents the probability of occurrence of the dependent variables (condom useat last sex) *x*1*i*, *x*2*i* and *xki* represent the independent variables (age, place of residence,educational attainment, and age at first sex, religion and likelihood of infection). β 1... β *k* represent the slopes of the variables *x*1...*xki* respectively.

In this case condom use was measured with a dichotomous variable indicating whether therespondent had used a condom at the last penetrative sex or not. All the demographic variables of interest have been re-coded into dichotomous variables, e.g. the dichotomous variable 'age at first sex' is defined as follows: value of 0=below 15 years and 1=16-24 years. The cutoff point has been determined by the mean age at first sex which is 14.9 years. For ease of interpretation, odds ratios which are equal to exp^5 (B) were used. An oddsratio of 1.0 indicates that there is no relationship or association between the independent and dependent variables. The goodness of fit (Chi-square) is the fit of the observed values (Y) to the expected values. In such analysis, if the p-value for the overall model fit statistic is lessthan the conventional 0.05 then there is evidence that at least one of the demographic characteristics e.g. age at first sex contributes to the prediction of the outcome (Long,1997), which is condom use at last sex.

⁵ Exponential of the Beta coefficient

3.5.4 Limitations

Firstly, literature states that people hesitate when asked about the most intimate area of their lives, their sexual behavior and will often report on it inaccurately (Poulin 2010). Thispaper makes conclusions on information on age at first sexual encounter that was based exclusively on respondents'self-reports. This data collection method often has limitationsthat are attributed to the tendency for people to under-report socially unacceptablebehaviours (e.g. having multiple sex partners) and to over-report socially acceptablebehavior (Malawi Behavioural Surveillance Survey, 2006; Kaestle, 2005). Smith and Watkins (2005) argue that self-reports may be particularly problematic for behaviors that are targeted (or encouraged) by government prevention campaigns. Recent results from astudy in Malawi indicate that a significant fraction of young women who claimed in thesurvey to have never been sexually active affirmed sexual experience during the in-depthinterview, fielded shortly thereafter (Poulin, 2010). Other researchers agree that 'sexualbehaviours rely on sensitive self-reported information, which is susceptible to misreporting. In particular, it is believed that men tend to over report, whereas women may underreport sexual behaviors' (Nnko, et al., 2004; cited in Clark, Poulin and Kohler, 2009: p32).

Research indicates that if early sexual experience is misreported in the light of campaignspromoting abstinence for youths, two types of reporting error are expected: young sexuallyactive respondents may deny having ever had sex, and among older respondents (many ofwhom would be married and therefore not need to deny sexual activity) there may be atendency to report that first sex had occurred at a later age than it had, particularly if firstsexual encounter was with someone other than their spouse (Poulin 2010). Furthermore, some researchers argue that age may be misreported by some respondents who do not knowtheir exact age, or misreporting under the influence of interventions (Zaba, 2004). InMalawi HIV prevention programmes strongly promote postponement of first sex andteenagers may become reluctant to report sexual activity (Poulin 2010). To

substantiate thearguments, Mensch et al., (2008) compare self-reports of ever having sex with STIbiomarkers for young women in rural Malawi and find that approximately 8–10 percent of women who claim never to have had sex are infected with an STI. Rosenbaum (2006) citedin Poulin (2010) found out that in a sample of adolescents who reported having had sex inan initial interview claimed to be virgins in a follow-up interview after taking virginityoaths. However, McGrath et al, (2008) argue that generally, sexual behaviour surveys are subject to social desirability bias and inconsistent reporting of age at first sex for otherreasons. The other limitation is that the study lacks longitudinal data to allow a more indepth analysis and examination of a causal effect. Instead, analysis is based on one cross-sectional survey in Malawi, with independently drawn samples between men and womenand which do not produce significant results.

Lastly, as a result of the study's cross sectional design, no causal conclusions can be drawnabout whether factors associated with age at first sex predict HIV risk perception amongyoung people in Malawi.. It was not possible in this study to use pooled data from the 1998,2001, 2004, 2006, 2008 datasets since after thorough exploration and validation of thedatasets, there was an indication that the independent variable of interest (age at first sex)was available only in the 2001 and 2004 datasets only, but with a very small sample of thesame respondents interviewed in both years, since a new adolescent sample was added tooffset sample attrition from the 2001 sample. Furthermore, sample attrition and marriagesafter each wave resulted in addition of new respondents in the households, which make itdifficult to compare the same respondents over time.

In summary this chapter describes in detail the sampling design, study area, data collectionmethods, analytical strategy and limitations of the study. This background to the researchdesign is meant to give the reader an insight into the analytical strategy, as well as the imitations in the study into consideration when interpreting the results.

CHAPTER 4

ANALYSIS AND FINDINGS

4.1 Introduction

This chapter presents the findings from the analysis of the cross sectional data from the2004 wave of ongoing longitudinal surveys of the Malawi Diffusion and Ideational ChangeProject. In the first instance, descriptive statistics are presented in the chapter and inferential statistics are then provided in the second part. The study population includes 403young men and 256 young women in the 2004 adolescent sample of the MDICP dataset. The chapter starts by providing a description of the sample. It outlines the socio-demographic characteristics of menand women including their marital status, level ofeducation, religious affiliation and relative household wealth in terms of assets. The chapterfurther gives a picture of the respondents' sexual behaviours in which age at first sex, number of sexual partners, and condom use at first and last sex and perception of risk of HIV infection are examined. The chapter also provides results from multivariate analysiswhich explain the association between age at first sex and HIV risk perceptions, in which the association between age at first sex among young people, and their perceptions on HIV/AIDS are highlighted. Data on some of the important sexual and risk perceptioncharacteristics were small, hence the exclusion of factors such as consistency of condomuse, number of former partners, frequency of sex etc. The following are the risk perceptions that were valid with regards to data: worry about catching HIV/AIDS, likelihood ofcatching AIDS, condom use with first partner as well as with a spouse, perceptions oncircumcision effect on HIV, and perceptions on whether HIV can be transmitted with onetime sex or not. Finally, based on the multivariate analyses, the chapter discusses the resultsand makes an inference on whether there exists an association between socio demographiccharacteristics and HIV risk perceptions among young men and women in the study area.

4.1.1 Socio-Demographic Characteristics

In order to make meaningful interpretation of the young men and women's sexualbehaviour with regards to whether socio demographic characteristics do set precedence

on their sexual behaviours, it is important to first understand the respondents' social and demographic characteristics. This section gives a snapshot of the social and demographiccharacteristics of the adolescent sample from the three regions of Malawi in 2004. Thisstudy compares some of the findings to the results of the 2004 Malawi Demographic HealthSurvey (MDHS) since it is the only nationally representative survey which was conducted in the same year (2004) as the MDICP study whose data this study makes an inference on.

Social and Demographic Characteristics		Males (N=256)		Females (N=403)		All Respondents	
Variable	Response	N	Percent	Ν	Percent	Ν	Percent
Province	Southern	133	33	77	30.1	210	31.9
	Central	128	31.8	85	33.2	213	32.3
	Northern	142	35.2	94	36.7	236	35.8
Age of respondents*	Median Age	16.9			18.7		
	10-15 years	56	14.9	65	28.1	121	20
	16-20 years	240	64	156	67.5	396	65.3
	21-24 years	79	21.1	10	4.3	89	14.7
In School**	No	170	42.2	87	58.4	257	39
	Yes	233	57.8	169	66	402	61
Highest educational level	Primary School	302	74.1	198	77.2	500	76.1
	High School	101	25.9	54	22.8	155	23.6
	Tertiary	0	0	2	0	2	0
	None	0	0	2	0	2	0
Marital Status**	Separated/widowed	25	6.1	8	2.3	33	4.9
	Never married	378	93.9	248	96.9	626	95.1
Religion**	Christian	301	74.7	171	71	472	73.3
	Muslim	83	20.6	49	20.3	132	20.5
	no religion/other	19	4.7	21	8.7	40	6.2
House Material (Wall)	Mud/Sun brick	229	57	134	55.5	363	56.4
	Firebrick	174	43	107	44.5	280	43.6
House Material (Roof)	Metal	86	21.4	66	27.2	152	23.6
	Thatch	317	78.6	175	72.8	491	76.4

Table 4.1 Distribution of Selected Social and Demographic Characteristics

*p<0.001

**p>0.05 Note: Religion and house material (15 missing cases for females),.
The results show that the total respondents were evenly distributed in the 3 provinces of Malawi, with about a third of the sample in each region for both males and females. The distribution of males was 32.5 percent in the Southern region, 31.8 percent in the Central 35.7 percent in the Northern region. A similar distribution is evident among females;30.1 percent in the Southern region, 33.2 percent in the Central and 36.7 percent in the Northern region.

Further results reflect a youthful population under study, where the median age for theyoung men is 16.9 years and 1.8 years younger than the median age of the young women(18.7 years). The majority of respondents were aged between 16-20 years (65 percent for both sexes) with more females (67.5 percent) than males (64 percent) and followed by those aged 10-15 year (20 years for both sexes). The least respondents were aged 21-24 years (14.7 percent for both sexes) and almost one in five(21.1 percent) compared to 4.3 percent of females.

The results further indicate that the majority (95.1 percent) of the respondents were nevermarried, with more respondents among women (96.9 percent) than men (93.9 percent). These results are not surprising given that the design of the study was aimed atinterviewing the never married adolescents. It is quite surprising that a proportion of youngmen (6.1 percent) and young women (2.3 percent) were already separated or widowed atthe time of the study. These results imply that 9 out of ten of the young men andwomen in the sample were never married, which was expected in the study since the aim was to interview the never married adolescents.

Religion plays a vital role in shaping young peoples' attitudes and behaviours as well astheir HIV risk perceptions and protective strategies. It is common belief among Malawiansthat religion a major role in combating HIV/AIDS hence the Faith plays BasedOrganisations that exist and work in sensitising the population on HIV/AIDS (Munthali etal., 2006). It is also believed that certain religious beliefs can possibly make young peoplemore prone to contracting HIV as with case of polygamy among Muslims in Malawi. Theresults in this study indicate that seven out of ten of young men(71.0 percent) and young women (74.7 percent) were Christians compared to Moslems whocontributed to 20.6 percent of young men and 20.3 percent of the young women. The minority of therespondents contributed to 4.7 percent of young men and 8.7 percent of women who had no religion or belonged to other religion category. Otherreligion in Malawi includes traditional religion or cult such as '⁶nyau' and those who claim to have no religion. The Malawi Demographic and Health Survey (2008) indicates that there more Christians in Malawi (86 percent) compared to 13 percent of Moslems, 0.9 percent who had no religion (MDHS, 2010).

The respondents were also asked about their educational background. Much as education isa key determinant of the lifestyle and status an individual enjoys in a society, it also affectsmany aspects of life, including demographic and health related behaviour (MDHS 2006). Studies have consistently shown that educational attainment has strong effects onreproductive behaviour, contraceptive use, fertility, infant and child mortality, morbidity,attitudes and awareness related to family health and hygiene (MDHS 2006). The results inthis study reflect that more of the young men (58.4 percent) and young women (66.0percent) reported being in school at the time of the study, compared to four out of every tenyoung men (41.8 percent) and three out of every ten young women (33.9 percent) whoreported not attending school. Furthermore, respondents were asked on their level ofeducation attainment, the majority of them had some primary school (74.3 percent among males and 77.3 among females), compared to 25.7 percent and 21.1 percent of females whohad some secondary education. Very few of the males (1.6 percent) and none of femaleshad tertiary education.

The type of material used for walls and roofing of dwellings is may be a proxy indicator of the economic standing of the household and reflects the potential exposure to disease-causing agents (MDHS, 2006). The results from this study indicate that slightly above halfof the young men (57.0 percent) and young women (55.5 percent) lived in mud or sun brickmade houses, compared to 43.0 percent of young men and 44.5 percent of young womenwho lived in fire brick houses. The majority of the respondents had houses mostly roofed with thatch (76.6 percent of men vs 72.8 percent of women) compared (21.4. percent of men vs 27.2 percent of women) who had metal roofs. These results imply that at the time of this survey, the majority of these young men and women in Malawi lived in houses madeof traditional material, and that young women were more likely to live in fire brick housesand metal roofs compared to men (usually without electricity); nine percent of householdsin Malawi had electricity at about the same time (MDHS, 2010). The Welfare MonitoringSurvey (2009) in Malawi showed that mud bricks (31 percent) and burnt bricks (49percent) were the

⁶ Nyau are a form of masquerades cult who worships at the graveyard. They are mostly found in the Central Region.

most common materials used for walls and that 60 percent of householdsused grass for roofing their dwellings and 40 percent used iron sheets as roofing material (WMS, 2009).

4.1.2 Relative Household Wealth

This following section looks at the household wealth as measured by non-monetary assetownership. Ownership of goods provides proxy measures of the economic status and healthstatus of the respondents' lives. Respondents were asked about their household's ownership particular durable goods and Table 4.2 highlights the results.

Household Assets	Females	Males	(n=403)	All Respondents		
	Ν	Percent	Ν	Percent	Ν	Percent
Paraffin lamp	137	53.52	206	50.37	343	52.0
Bicycle	131	51.17	234	57.21	365	55.4
Radio	190	74.22	306	74.82	496	75.3
Television	11	4.3	22	5.38	33	5.0
Bed with mattress	99	3.8	111	27.14	210	31.9
Sofa Set	57	22.3	61	1	118	17.9
Table and chair(s)	132	51.6	172	4	304	46.1

Table 4.2: Percentage of Ownership of Household Assets,

Poverty can also be measured in non-monetary terms and the above table reflects the wealthdistribution of the respondents in non-monetary dimensions. The most commonly ownedasset by the households among young men and women in 2004 was a radio (74 percent)followed by almost above half of the households owning a bicycle (54 percent on average).Furthermore, about half of the young men and young women owned a lamp, with less thana third of households among young men (27.4 percent) and about a third of young women(36.8 percent) had a bed with a mattress, 42.1 percent of men and 51.6 percent of womenhad a table with chairs, while a minority (less than a quarter) of both sexes had a sofa set.The least owned asset among the respondents 'households was the television (5.4 percent of boys) and 4.3 percent of girls. At about the same period, nationally, 62 percent ofhouseholds owned a radio, four in ten households owned a paraffin lamp, and only 5percent) of households owned a television (MDHS, 2004). The implication of these results that at least 7 out of ten households in this study had access to electronic media, morethan half had bicycles as the most common type of vehicle and means of transport, withonly one

in twenty households having access to visual mass communication (television). Although data on access to sanitation was not included in this chapter, at about the same time, MDHS results indicate that use of traditional pit latrinesaccounted for 79 percent of all households (MDHS 2006). In the same year that this studywas undertaken, based on the poverty lines it was found out that 52.4 percent of thepopulation in Malawi was poor (Second Integrated Household Survey, 2005).

4.1.3 Family Planning

Contraceptive Use		Females(N=256)		Males (n=403)		All Respondents	
		Freq	Percent	Freq	Percent	Freq	Percent
Ever Used Family	Yes	22	8.6	77	19.1	99	15.0
Planning	No	223	87.1	301	74.7	524	79.5
	Can't Remember	11	4.3	25	6.2	36	5.5
Currently Using	Yes	10	45.5	36	46.8	46	46.5
Modern Family Planning	No	12	54.5	41	53.2	53	53.5

Table 4.3: Showing Contraceptive Use by Gender

With regards to modern family planning, the results indicate that the percentage of thoseyoung women and men who reported ever using modern contraceptives was below 20percent, indicating a low modern contraceptive prevalence rate. Of those who reported everusing the modern contraceptives, the results indicate that 45.5 percent of the women and46.8 percent of the men reported currently using the modern methods at the time of thesurvey. Furthermore, the higher percentage (46.8 percent) of the boys who reported currentlyusing contraceptives at the time of the survey could be attributed to condom use meant toprotect them from HIV/AIDS with other partners and not meant to limit their family sizes.Contraceptive prevalence rate in Malawi increased from 26 percent in 2000 (MDHS 2000)to 41 percent in 2005 (MICS, 2006) and 65 percent in 2008 (MDHS, 2010).

4.2 Sexual Characteristics

This section examines some of the patterns of sexual behaviour among the young men andyoung women in the 2004 MDICP dataset. Firstly, it is worth noting that the age at whichyoung men and women initiate sexual intercourse marks the beginning of their exposure toHIV/AIDS and other reproductive risks. The association between the age at first sex andthe HIV risk perceptions in young menand women's future animportant factor toexamine in order to make meaningful, effective and relevant HIV intervention programmes for the youths. Table 4.3 indicates the sexual characteristics of the respondents.

Sexual Characteris	stics	Female	es	Males	•		1
	<u>_</u>	N=256)) 	(N=403	5)	All Kes	pondents
Variable	Response	Freq	Percent	Freq	Percent	Freq	Percent
	Yes	105	41.0	276	68.5	381	57.8
Even had sev	Never	90	35.2	58	14.4	148	22.5
Ever hau sex	Refused/Missing	61	23.8	69	17.1	130	19.7
	Total	256	100.0	403	100.0	659	100.0
	Less than 10 years	1	1.0	35	12.7	36	9.4
	11-15 years	58	55.2	129	46.7	187	49.1
Age at first sex	16-20 years	45	42.9	80	29.0	125	32.8
	21-24 years	1	1.0	32	11.6	33	8.7
	Sub Total	105	100.0	276	100.0	381	100.0
	Yes	41	39.0	161	58.3	202	53.0
Had Sou in the	No	34	32.4	115	41.7	149	39.1
past 12 months	Don't Know/Missing	30	28.6	0	-	30	7.9
	Sub Total	105	100.0	276	100.0	381	100.0
Total Number of	One	40	38.1	91	33.0	131	34.4
sexual partners	More than 1	35	33.3	185	67.0	220	57.7
	Don't	30	28.6	0	-	30	7.9
	Know/Missing						
	Sub Total	105	100.0	276	100.0	381	100.0
Median Age at	(Years)	1	4.9	-	16.1		15.5
first sex							

Table 4.4: Indicating Selected Sexual Characteristics of the Respondents, Men and Women

 2004

The above results indicate that the median reported age of first sex among those whoreported being sexually experienced was 16.1 among young men and 14.9 among youngwomen.A possible explanation could be that young girls begin having sex earlier thanyoung boys due to several reasons of causes of early sexual debut cited in chapter 2. The recent MDHS (2008) reports that the median age at first sex for all males (20-54 years) and females (20-49 years) was 18.6 years and 17.3 years respectively (MDHS, 2010), showing that young men debut later than the young women.

Furthermore, 68.5 percent of the young men reported that they had ever had sex compared to 14.4 percent who reported never having had sex. The results indicate that 41.5 percent ofyoung women reported that they had ever had sexual intercourse, compared to 35.2 percentwho reported that they had never had sex. Given the nature of sensitivity of the questions on sexual behaviour, significant proportion (17.1 percent of young men and 23.8 percent of young women) refused to answer thequestion. Although these results should be interpreted with caution since there was a lot ofmissing data for both females and males, they however give an indication that young menwere more likely to have had sex at the time of the survey compared to the young women.



Figure 3: Reported Age at First Sex, Men and Women, 2004

Table 4.3 indicates that 55.2 percent of those who had sexual experience among theyoung women had early sexual debut (11-15 years) followed by those who had first sex atbetween the ages 16-20 years (42.9 percent), followed by those who had sexual debut after20 years (0.1 percent). The results indicate that 46.7 percent of the young men had sexualdebut between 11-15 years, followed by those aged 16-20 years (29 percent) and those who had sexual debut before the age of ten (12.7 percent). Sexualdebut among the young women was uncommon among the age groups 21-24 years andamong both sexes. These results are graphically depicted in figure 2 above.

These results generally imply that both youngwomen and men were more likely to experience first sex between the ages 11-15 years, andthat young women were more likely to debut earlier than young men. The results also imply that more of both young men and women experienced early sexual debut, followed by those who had first sex between 16-20 years. Some researchers have evidence supporting that age at first sex for most young women

in Africa occurs between ages 17–20years (Pettifor *et al.*, 2009) and that adolescent females in Sub-Saharan Africa tend to havesex at an earlier age than their male counterparts, and thus are at particular risk for HIV,unwanted pregnancy and other adverse outcomes (Biddlecom *et al.*, 2007).

Furthermore, of the sexually active respondents, the majority of young men (58.3 percent)reported that they had sex in the last 12 months compared to young women (39 percent)compared to 41.7 of young men and 32.4 percent of young women who had had sex in the same reference period. 28.6 percent of the young women either refused to answer the question or the data was missing.

Further results indicate that twice as much young men (67.0 percent) as of young women (33.3 percent) had had multiple partners at the time of the study. The results for the young women may be skewed due to the high non response rate (28 percent). The implication of these results could be that young men were more likely than young women tohave multiple partners and to have had sex in the last 12 months. These results may not be interpreted without taking into consideration the fact that there is literature evidence which states that men tend to over report the number of sexual partners while women tend to underreport (Pettifor *et al.*, 2004) and the fact that there was a higher non response or refusal rate among the females.

For the purposes of this study, early sexual debut has been classified as having sex beforethe age of 15 years, and in line with legislative policies in ⁷Malawi. The results imply thaton average about six out of ten (57.5 percent) adolescents aged less than 15 years in the sample had ever had sex at the time of the survey. These could be more given the refusal/non response rate and this should be aworrisome scenario in Malawi if more and more children have sex below the age of 16 years and fall pregnant orcatch HIV/AIDS, especially for the girls who may pass the virus to the newly born baby.

4.2.1 Perceived Risk for HIV

In order for young people to take precautions to protect themselves from HIV, they firsthave to think that they are potentially at risk for becoming infected with HIV (Pettifor *et al*,2004).

⁷The Malawi constitution describes a child as someone who is 16 years and below.

The respondents were also asked about their perceptions towards worry overHIV/AID, their perceived likelihood of infection, their optimism or pessimism about HIV/AIDS, theirperception on whether one could catch the virus by having sex once and if there was an effect of circumcision on HIV/AIDS. Table 4.4 indicates the results;

Perceived Risk Perceptions and Worry Over HIV		Females (N=256)		Males(N=403)		All Respondents	
Variable	Category	N	Percent	Ν	Percent	Ν	Percent
Likelihood of	No Likelihood	176	75.5	233	62.6	409	67.6
HIV Infection*	Low	40	17.2	108	29	148	24.5
	Medium	8	3.4	17	4.6	25	4.1
	High	9	3.9	14	3.8	23	3.8
Likelihood of partner HIV	No Likelihood	74	74.7	115	36.9	189	46
	Low	11	11.1	105	33.7	116	28.2
Infection*	Medium	8	8.1	63	20.2	71	17.3
	High	6	6.1	29	9.3	35	8.5
Likelihood of	No Likelihood	91	47.2	123	63.7	214	55.4
future HIV	Low	61	31.6	53	27.5	114	29.5
Infection*	Medium	30	15.5	7	3.6	37	9.6
	High	11	5.7	10	5.2	21	5.4
How worried are	Not worried at all	107	43.5	169	41.9	276	42.5
you in	Worried a little	50	20.3	79	19.6	129	19.9
HIV/AIDS?	Worried a lot	89	36.2	155	38.5	244	37.6

Table 4.5: Risk Perceptions and Worry Levels among Males and Females, 2004

*p<0.001

Table 4.4 indicates that 7 out of ten young women (75.5 percent) feltthat there was no likelihood at all that that they could be infected with HIV, followed bythose who felt at low risk (17.2 percent) and less than 5 percent who felt at moderate risk(3.4 percent) and those who felt at high risk (3.9 percent). A similar trend is evident among young men; 6out of ten young men (62.6 percent) felt at no risk at all, followed by those who felt lowrisk (29.0 percent) and less than 5 percent who felt at moderate (4.6 percent) or high risk(3.8 percent). These results imply that the adolescents were more likely to perceive no risk of HIV infection at all (67.6 percent) and that young women were more likely to perceive no risk at all thanthe males.

The results also indicate that almost three quarters of young women (74.7 percent)perceived no chances of being infected by their partner, followed by those who

perceivedlow (11.1 percent), medium (8.1 percent) and high risk (6.1 percent). Among young menhalf as much as the females (36.9 percent) perceived no risk of infection from a partner, while 33.7 percent perceived low risk and medium risk (20.2 percent) and high risk (9.3percent). Overall, the adolescents were less likely to perceive no risk of infection from apartner (46.0 percent), males were less likely to perceive no risk of infection than females. It is therefore evident from these results that females felt no threat of HIV infection from a partner than the males. This clearly shows that more girls are of the opinion that there are less chances of beinginfected by the partners they would marry in future, compared to the boys who may bemore cautious and careful about catching HIV/AIDS from their partners.

With regards to one's future infection the results indicate that among the young women, less felt that there were no chances of HIV infection in the future (47.2 percent), compared to 31.6 percent who perceived low risk, 15.5 percent who perceived medium risk and 5.7 percent who perceived high risk. An almost different pattern is evident among men wheremore of them perceived no HIV risk in future (63.7 percent), followed by 27.5 percent whoperceived low future risk, and less than 10 percent who perceived medium to high risk.

Overall results indicate that adolescents were generally more likely to perceive nolikelihood of future HIV infection (55.4 percent) and males were more inclined to perceiveno future risk than females. This may imply that the adolescents were optimistic and hadsome future hope about HIV infection, or it could be that they did not feel the immediatethreat of AIDS and felt it is could be a distant threat. Overall one in four (42.5 percent) of the total respondents were not worried aboutcontracting HIV at all, while 19.9 percent reported being worried a little and 37.6 percentwere worried a lot. While these general results indicate that adolescents were more likely toworry over contracting HIV, gender stratification reveals that females were slightly morelikely than males (43.5 percent versus 41.9 percent) to report not being worried at all, andmales were more likely to be worried a lot than females. Bearing in mind that Malawi has youthful population, and has an HIV prevalence rate of 14 percent (MDHS, 2006) it is worrisome that almost 4 out of ten boys and girls were not worried about catching HIV. Apossible explanation for the low levels of worry at this period could be due to the HIVadvocacy and awareness campaigns in Malawi during which advocacy campaigns wereintensified, which made young people recognize the dangers of the disease. Studies of otherpopulations in sub-Saharan Africa have documented similar levels of worry about AIDS and relationships between characteristics and worry: the married worry more than thesingle (Kengeya-Kayondo et al., 1999; Lindan et al., 199; cited in Smith and Watkins, 2005), women worry more than men(United Nations, 2002), and whereas women worry most about their spouses, men worrymost about their extramarital partners (Kengeya-Kayondo, *et al.*, 1999; Bunnell, 1996;Lindan, *et al.*, 1991 cited in Smith and Watkins (2005).

However the study did not go further into examining the reasons or factors behind theirhope of not contracting HIV in subsequent years taking into consideration the increase innumber of people dying of AIDS within the population at that time, with some respondentseven experiencing HIV related deaths of a relative or someone they knew. Perhaps thiscould be an area for further research: examining individual's hope factors associated withHIV future infection, and determining levels of hope among the adolescents.

In summary, the results above indicate that adolescents were generally more likely to worry over contractingHIV, with females more likely than males to report not being worried at all, and males weremore likely to be worried a lot than females. Adolescents were more likely to perceive nolikelihood of HIV infection with young women more likely to perceive no risk at all compared to their male counterparts.Furthermore, more young women were of the opinion that there were less chances of being infected by theirpartners they would marry in future, compared to the boys who felt at a higher risk ofcatching HIV/AIDS from their partners. Finally adolescents were generally more likely toperceive no likelihood of future HIV infection and males were more inclined to perceiveno future risk than females.

4.2.2 HIV Related Knowledge and HIV Testing

Knowledge about the respondents' condom use, attitudes and practices, as well as familyplanning is very important in understanding their protective behaviour patterns againstHIV/AIDS. Adolescents' views on chances of infection by having a once off sex act, effects of circumcision and their views on whether the AIDS problem was getting better orworse were included in the analysis. These factors are directly linked to self-evaluation and assessment of HIV infection and are crucialin makinginformed decisions about HIV intervention programmes for the youths. The analysis also included an indicator of whether the respondent had been tested for HIV since some researchers argue that HIV test could bean experience that could act as a catalyst for condom use, regardless of the test outcome(Sweat *et al.*, 2000). The following tablepresents selected indicators related to HIV knowledge.

							All	
HIV/AIDS Knowledge		Femal	les(N=256)	Male	s (n=403)	Resp	Respondents	
Variable	Category	N	Percent	Ν	Percent	Ν	Percent	
	Yes	197	77.0	277	68.7	474	71.9	
Ever heard or seen	No	54	21.1	116	28.8	170	25.8	
AIDS Information	Can't Remember*	5	2.0	10	2.5	15	2.3	
Know anyone who died of AIDS	Yes	164	64.1	291	72.2	455	69.0	
	No	89	34.8	110	27.3	199	30.2	
	Can't Remember*	3	1.2	2	0.5	5	0.8	
Do you think AIDS	Better	78	30.5	116	28.8	194	29.4	
problem is getting better or worse	Worse	178	69.5	287	71.2	465	70.6	
Chances of infection by	No Chances	16	6.3	47	11.7	63	9.6	
having sex once?	Chances	240	93.8	356	88.3	596	90.4	
Does circumcision	Has an effect	24	9.4	29	7.2	53	8.0	
affect chances to HIV infection	Does not have an effect	232	90.6	374	92.8	606	92.0	
	Yes	217	84.8	333	82.6	550	83.5	
lested for HIV?	No	39	15.2	70	17.4	109	16.5	

Table 4.6: Showing HIV/AIDS Knowledge and Perceptions

P < 0.05

*can't remember or missing information

The table above indicates that at the time of the study 7 out of 10 adolescents had heard of AIDS information compared to about a quarter who had not and that females were more likely tohave heard or seen AIDS information compared to the males. Furthermore, the majority of respondents (69.6 percent) knew somebody who had died of AIDS, and males were morelikely to know of an AIDS death than females (72.6 vs 64.8 percent). These results implythat there is widespread knowledge about AIDS information; however mere knowledge thatis not comprehensive is not adequate catalyst in behaviour change among adolescents.Probably death of someone due to AIDS may only send a strong message across if the onewho died happens to be a close relative, since their absence and the impact and dangers of HIV can be felt.

Furthermore, the results indicate that the majority of the adolescent boys (71.7 percent) andgirls (69.4 percent) were of the opinion that the AIDS problem was getting worse, probablyimplying that there could be a fatalistic perception with regards to their thinking aboutfuture partner infection. These results do not support the results which show that lessthan half of the young men and women felt they was a chance that they could be

infected with the HIV in future. This probably is an indication that the young men and women werepessimistic about the problem of HIV/AIDS, but still had some hope of not getting future infection themselves and could perhaps be a good yardstick and indicator to consider forintervention programmes among the youth. There is widespread knowledge among young men nowadays that AIDS is spread mainlythrough heterosexual sex. The respondents were asked if having sex once would infect onewith HIV. The results indicates that over 9 out of ten young women (93.6 percent) andmen (88.6 percent) felt that there were chances that one could be infected with HIV byhaving sex once. Only less than 15 percent of the boys and girls felt that there were no chances of infection by having sex once. This may be attributed by the fact that comprehensive knowledge of HIV is now universal among young Malawians, and the fact that HIV is mostly contacted through heterosexual sex is also almost universal. Howeverrecent scholarly debate reiterates that one time sex may not get one infected, but themajority of Malawians still have the fact and strongly believe that one gets infected withone time unprotected sex. The survey also sought to know if abstinence could protect one from HIV and the results revealed that the majority (96.7 percent) of the boys and girls(72.4 percent) were positive. In the same vein 90.9 percent of girls and 99.2 percent of boysperceived that traditional medicine cannot protect one from HIV/AIDS.

There has been a lot of debate worldwide over circumcision having an effect on HIV/AIDS.In Malawi, circumcision is practiced in many communities and often serves as a Recently, male circumcision has been associated with rite of passage to adulthood. lowertransmission of STIs, including HIV (MDHS 2004). The results from this study indicatethat very few young men (9.2 percent) and women (7.1 percent) felt that male circumcisionhad an effect in decreasing or increasing HIV/AIDS, while the majority of boys and girls(9 out of ten) felt circumcision did not have an effect on HIV/AIDS. The study did notexamine the reasons for their varied beliefs, nor did it examine the direction of theperceived effect. National results at about the same time in Malawi indicate that circumcised men had a slightly higher HIV infection rate than men who were notcircumcised (MDHS, 2006). However, some researchers indicate that observational various settings have found consistent, significant, studies in and negative associationsbetween male circumcision and HIV infection (Poulin, 2007). Using the same sample as inthis study, it was found out that when comparing HIV prevalence rates across the MDICP'sthree study sites, the region home to the highest HIV prevalence (8.6 percent) was alsowhere most husbands (80 percent) were circumcised (Poulin, 2007). MDHS results

alsoleave a lot to be desired, if circumcision does reduce HIV infection; they indicate that in2004 circumcised males contributed to 13.2 percent of the HIV positive compared to theuncircumcised (9.5 percent) (MDHS, 2006).

With regards to HIV testing, the results indicate a low testing rate among young men andwomen at the time of the study. The results of this study show that the proportion ofsexually active young people who had not had an HIV test in the preceding 12 months was84.8 percent among young women and 82.6 percent among young men, regardless of theirknowledge of their status. Results of a national survey in the same period in about the sameperiod this survey took place, results show that 25.2% of women and 26% of men age 15-24 had tested for HIV (MICS, 2006), an indication of lower testing levels amongadolescents. Knowledge of the percentage of sexually active young women and men aged15-24 who received an HIV test in the last 12 months and who know their results formsUNAIDS Indicator 5 and WHO Universal Access Indicator 38 (WHO, 2001). Recentresults from the same study area indicate that 80% of unmarried youths reportedwillingness to be tested for HIV along with their partners at the time when they are gettingmarried and that encouraging engaged couples to take one or more HIV tests during themarriage process would greatly reduce the odds of either partner bringing HIV into theunion unwittingly (Clarke, Kohler and Poulin, 2009).

In summary, the majority of adolescents had heard or seen AIDS information as well asexperienced or heard of a death related to HIV. Females were more likely to have heard orseen AIDS information compared to the males, while males were more likely to know of an AIDS death than females. Furthermore the adolescents were of the opinion that the circumcision did not have an effect on HIV/AIDS and that the AIDS problem was gettingworse, while 9 out of ten young women and men felt that there were chances that one couldbe infected with HIV by having sex once.

4.2.3 HIV Prevention and Condom Use

Condom use is one of the preventive strategies against HIV. Forste *et al.*, (2002) argue thatworry and concern about HIV could motivate condom use, and that individuals who believethat they are already infected or would become infected with HIV/AIDS are more likely tobe worried than individuals who perceive their risk of infection as negligible. Furthermore, condom use at first sex is also an important determinant of HIV risk among

young girls andboys since the first encounter may result into pregnancy or HIV infection. Abstinenceversus condom use has been a bone of contention between some Faith Based Organisations⁸in Malawi and most NGOs fighting NGOs, while some individuals in Malawi still optedfor traditional medicine for STI diseases due to embarrassment one would encounter byvisiting the hospitals or clinic. The following section discusses results on condom use, abstinence and other protective measure (traditional medicine).

						All	
HIV Prevention Strategies		Female	s(N=105)	Males	s (n=276)	Respondents	
Variable	Response	Ν	Percent	Ν	Percent	Ν	Percent
Condom use at first sex*	No	38	51.0	72	47.2	110	47.2
	Yes	37	49.9	86	52.8	123	52.3
Condom use at last sex*	No	27	25.7	64	23.2	91	25.4
	Yes	78	74.3	183	66.3	261	74.6
Frequency of	Beginning	13	17.3	24	15.1	37	15.9
Condom Use*	Sometimes	44	58.7	83	52.6	127	54.5
	Every time	18	24.0	51	32.3	69	29.6
Is it acceptable to use a	Acceptable	38	36.2	93	33.7	131	34.5
condom with a spouse?	Not Acceptable	67	67.3	183	64.2	250	65.5
Can Abstinence protect one	No	73	96.7	114	72.4	187	34.5
from HIV?	Yes	2	3.3	44	27.6	46	65.5
Can traditional	Yes	1	1.3	2	0.7	3	1.3
medicine Protect from HIV?	No	74	98.7	156	98.7	230	98.7
	Total	105	100.0	276	100.0	381	100.0

Table 4.7: HIV Preventive Strategies among Males and Females, 2004

*p<0.05 These are responses based on only those who responded to ever having sex and missing cases and 'can't remember/don't know/ missing' were treated as invalid responses (30 females +118 males).

Table 4.6 indicates that condom use at first sex was higher among the young men (52.8percent) compared to the young women (49.3 percent). This could be explained by the fact that some girls may be coerced into having sex, usually with older boys and do not usecondoms, and also that the boys may have over reported on condom use. Some researchersargue that the association between early debut among young women and condom use mayvary according to whether they had had forced sex or not (Pettifor *et al.*, 2009), which wasnot explored in this study. With regards to condom use at last sex, the

⁸The Roman Catholic Church doctrines do not support condom use.

results indicate higher likelihood of condom use among all adolescents (74.6 percent), with young women more likely to have reportedhaving used a condom at last sex (74.3 percent) than young men (66.3 percent). These results are self-reported and should be interpreted with caution, especially in that era (2004)when condom advocacy by NGOs and HIV awareness was at its peak, and also the probability of adolescents' responses being biased. On the other hand, higher condom use may be due to the girls' fear of being pregnant or having a baby of which boys probably may not have to worry much about. Low consistency use of condoms is reflected by the results, whereless than 30 percent of total adolescent boys andgirls reported using a condom every time they had sexual intercourse. About a third 32.3 percent of the young men reported that they used a condom every time compared to 24 percent of young women. Less than 20 percent of both young men (15.1 percent) and women (17.1) reported that they had used a condom only at the beginning of their sexual encounters. This means that although there might be widespread knowledge of condoms, there wasno consistency in the use, which is a crucial factor in the fight against HIV. Taking into consideration that during the time of the study, there was widespread campaign on HIV prevention in Malawi, the higherproportion of condom use could partly be due to respondents' reluctance to admit a passiveattitude towards AIDS. However, an important aspect of this analysis is that only a smallproportion of adolescents reported consistent condom use as a preventive strategy againstAIDS. It should also be noted that the response rate on condom use for the young womenwas low and therefore results for women should be taken with caution.

Condom use among married couples is not widely approved in Malawi, it is almost a taboo.Further results in this study indicate that 67.3percent of young women felt that it was not acceptablefor one to use a condom with a spouse, compared to 64.2 percent of young men who had similar perceptions. This means that about two thirds of boys and girls disapproved condom use with aspouse, implying that the condoms may not widely be used in wedlock. In a national surveyin 2001 and 2004, results indicate that never-married women and men were much morelikely than ever-married individuals to have used condoms (MDHS 2006). This may be truesince using a condom in wedlock may lead one partner especially the wife , to suspectpartner infidelity; moreover it is almost a taboo for a woman in Malawi to even suggestcondom use in wedlock, lest she would be labeled loose⁴ or promiscuous. Lastly, themajority (over 90 percent) of both young men and women felt that one could protect themselves from HIVby abstaining from sex and not through traditional medicine. Females were much morelikely than males to feel that abstinence protects one from HIV and were

equally likely asmen to perceive that traditional medicine could not protect from HIV infection.

4.3 Multivariate Analysis

Three models have been fitted in this analysis. The first model examines the relationshipbetween likelihood of infection and socio-demographic variables including age at first sexas the variable of interest. The second model examines the relationship between worryabout HIV/AIDS and the socio economic characteristics. The third model examines therelationship between condom use at last sex and the socio economic variables.Confounding factors which include knowledge of someone with HIV and knowing someone who had died of AIDS have been included in all the models as controls. Thefollowing section explains the three models that have been used in more detail.

4.3.1 Outcome Variables

There are three outcome variables that are being examined in this study; likelihood of infection, worry over HIV infection and condom use at last sex. Likelihood of infectionwas measured using the 3 questions: 'what is the likelihood that you are infected with HIV?' and, 'What is the likelihood that you may be infected in future' and 'what is the likelihood that your future partner will be infected with HIV? Running multinomial logistic models usingeach question does not yield significant results. In order to enhance the interpretability ofthe results and increase the explanatory power of the model, the responses from the threequestions have been pooled in order to come up with one variable of 'perceived likelihood of ofinfection'. Worry was measured using the question 'how worried are you of contractingHIV/AIDS?' while condom use was measured through the question 'did you use a condom thelast time you had sex?' The following table shows the dependent variables, categories andtheir codes:

Table 4.8Outcome Variable: Categories and Codes

Outcome Variables		
Variable	Category	Code
	No Likelihood	0
Perceived likelihood of HIV	Low	1
Infection	Medium	2
	High	3
	Not worried at all	0
How worried are you in contracting	Worried a little	1
IIIV/AIDS	Worried a lot	2
Condom Use at Last Say	No	0
Condonii Use at Last Sex	Yes	1

4.3.2 Correlates of Risk Perception and Condom Use

The predictor variables or covariates are the respondents' background characteristics: ageof respondent, marital status, religion, province, educational attainment and age at first sex.Potential confounding factors included in the models are; AIDS-related knowledge and,HIV test status and personal acquaintance with a person who died of AIDS. The followingtable shows the predictor variables, categories and codes.

Independent Variables		
Covariate	Category	Code
	Southern	0
Province	Central	1
	Northern	2
	Primary	0
Level of Education	Secondary	1
	Tertiary	2
	No	0
Ever seen or heard of AIDS information	Yes	1
	No	0
Know anyone who died of AIDS	Yes	1
	No religion/other	0
Religion	Moslem	1
	Christian	2
	10-15 years	0
Current Age	16-20 years	1
	21-24 years	2
	Less than 10 years	0
	11to 15 years	1
Age at First Sex	16-20 years	2
	21-24 years	3
	Refused to answer/missing	4
HIV Test	No	0
	Yes	1

Table 4.9:Predictors: Categories and Codes

4.4 Model 1: The Association between Likelihood of Infection and sociodemographiccharacteristics

A multinomial regression model was fitted to the data in order to examine the direction andstrength of the association between perceived likelihood of HIV infection and socio demographic variables. The following table presents Relative Risk Ratios between the associations of perceived likelihood of infection and the socio economic variables among young men and women in the sample. Thefollowing tables present results of the association between socio demographiccharacteristics and likelihood of HIV Infection among young men and women in 2004.

Likelihood of Infect	ion	No Like	lihood	Medium		Low	
Covariate	Category	RRR	P > z	RRR	P > z	RRR	P > z
Gender	Female(ref)	1		1		1	
	Male	1.920	0.09**	0.765	0.03*	0.573	0.06**
Province	Northern(ref)	1		1		1	
	Southern	0.808	0.79	1.464	0.72	0.808	0.79
	Central	0.964	0.95	1.059	0.94	0.964	0.95
Level of	Primary (ref)	1		1		1	
Education	Secondary	0.393	0.19	0.336	0.20	0.523	0.38
	Tertiary	0.543	0.32	0.643	0.09**	0.722	0.87
Heard/seen of	Yes (ref)	1		1		1	
AIDS info	No	0.543	0.32	0.862	0.02*	0.676	0.54
Know Anyone	Yes (ref)	1		1		1	
who died of AIDS	No	2.757	0.04*	1.046	0.95	3.302	0.08**
Religion	Moslem (ref)	1		1		1	
	No Religion	0.791	0.16	0.771	0.03*	0.157	0.04*
	Christian	0.193	0.37	0.991	0.02*	1.294	0.03*
HIV Test	Yes (ref)	1		1		1	
	No	1.384	0.05**	0.857	0.85	1.048	0.07**
Age of	21-24 years (ref)	1		1		1	
Respondents	10-15 years	0.891	0.86	0.832	0.09**	1.005	0.99
	16-20 years	0.964	0.97	1.267	0.84	0.779	0.71
Age at First Sex	20+ years (ref)	1		1		1	
	Less than 10 years	1.936	0.001*	0.56	0.015*	0.276	0.05**
	11-15 years	1.55	0.01*	0.81	0.08*	0.395	0.06**
	16-20 years	0.814	0.37	1.134	0.27	0.931	0.16
Multinomial logistic rep LR chi2(27) = 81.63 Prob > chi2 = 0.0000 Log likelihood = -533.8 Pseudo R ² = 0.071	gression Number of obs 36349	= 582					

Table 4.10: Relative Risk Ratios Showing the Association between Socio Demographic

 Characteristics and Perceived Likelihood of Infection.

(High likelihood is the base outcome) **Significant at 10% level *Significant at 5% level The 20+ years reference category in the 'age at first sex' variable includes those who had never had sex.

4.4.1 Correlates of No Likelihood Perception

Among those who perceived that there was no likelihood of infection, the following were the results. Gender, knowledge of someone who died of AIDS variables werestatistically

significant factors of no risk perception at 10% level, while religion, HIV teststatus, and age at first were statistically significant at 5% level. The results indicate that gender plays an important role in shaping adolescents' thinkingover the likelihood of infection. Relative to those who perceived high likelihood, maleswere 1.9 times more likely to perceive themselves as having no likelihood of HIV infection(RRR=1.9; p=0.5) compared to females.

Having known someone who died of HIV/AIDS may affect one's perception towards HIV. The results indicate that there is a strong positive association between likelihood of HIV infection and knowledge of knowing someone who died from AIDS (p<0.5). Relative to high risk perceivers, those who had no knowledge of someone who had died of AIDS were more likely (RRR=2.8; p<0.5) to report that they had no likelihood of infection than those who had. This basically implies that those who had knowledge of someone who had died from AIDS were more likely to perceive some risk of HIV infection.

HIV/AIDS testing and counseling potentially play an important role in the construction of these subjective assessments of HIV status. Results indicate that relative to high risk perceivers, those who had no HIV test at the time of the survey had 40% increased odds(Relative Risk Ratio=1.4, P<0.1) of reporting that they was no likelihood of HIV infection compared to those whohad had a test.

Relative to high risk perceivers, adolescents who had their first sexualintercourse before the age of 10 years had about 90% higher odds of reporting no likelihood of infection thanthose who had had sex after 20 years or older (OR=1.9; p<0.05). An increase in the ageresults in a decrease in the likelihood of reporting no risk but in the same direction; where those aged 11-15 years had 55% higher odds of reporting no risk (OR=1.55, p<0.05)compared to those who had sexual debut after 20 years and older ages. The results indicate that afurther increase in the age significantly decreases the likelihood of no risk perception probably because as one grows up they are likely to be better informed about the dangers of HIV and the probability of contracting it if they do not abstain.

Furthermore, relative to high risk perceivers, those who had sex after 15 years were less likely toperceive no risk of infection compared to those who had had sex for the first time at a laterage, although the results are not statistically significant at both 5% and 10% levels(Relative Risk Ratio=0.8; p>0.1). These results could imply that young men and women who had first sex prior toage 15 years had increased odds of perceiving no likelihood of HIV infection, although theassociations were stronger among those age 10 years and less.

It then follows that thosewho had sexual debut at a later stage were less likely to perceive no risk of infection.

Place of residence (province), level of education, hearing or seeing AIDS information, ageof respondents were not significant predictors of HIV risk perception with regards to nolikelihood of infection.

4.4.2 Correlates of Low Likelihood Perception

Generally, the results indicate a weak association between perceived low likelihood of infection andgender, knowledge of someone who died of HIV and having an HIV test, ageat first sex, and religion; results aresignificant at 10% level (p<0.1) except for religion (p<0.05).Relative to high risk perceivers, youngmen had significantly decreased odds of low likelihood perception with regards to HIVinfection compared to young women. There is 0.5 times less likelihood that males perceived low likelihood of infectionthan females (Relative Risk Ratio=0.57; p<0.1). This may imply that females may perceive low likelihoodof HIV infection compared to the young men, which is at least some risk perception, which maybe a catalyst of behavior change assuming the adolescents have comprehensive knowledgeof HIV/AIDS.

Further results reveal an association between religion and low likelihood of infection, at 5% significance level. Relative to high risk perceivers, those who had no religion or who belonged to traditional cults were significantly less likely to report that theyperceived themselves as having low likelihood of infection than their Moslem counterparts(Relative Risk Ratio=0.16; P<0.05). Belonging to Christian religion increased the odds of perceiving low likelihood of infection; adolescents had significantly higher odds of reporting low likelihood of infectionthan their Moslem counterparts (Relative Risk Ratio=1.2; P<0.05). This means that those who belonged to a religion had some feeling of not being safe with regards to HIV infection compared tothose who had no religion who did not even feel even low risk of infection.

A strongassociation exists between knowledgeof someone who died of HIV and perceived lowlikelihood of HIV infection. Relative to high risk perceivers, those who did not have knowledgeof someone who had died of AIDS were more likely (Relative Risk Ratio=3.3; p<0.1) to report thatthey perceived low likelihood of infection than those who knew someone who had died of AIDS. This implies that those who knew someone who had died of AIDS

were more likelyto perceive higher risk of infection than those who did not know any death due to AIDS. These results confirm other studies which state that having a personal ties to someone withHIV/AIDS could make adolescents more cautious in their own risk and protectivebehaviours (Munthali *et al.*, 2004). Furthermore, knowing someone who has died withAIDS also increases the perception of personal risk, presumably because it may increase the illusion of not being safe within one's social circle and hence high likelihood perception(Eaton *et al.*, 2003).

Results also indicate that HIV test is not a strong predictor of low risk perception. Theresults indicate increased the odds of no risk perception among those who had no HIV test(Relative Risk Ratio=1.05; p<0.1) than those who had a test. This implies that those who had an HIV test were more likely to report that they perceived themselves as having some chances of infection; however results should be interpreted with caution since the majority of respondents had not tested for HIV.

Further results indicate that relative to high risk perceivers, there was a strong positiveassociation between age at first sex and low likelihood of infection; the results aresignificant at 5% level and 1% level. The results indicate that those who had very early sex, in this case below 10 years were significantly less likely to report that they perceived lowlikelihood of infection than those who had sexual debut after 20 years and later (Relative Risk Ratio=0.2;p<0.05). An increase in the age does not change the direction of the association butweakens it. The results indicate that relative to high risk perceivers, there are 0.4 timesdecreased odds that those aged 11-15 years would report low likelihood of HIV infectioncompared to those who had sex after 20 years (OR=0.4; P<0.1). While these results point tothe same direction as those who perceived no likelihood at all, a conclusion can be madethat those who had early debut (sex before 15 years) felt at no risk or low risk of HIVinfection compared to those who had sexual debut at later years.

4.4.3 Correlates of Medium Likelihood Perception

With regards to moderate risk perceivers, the trend is a bit different to the 'no risk' and 'low risk' perceivers where age of respondents and having heard or seen AIDS information aresignificant predictors in addition to gender, religion, and age at first sex. Religion, age at first sex, having seen or heard of AIDS information, and gender show astrong positive

association to medium risk perception of HIV; results significant at 5% level, while age of respondents shows a weak positive association with results significant at10% level.

The results indicate that relative to those who perceive high likelihood of infection, youngmen were less likely to perceive medium likelihood of infection (Relative Risk Ratio=0.76; P<0.05)compared to young women. This implies that the young men had less moderate feeling of safetyagainst HIV compared to the females who felt unsafe and considered themselves at moderate risk of HIV infection.

Further results indicate decreased odds of not having seen or heard AIDS information with respect to moderate perception among the respondents (Relative Risk Ratio=0.86; P<0.05). This implies that compared to high risk perceivers, adolescents who had not heard or seen HIV information were less likely to perceive themselves as having moderate HIV risk infection. This means that those who had seen or heard of AIDS information were more likely to perceive medium likelihood of HIV infection, hence the importance of dissemination of AIDS information.

With regards to religion, relative to high risk perceivers, and compared to Moslems there isless likelihood that adolescents who did not belong to any religion or belonged totraditional cultsperceived themselves as having medium risk of contracting HIV(Relative Risk Ratio=0.79; p<0.05). Results also indicate lower odds even among Christians (Relative Risk Ratio=0.99; p<0.05)compared to the Moslems. Conventional knowledgestates that not belonging to a religious group could decrease perception of HIV risk andthat belonging to a religious group increases the likelihood of high risk perception. However, these results should be interpreted with caution since there were fewer responses for themedium likelihood category.

Current age is an important factor in predicting HIV risk perceptions, since the more onegrows in age the more they know about HIV/AIDS and the more they have acquaintancewith people who have died of AIDS. Relative to the high risk perceivers, and compared to those who were aged above 20 years, the results indicate a weak association; lower odds of reporting moderate risk among those who were aged 10-15 years (Relative Risk Ratio=0.8; p<0.1). Although the results that those who were aged 16-20 years were more likely toperceive moderate perception were statistically insignificant (Relative Risk Ratio=1.3 p>0.1), there is ageneral indication that the older the person grows the more likely they self-evaluate themselves as having an HIV risk, which is a catalyst for behavior change.

These results imply that relative to high risk perceivers, those who had early sexual debutwere less likely to perceive medium risk of infection than those who had early debut.

In summary, the above results indicate that among the low risk perceivers and with regardsto high likelihood perceivers, males had higher odds of reporting 'no risk' compared tofemales, had lower likelihood of reporting low and medium likelihood perception than females. Thisimplies that males may feel safer than females with regards to HIV risk and females mayfeel at risk and unsafe from contracting HIV. Adolescents who had no religion were lesslikely to perceive any HIV risk compared to Christians and Moslems who perceivedrisk, although low. Those who did not have knowledge of someone who had died of AIDS were more likely to report that they perceived lowlikelihood of infection than those who knew someone who had died of AIDS. Furthermore, those who had not had an HIV test were more likely to perceive low risk of infection andthose who had early sex (before 15 years) generally perceived no or low likelihood of infection than those who had later sex at later ages (20-24 years).

Relative to high risk perceivers, there is a general indication among no risk perceivers thatmales were more likely to perceive no risk at all compared to the females; those who did not know of someone who died of AIDS were more likely to perceive no risk at all, implying that they felt safer from HIV. This could be attributed to their experience of an AIDS death and its impact within their social circles. Those who did not belong to anyreligious group were more likely to perceive no risk, implying that belonging to a religiousgroup may shape one's thinking towards behavior change whose catalyst could be somefear of contracting HIV, which could be determined by perception of oneself as being atrisk. Furthermore, those who had no HIV test at the time of the survey were more likely to perceive no risk of infection, and those who had early sexual debut were more likely to feelsafe (no likelihood of infection) compared to those who had sexual debut at 20-24 years.

Among the medium risk perceivers, males were less likely to perceive themselves as being at moderate risk compared to females, those who had not heard or seen AIDS informationwere less likely to perceive moderate risk compared to those who had information, non-religious adolescents were less likely to perceive moderate risk compared to Moslems.

Furthermore, age of the respondents was also a predictor of moderate risk where those whowere aged 10-15 years were less likely to perceive moderate risk compared to those aged20-24 years. Lastly those who had first sex before 10 years and between 11-15 years

wereless likely to perceive moderate risk compared to those who had sexual debut at 20-24years and older ages.

Much as some literature points out that education levels and residence (province) mayinfluence perception of HIV risk, there is no such evidence in this study since the resultsare not significant (p>0.05), i.e. province and level of education were not predictors of HIVrisk perceptions in this study.

4.5 Model 2: The Association between Worry about HIV/AIDS Infectionand its Correlates

This second model examines the relationship between worry about contracting HIV/AIDS and the socio economic characteristics. A multinomial logistic regression was used to analyze the relationship between worry about HIV infection and socio demographic characteristics. The following table presents odds ratios between the association of worryabout HIV infection and the socio economic variables among young men and women inMalawi.

Taking the group 'worried a lot as the base outcome', the results were categorized into 2 groups: those who were not worried at all and those who were worried a little. Thefollowing section explains the results in details.

Worry About HIV					
infection		Not Wor	ried at All	Worried	l a little
Covariate	Category	RRR	P > z	RRR	P > z
Gender	Female(ref)	1		1	
	Male	0.917	0.698	1.028	0.921
Province	Northern(ref)	1		1	
	Southern	2.544	0.054**	1.79	0.018*
	Central	3.221	0.021*	2.63	0.007*
Level of Education	Primary (ref)	1		1	
	Secondary	0.499	0.99	1.983	0.001*
	Tertiary	0.889	0.44	1.361	0.987
Heard/seen of	Yes (ref)	1		1	
AIDS info	No	0.986	0.951	0.788	0.399
Know Anyone who	Yes (ref)	1		1	
died of AIDS	No	0.076	0.752	0.076	0.752
Religion**	Moslem (ref)	1		1	
	No Religion	0.477	0.041*	1.885	0.049*
	Christian	0.864	0.003*	0.975	0.95
HIV Test**	Yes (ref)	1		1	
	No	2.62	0.094**	0.606	0.143
Age of Respondents	21-24 years (ref)	1		1	
	10-15 years	0.942	0.842	0.73	0.363
	16-20 years	1.285	0.508	0.834	0.689
Age at First Sex*	21-24 years (ref)	1		1	
	Less than 10 years	1.028	0.001*	1.735	0.035*
	11-15 years	1.325	0.013*	0.497	0.004*
	16-20 years	0.243	0.104**	1.591	0.09**
Multinomial logistic regression LR chi2(27) = 40.3 Prob > chi2 = 0.0019 Log likelihood = -533.86349 Pseudo R ² = 0.063	n Number of obs =581				

Table 4.11: Relative Risk Ratios Showing the Association between Worry about HIV/AIDSInfection and its Correlates.

(Worried a lot is the base outcome). **Significant at 10% level *Significant at 5% level

The results indicate that relative to those who were worried a lot, province and age at first sexwere strong predictors of not being worried at all, results are significant at 1% level (p<0.1), while religion and HIV test were significant at 5% level (p<0.05).

4.5.1 Correlates of 'Not Worried at All'

Compared to those residing in the Northern region, there are 2.5 times increased odds that those in the Southern region perceived themselves as not being worried about contracting HIV/AIDS(Relative Risk Ratio=2.5; p<0.05). Residing in the Central region increased the likelihood of perceiving themselves as not being worried at all (Relative Risk Ratio=3.2; P<0.05). This implies that the Southernerswere better off in terms of having a sense of worry about contracting HIV, assuming that worrycould lead to behavioural change. Controlling for the central or southern region gives the same sults and a conclusion can be made that these results do not explicitly inform on whetherprovince is indeed a predictor of worry about HIV, probably because the respondents were evenly distributed throughout the provinces.

Furthermore, the results indicate that relative to those who perceived a lot of worry, there wasless likelihood that those who had no religion (Relative Risk Ratio=0.48; p<0.05) would perceive themselves not being worried at all about HIV than their Moslem counterparts. Being Christianstrengthens the association (Relative Risk Ratio=0.86; p>0.05) and maintains the same direction; lesslikelihood to perceive no worry at all. Although the association is weak, the implication is that the adolescents were more likely to be somewhat worried about contracting HIV, despite belonging to a religion or not.

A strong association exists between the outcome variable (worry) and age at first sex among young men and women. Relative to those who considered themselves as being worrieda lot, those who had first sex before 10 years significantly increased odds of not worrying at allover HIV infection (Relative Risk Ratio=1.02; p<0.01), compared to those who had late debut (20-24 years).

Results indicate a similar pattern as age group increases to 11- 15 years; 32 percent increasedodds (Relative Risk Ratio=1.32; p<0.01) of not being worried at all about HIV infection than those who debutat a later stage. These results generally imply that those who had first sex in the early stages inlife were more likely not to worry about contracting HIV, hence a conclusion that late debutmay set precedence in being worried over HIV/AIDS among adolescents.

Further results indicate that relative to those who perceived lots of worry, young men andwomen who had had no HIV test were 2.6 times more likely to perceive no worry at all thanthose who had tested for HIV (Relative Risk Ratio=2.6; p<0.1). These results are

significant at 10 percent level. This implies that those who had an HIV test were more likely to perceive some worry about contracting HIV.

In summary, the aboveresults may imply ageat which they had first intercourse, an HIV test; province and religion were significant predictors of not being worried at all about contracting HIV. Gender, level of education, exposure to AIDS knowledge and AIDS death and age of respondents were not significant predictors of not being worried at all.

4.5.2 Worried at a Little

The results indicate that province and level of education were strong predictors of little worry, results are significant at 1% level (p<001), while age at first sex and religion are significant at5% level (p<0.05). These results indicate a similar patternas among thosewho werenot worriedat all. Relative to those who considered themselves as being worried a lot, there were 1.8 times increased odds that those from the Southern region would perceive little worry (Relative Risk Ratio=1.8; p<0.01) and thatthose from the Central region were 2.6 times likely to perceive little worry over contractingHIV (Relative Risk Ratio=2.6; p<0.01). This implies that province is a predictor of worry over HIV and those adolescents were more likely to show some level of worry about contracting HIV.

Education was also a significant predictor of little worry over HIV andthere is evidence of increased odds of adolescents who had attained secondary (high) schooleducation to perceive little worry compared to those that had attained primary school education (Relative Risk Ratio=1.98; p<0.01).Those who had attained tertiary education displayed a similarcharacteristic, though results were not significant. These results imply that those who hadhigher educational attainment were more likely to perceive some worry over HIV infectionthan those with lower educational levels.

Results also indicate that those who had no religion were more likely to perceive little worry over HIV (Relative Risk Ratio=1.9; p<0.05) compared to those who were Moslems. Being Christian was notsignificant although results indicate a reverse trend; less likelihood of perceiving little worry over HIV.

A strong association exists between age at first sex and little worry; thoseadolescents who had sex before age 10 were more likely to have little worry (some worry) overcontracting HIV compared to those who had sexual debut in the 20s (Relative Risk Ratio=1.74; p<0.01). Furthermore, relative to those who perceived lots of worry, adolescents who had sexual debut between ages 11-15 years were less likely to perceive some worry over HIV infection (Relative Risk Ratio=0.5; p<0.05), while those who had debut between 16-20 years were more likely to perceive someworry (Relative Risk Ratio=1.5; p<0.05). These results are very interesting since they show that those aged 11-15 years were less likely to perceive some worry about contracting HIV, results which may be contrary to a lot of literature.

Lastly the results indicate that gender, having seen or heard of AIDSinformation, knowing someone who had died of AIDS, HIV test and age of respondent werenot significant and hence not predictors of little worry over HIV/AIDS among the adolescents.

In summary, the results indicate that relative to those who considered themselves as beingworried a lot, province could be a strong predictor of worry about contracting HIV. This thenimplies that place of residence (province) is more likely to influence young people in the waythey worry about HIV, i.e. it may play a role in setting precedence on adolescents' thinkingabout worry over HIV/AIDS. HIV test plays a role in determining worry levels, those who hadno HIV test were less likely not to worry about contracting HIV. Furthermore, those without areligion were less likely not to worry about HIV compared to those who had a religion.

Withregards to those who perceived little over as compared to those who worried a lot, the results indicate a similar pattern as among those who were not worried at all; compared to those from the Northern region, those from the Southern and Central region perceived little worry. Theresults further indicate that those who had higher educational attainment were more likely toperceive some worry over HIV infection than those with lower educational levels. Results onage at first sex as a predictor of worry about HIV are quite interesting: they show that thoseaged 11-15 years were less likely to perceive little worry, and those who debuted before 10 years were more likely to perceive some worry about contracting HIV, results which may becontrary to a lot of literature which states that those who debut early are more likely to perceivelittle worry.

4.6 Model 3: Association between Condom Use at Last Sex and itsCorrelates

This third model is a binary logistic regression which examines the relationship betweencondom use at last sex and the socio economic characteristics. Condom use at last sex was usedsince this is a more valid measure of condom use in terms of recall of use. Running separatemale and female models produced null results, hence results of the pooled sample has beencomputed. The following table presents results of odds ratios between the association of condom use atlast sex and the socio economic variables among young men and women in Malawi.

Table 4.2:	Odds	Ratios	Showing	the	Association	between	Condom	Use	at	Last	Sex
and its Ass	ociate	d Correl	lates								

Dependent/Outcome Variable (Condom Use)	Category	Relative Risk Ratio (RRR)	P> z
Socio Demographic Characteri	stics		- / 121
	provnort(Northern) (ref)	1	na
Province	provsout(Southern)	1.08	0.895
	provcent (Central)	0.988	0.981
	(primarysc) (primary-ref)	1(ref)	na
Education	educsesco (secondary)	1.498	0.07**
	moslem (ref)	1	na
Religious affiliation	noreligi (No religion)	0.277	0.17
	christia(Christians)	0.36	0.03*
	age 21-24 years	1(ref)	na
Respondent age	age 10-15 years	0.924	0.88
	age 16_20 years	0.695	0.385
	never had sex (ref)	1	na
	agefirst(<10 years)	0.662	0.161
Age at fisrt sex	agefirst(<11-15 years)	0.96	0.074**
	agefirst(16-20 years)	1.109	0.021*
	agefirst (21-24 years)	0	0.001*
Knowledge of AIDS Informatio	n		
	Heard or seen AIDS info (No)	1(ref)	na
Seen/heard AIDS info	HeardseenA~o (yes)	1.278	0.042*
Versee servers diad of diad	Know AIDS death (No)	1(ref)	na
Know anyone died of died	KnowAIDSde~h (Yes)	1.075	0.861
HIN/ Test	HIV Test (No)	1(ref)	na
HIV IEST	HadHIVtest (Yes)	1.036	0.935
Perception of Risk			

Worry Over HIV	worriedl(Worried a lot)	1(ref)	Na
	notworri(Not worried at all)	1.186	0.072**
	worried(Worried a little)	0.907	0.84
	highlikeli(High likelihood)	1(ref)	Na
TH 11 1 6T 6 /·	nolikeli(No likelihood at all)	4.664	0.055**
Likelihood of Infection	lowlikel(Low likelihood)	2.22	0.293
	mediumli(Medium likelihood)	2.259	0.365
Number of obs $=$ 25.	LR chi2(19) = 3	32.67	
Prob > chi2 = 0.0	62 Log likelihood =	132.54358	
Pseudo R2 = $0.$	997		

**Significant at 10% level

*Significant at 5% level

Binary logistic regression was used to examine the relationship between condom use and perception of HIV risk while controlling for other factors that are likely to influence sexual behavior. This done by starting with a model that includes socio demographic variables as the only explanatory variable and then perception of risk and knowledge of AIDS factors were added in a stepwise approach and their effect was observed. Separate models for males and females didnot yield significant results and combining both males and females improved the explanatory power.

The results indicate that relative to those who had used condoms at last sex, there were 4variables that were strongly associated to condom use and significant at 5% level. These areworry over HIV, perceived likelihood of being infected, age at first sex, knowing someone who diedof AIDS, and religious affiliation while educational attainment showed a weak association (resultssignificant at 10% level). Gender and tertiary education variables were dropped from the modeldue to collinearity of variables in the models.

Results also indicate that compared to those who reported no condom use atlast sex, there was 50 percent increased odds of condom use among those who had secondaryschool education (Odds Ratios=1.5; p<0.1) compared to those who had some primary school education. This implies that those who had less educational levels (below secondary school) were lesslikely to use condoms at last sex, which is what most research has found out.

Religion is also a predictor of condom use. Relative to those who reported condom use, adolescents who belonged to Christian religion were more likely to use condoms at last sexcompared to Moslems. Having no religion was not significant both at 5% and 10% levels. These results imply that belonging to Christian religious group increased the likelihood

ofcondom use, and these results may be linked to worry over HIV which shows that those who had noreligion were more likely to perceive no likelihood of infectionat all, a fatalisticperception given the high prevalence of HIV.

Adolescents who perceived 'no likelihood of infection' were less likely to have used condoms at last sex compared to those who perceived high likelihood of infection. This is evident from the decreased odds of condom use at last sex among those who perceived nolikelihood of HIV infection (OddsRatios=0.83; p<0.05).

Furtherresults indicate a strong positive association between reported age at first sex and condom use at last sex; results are significant at 5% level. The results indicate that relative tothose who reported condom use at last sex, adolescents who reported early sexual debut(before 11-15 years) had lower odds of condom use at last sex than those who had never hadsex (Odds Ratios=0.66;P<0.05). An increase in the age at first sex increases the relative strength of the sociation (Odds Ratios=0.9; p<0.5), where those aged between 16-20 years displayed a higher likelihood of condom use at last sex. Those who reported that they had sex at a later stage (20-24 years) in their lives were more likely to use condoms at last sex compared to those who had never had sex (Relative Risk Ratio=1.1; p<0.05). These results imply that relative to those who used condomsat last sex, adolescents who had sexual debut before 15 years were less likely to have used acondom at last sex compared to those who had sex at 20-24 years.

Knowing someone who died of AIDS has been found to be a significant predictor of condom use. Results indicate that adolescents who knew someone who had died of HIV/AIDS were more likely to have used acondom at their last sexual encounter (Odds Ratios=1.2; p<0.05) compared to those who had not experienced an AIDS death.

Worry about HIV may shape adolescents' sexual and protective behaviour. The results indicate a weak association between worry and condom use among the respondents. There were 1.1 times as much likelihood of condom use at last sex among those who reported no worry aboutHIV compared to those who were worried a lot (OddsRatios=1.1; p<0.1). This implies that those whowere not worried about HIV infection were most likely not to use a condom at last sex than thosewho were worried a lot.

Place of residence, likelihood of future infection, likelihood of futurepartner infection, age and hearing about AIDS information were not significant at either 5 % level (p-value>0.05)

as well as at 10% level (p-value>0.1). This implies that these sociodemographic covariates have nothing to do with condom use among adolescents in the future.

CHAPTER 5

DISCUSSION

5.1 Correlates of Risk Perceptions and Protective Strategies

This dissertation set out to examine the association between socio demographic characteristics and self-perceived risk of HIV infection and condom use among youngwomen and men in rural Malawi. The results generally indicate some associations betweenage at first sex which is the variable of interest, and self-perceived risk. Other significant predictors of risk perception and worry are religion, gender, knowledge of someone who died of AIDS, education and religion. The study assumed that background factors atnational, community and individual levels operate through a range of intermediate sociodemographic and psychosocial factors (AIDS information) to influence perception of HIVrisk and condom use. The Social Cognitive Model employed in this study states that among the crucial personal factors are the individual's capabilities to symbolize behavior, to anticipate the outcomes of behavior, to learn by observing others, to have confidence in performing a behavior (including overcoming the problems in performing the behavior), to self-determine or self-regulate behavior, and to reflect on and analyze experience (Bandura, 1997). The results could be related to this model in order to come up with informed decisions about adolescent's sexual behavior with regards to HIV prevention strategies and intervention programmes.

Because reports of sexual behaviour are unreliable. this study measured behavioural characteristics indirectly by analyzing the association between perceived risks includingworry over HIV infection and protective strategies measured by socio demographiccharacteristics including age at first sex as a variable of interest. It is important to understand this relationship because for adolescents to take appropriate measures to protect hemselves against HIV/AIDS, they need to have a correct perception of their risk and correct information on protective strategies. The results of this study are important because for adolescents to take appropriate measures to protect themselves against HIV/AIDS, theyneed to have a high perception of their risk. Consistent with previous research, this studyfound that a majority of the youth in the study perceived themselves as being at little or norisk of HIV infection. Although perceived risk may not be the same with the actual risk, and assuming risk assessment is a prerequisite for behavior change, then the results areworrisome. The following section will discuss the results, starting with the outcome variables and lateron the predictors.

Condom use: Condom use is one of the few interventions known to be effective inreducing the risk of HIV infection. Educational levels, worry about HIV, age at first sexand likelihood of infection emerged significant predictors of condom use among theadolescents. Firstly, males were much less likely than females to report use of condoms atlast sex. Secondly, those who had higher school educational attainment and those whoindicated some worry over contracting HIV were more likely to use condoms. Lastly thosewho perceived themselves at no risk of infection and those who had early sexual debut (before age 15 years) were less likely to have used condoms at last sex. While condom use at last sexmay be a more valid measure of condom use in terms of recall of use, it does not capture consistency of use, an essential element of condom effectiveness (Pettifor *et al.*, 2004). With regards to other protection measures from HIV, descriptive statistics indicate that,females were much more likely than males to feel that abstinence protects one from HIV and were less likely than men to feel that traditional medicine could protect from HIV.

Worry over HIV: Province, religion, level of education, and age at first sex were strongsignificant predictors (p<0.05) of worry levels among adolescents, while HIV test was aweak predictor (p<0.1). There are decreased odds of Christians and those with no religionin not worrying at all, implying they too were worried over contracting HIV compared toMoslems. In this regard it is difficult to conclude whether religious affiliation does set precedence to adolescents thinking with regards to worry. Those with higher school education attainment were less likely to perceive no worry over HIV, probably becausethey were exposed to information and were cautious as not to be indulged in riskybehaviour and probably felt unsafe. An interesting result is that those who had early sexual debut (less than 15 years) weremore likely to worry a little and less likely not to worry at all about contracting HIV, implying that they too had some sense of insecurity about contracting HIV. Consistent withliterature that those who debut early are less likely to worry about contracting HIV, these results also show that they were equally worried as those who had late debut. One may have early sexual intercourse and stop, and lead the rest of their lives in a non-riskybehavior way, while some individuals may have late sexual debut and not stop in so doing exposing themselves to HIV/AIDS. Descriptive statistics indicate that 42.2 percent of therespondents were not worried at all, and males (41 percent) were slightly less

likely to beworried compared to females (43.5 percent). Literature states that whereas men are moreaccurate in assessing their likelihood of infection and display less uncertainty in assessing their HIV status, women who are worried about AIDS infection are significantly morelikely to be infected than women who worry less (Anglewitcz, 2004).

Perceived Likelihood of Infection: Gender, religion and age at first sex were the most significant predictors of likelihood of infection (p<0.05), while AIDS information, knowledge of someone who had died of AIDS, and HIV test were weaker predictors (p<0.1). The results indicate that males were less likely to perceive some likelihood of infection as compared tothe females; which implies that they were more likely to perceive no likelihood of infection. This could be attributed to their higher condom use than the females, but thequestion of consistent condom use lowers their safety levels. Those who knew someonewho had died of AIDS were less likely to perceive no likelihood or low likelihood ofinfection, implying that knowledge gave them a sense of insecurity with regards toinfection. Furthermore, those who did not belong to any religion were less likely toperceive themselves as being at low or medium risk, but were more likely to perceivethemselves at no risk compared to Christians and Moslems. Age at first sex was a strongpredictor of HIV risk perceptions; those adolescents who had early debut (before 15 years)were more likely to perceive low or no likelihood of infection, compared to those whodebuted late (above 20 years). A positive attitude among those who had early sex was thetendency to worry about HIV/AIDS, as explained earlier in the text. Lastly, those who hadno HIV test were more likely to perceive no risk of infection, although they were less likelynot to be worried at all, as explained earlier in the text.

The results in this study indicate the same significant predictors across the three outcomevariables, and in order to avoid repetition, the following section will discuss the correlates.Significant results are shown for the following correlates of risk perceptions, worry and condom use.

Gender: Gender plays a role in shaping adolescents thinking with regards to likelihood of infection and results indicate that males were less likely to perceive low or mediumlikelihood of infection. Instead they were more likely to perceive no likelihood of infection.Males were also more likely to use condoms at last sex compared to females, a results which should be interpreted with caution since condom use may be over reported by
maleswho may want to give the impression of being cautious. Gender in this study is asignificant predictor of worry over HIV.

Age at First Sex: The results generally show that age at first sex is a significant factor withregards to adolescents' perceptions towards HIV as well as condom use, i.e. age at first sexis an important predictor of both HIV risk perceptions and protective strategies. The strong associations indicated in the three models indicate that adolescents who had early debutwere less likely to perceive themselves as having high risk of HIV infection, were less worriedabout contracting HIV and wereless likely to have used condoms at their last sexualencounter compared to those who had sex at 20-24 years. In this study, those who had their first sexual intercourse before the age of 15 years had more than 50% lower odds of reporting no risk or low risk of infectioncompared to those who had first sex at 15 years or older age. These results confirm otherresearch findings done elsewhere.Literature indicates that young people's sexual behaviour suggest that the younger the ageat which first intercourse occurs, the greater the frequency of sexual intercourse and there isincreased likelihood that it will be unprotected and resulting in a higher risk of pregnancy and HIV infection (L'Engle *et al.*, 2006; Koenig *et al.*, 2004; Smith and Watkins, 2005).

Educational Attainment: There is evidence that education may be a predictor of worry about contracting HIV and not of HIV risk perceptions. Those who had a higher educational attainment were more likely to worry about HIV and use condoms at last sexcompared to those who had primary school education attainment. Although there is a thinline between worry and risk perception, these results were not significant for riskperception, hence this dissertation makes a conclusion that education may play a role insetting precedence over worry and condom use among the adolescents.

Knowledge of someone who died of AIDS: Knowing someone who died of AIDS is apredictor of risk perception (low likelihood) and not of condom use and worry. The results indicate that generally, those who knew of someone who died of AIDS were more likely toworry about contracting HIV and felt at risk of infection. This could imply that they would be more cautious in their behavior than those who did not know of someone who died of AIDS. At the time of the survey there was still the tendency of not revealing that someone had died of AIDS even if there was clinical evidence, for fear of stigma. Most HIV related deaths were

sometimes attributed to witchcraft and hence even if mortality was on the risein the study areas they would not be attributed to AIDS.

Hearing and seeing AIDS information: This variable was not a strong predictor of worryand condom use except for moderate likelihood of infection; adolescents who had not heardor seen AIDS information were less likely to perceive themselves at moderate risk. Thismeans that those who had seen or heard of AIDS information were more likely to perceivesome likelihood of infection, hence the importance of dissemination of AIDS information.Hearing or seeing AIDS information is not a predictor of condom use and worry over HIV.These results are worrisome, since there was over 70% knowledge levels of HOVinformation among the adolescents, yet this information seem not to influence theirthinking with regards to protective behavior and worry levels. The insignificance of AIDSinformation as a predictor of both worry and condom confirms the conclusions by otherresearchers that the source of knowledge is not a very good predictor of behaviour andperception of risk (Prohaska *et al.*, 1990, cited in Akwara *et al.*, 2003).

Religion: Religion plays an important role in shaping peoples' sexual behaviour, theconversional wisdom among most Malawians was that those who are more religiousindulge less or not at all in infidelity and are at less risk to HIV/AIDS infection, althoughthis paper does not have evidence for this. Some researchers found out that church attendance was an important determinant of delayed sexual activity, but only when a child's friends attended the same church (Mott et al., 1996). The multivariate results indicate that religion is apredictor of risk perceptions, worry and condom use. Adolescents who did not belong toany religion were less likely to perceive themselves at risk and were less likely to worryabout HIV. These feelings of safety among those who had no religion may be fatalistic in the era of high HIV prevalence rates. Religious people may not necessarily be immune torisky sexual behaviour and may also indulge in religious activities just to cover up theirpast, for instance if they want to get married. Some researchers argue that that churchattendance may elevate the odds of having first sex among adolescents, perhaps because thechurch setting provides an opportunity for these males to meet potential partners (Haas andForste, 2002). In some communities people who have been known to be promiscuous haveturned religious and got married to someone who thought that their likelihood of infectionwas low. The distribution of respondents by religion in Malawi indicates that more thanhalf of all respondents are Christians (86 percent of women and 84 percent of men), while13 percent of women and 12 percent of men are Muslims. Less than 1

109

percent of womenand 3 percent of men reported no religious affiliation (MDGS 2010). Age of respondent: Age of respondents was significant among moderate risk perceivers, and the younger the age the less likely they were to perceive being at risk of infection and the older they were, the more prone they felt of being infected. There is a general indicationthat the older the person grows the more likely they self-evaluate themselves as having anHIV risk, which is a catalyst for behavior change. This may be so due to the fact that as onegrows they are gradually exposed to information on AIDS, are exposed to AIDS deaths knowledge, and may feel the negative impact of HIV if a close relative died of AIDS. Thisfactor was not a predictor of worry and condom use, which means that the age of theadolescents had nothing to do with their levels of worry over HIV and use of condoms atlast sex.

HIV test: HIV test was a predictor of worry and likelihood of infection and of condom use. Those adolescents who had not had an HIV test were less likely to perceive somelikelihood of infection and were less likely to worry about contracting HIV and use acondom at last sex. However, taking into consideration that over 80 percent of the respondents were not tested in the sample, reported high condom use at last sex, the resultsshould be interpreted with caution. HIV testing in Malawi was not popular at the time ofthe survey and although one's HIV status could be a catalyst for behavioural change, thispredictor may not be a good measure for determining risk perceptions and protectivestrategies among adolescents in this study.

CHAPTER 6

CONCLUSION

6.1 Conclusions and Recommendations

Age at first sex is an important indicator of exposure to risk of pregnancy and sexually transmitted infections during adolescence (Zaba et al., 2004). Studies have shown that therisk of HIV infection is low among women who begin sexual activity later (Mc Grath et al., 2009), and that school attendance was significantly associated with later age at first sex. The Measure DHS (2009) report states that age at sexual debut in Africa remains low, with considerable proportions of men and women having first sex before age 15 and that inrecent years, the median age at first sex has remained generally unchanged (Sneeringer andStacy 2009). Analysis of survey data on HIV prevalence in eight countries shows thoseyoung women 15–24 years old that had their sexual debut when they were younger than 15 years old are more likely to be HIV-positive. Some researchers argue that the association between perception of risk of HIV infection and sexual behaviour remains poorlyunderstood, although perception of risk is considered to be the first stage towardsbehavioural change from risk-taking to safer behavior (Akwara et al., 2003). This may betrue, but it should also be noted that most adolescents are not mature enough to view HIVas a risk when they engage in their first sexual encounter and the earlier the youths engagein sexual intercourse the higher the risk of being infected with HIV. While a number ofstudies in sub-Saharan Africa have documented and the relationships between age at firstsex and HIV risk perceptions, this study also adds to such literature. This study has endeavoured to explore socio demographic characteristics and their effects on HIV riskperceptions, worry about contracting HIV and condom use. The overall results in this studyindicate that the odds of sexual debut are significantly related to HIV risk perception andprotection strategies among young people in Malawi, i.e. age at first sex is among one of the crucial factors in shaping adolescent's future perceptions about HIV and the way they would protect themselves from contracting HIV. Other strong predictors that proved significant and crucial in setting precedence in adolescents' thinking about HIV riskperceptions worry and condom use were marital status, religion, knowledge of someonewho died of AIDS, and gender. Educational attainment, having access to AIDS knowledgewere weak predictors of HIV risk perception, worry and condom use at last sex.

Theseresults should however not be conclusive in the absence of the limitations to the study.Focusing on the correlate of interest, this study concludes that age at first sex does setprecedence in adolescents thinking about HIV risk perceptions and condom use. Theseresults confirm results from other studies conducted elsewhere which indicate that youngpeople who debut early are more likely to be exposed to risky behaviour and are unlikelyto use condoms, less worried about contracting HIV and perceived themselves as havinglow risk of HIV infection. This means that those young people who have sex at a later stagemay be more cautious in their sexual behaviour in the future years than those who debutlater. However, this might not be true assuming that those who debut early may have hadcoercive sex, and might have experienced the negative aspect of it than the pleasurableaspect, making them to be more careful and not repeat, or being selective in choosing apartner in future.

Further results reveal that being worried over HIV, likelihood of being infected, and worryabout contracting HIV, age at first sex, and educational attainment were significant predictors of condom use at last sex. The study concludes that those adolescents who perceived low risk of infection, had early sexual debut and had higher school educational attainment were more likely not to use a condom at last sex.

Focusing on the correlate of interest-age at first sex, the results indicate that early sexualdebut leads to risky sexual behaviour probably due to the fact that the young people are lessworried and have a certain sense of safety against HIV and a perception of low risk of contracting HIV. Such young people are less likely to use a condom especially for the firsttime, or may use them inconsistently when they engage in frequent sex. It is worth notingthat individuals may perceive their risk of getting AIDS to be high or low depending ontheir previous sexual behaviour or that of their partners. In this case, risky sexual behavior is the influencing factor on perception of risk. In some cases, a person's perception of riskmay be passive, and not necessarily based on his or her previous sexual behavior (Akwara*et al.*, 2003). Descriptive statistics show that young people generally think that it is notacceptable to use condoms with a spouse, think that there are no chances of contractingHIV/AIDS by having sex once. The perception of HIV/AIDS as being a distant disease, coupled with the problem of drugs and alcohol abuse is perpetuating the risky behavior among the young people. These perceptions are fatalistic in this era of HIV/AIDS wheremost Governments and donors are allocating resources in the fight against AIDS, hence aneed to change the mindset and sexual behaviour

of such adolescents. Results of suchstudies could be useful in designing programme specific intervention goals.

It is worth mentioning that some of the observed associations may be attributable tounmeasured factors (e.g., family structure or economic status) that could play a role inshaping their risk perceptions and protective strategies. Moreover, the cross-sectional nature of the data makes it impossible to establish causality in the observed relationshipsbetween the outcome variable and its correlates (demographic characteristics). This studyrecognizes the problems of using self-reports of sexual behaviour as dependent variables, and that it is highly likely that the proportions reported from the survey data may bedistorted, which in turn could affect the results. Researchers argue that feelings of personal vulnerability are very subjective and vary by context and time (Idele, 2002 cited in Akwara*et al.*, 2003). Thus, sexual behaviour is more likely to be based upon subjective perceptionsof risk rather than actual risk. Furthermore, some researchers also argue that some socio cultural contexts can influence people's perceptions through internalization of specific sexual beliefs, norms and practices and may not necessarily lead to increased perception ofHIV risk if these practices are deemed to be necessary for social support (Akwara *et al.*, 2003.

6.2 **Recommendations and Policy Implication**

Young people in Sub Saharan Africa are at high risk of HIV infection. Because first sexualexperiences may influence a young person's HIV risk, a better understanding of coital debutis needed. Early coital debut is associated with factors that may increase a young person'srisk for HIV infection, such as forced sex and having sex with older partners. Previousresearch in Kenya and elsewhere (Bauni & Jarabi, 2000; Nzioka, 1996; Ingham, Woodcock& Stenner, 1992 cited in Akwara, *et al.*, 2003) suggests that individuals, particularly youngpeople, often feel invulnerable to HIV infection. AIDS may be seen as a distant rather thanan immediate threat: a disease that affects other people (Akwara, *et al.*, 2003).

The AIDS epidemic is complex, and successful efforts to limit transmission must rely on acombination of medical, social and behavioral approaches. Several studies have made itclear that young people should be at the center of strategies to control HIV infection. Thepopulation-based data from this survey provide information that can enable programofficials to add value to the design of youth specific intervention programmes in the fightagainst AIDS. Malawi is one of the countries affected by HIV/AIDS in sub Saharan Africa, coupled with negative reproductive health indicators (MDHS 2010). There is conventionalknowledge among the population that among other socio demographic characteristics of adolescents early age at first sex is associated with risky sexual behaviour in later years, which is what this study has endeavoured to scientifically prove. The results provided in this study can be used as a source of strategic information on HIV-related indicators such as risk perceptions and condom use practice, which in turn can assistprogram managers and policymakers in devising tailored and effective responses to the HIV/AIDS epidemic in sub-Saharan Africa. Furthermore, a clear understanding of adolescent sexual relations is necessary to shape and fine-tune prevention interventions(UNICEF, UNAIDS, WHO and UNFPA, 2009).

The Malawi HIV Prevalence Report (2008) attributes younger ageat sexual debut to havecontributed to a higher HIV prevalence in the southern region (Blantyre) compared to thecentral region. Such conclusive remarks over the difference in HIV prevalence

114

betweenBlantyre City and Lilongwe City may not be valid in the absence of a study like this onesince the report results were merely based on the median age at first sex.

Research findings indicate that for adolescents, engaging in sexual activity may be integral to the dating and courtship process preceding marriage where sexual intercourse itself maybe viewed as a signal of greater intimacy, trust, and commitment (Clark, Poulin and Kohler,2009). Findings in this study that early sexual debut does have an influence in adolescents' thinking with regards to HIV risks as assumed in the hypothesis, may be useful in the design of youth intervention programmes. Instead of directing financial resources towardsprogrammes on IEC (Information, Education and Communication) programmes and lifeskills/orientation in schools where the young people learn about sex related issues, and howthey can protect themselves from HIV or pregnancy, the government should emphasizeprogrammes that advocate increasing age at sexual debut, among the young people.Intervention efforts should encourage youth to delay coital debut and promote strategies tomake young people's first sexual experience safer. Miangotar Yodé (2004) argues that delaying the entry into sexuality could be safer and more secure to protect the sexual andreproductive health in adolescence. Given the risks associated with early debut, preventioncampaigns should continue to encourage youth to delay the onset of first sex and aim toaddress contextual and structural factors that can help make this first experience safer foryoung people (Pettifor, et al., 2009). However it is not effective to deal with interventionprogrammes that deal with sexual debut only, this study recommends holistic approaches indealing with HIV/AIDS among young men and women whereby gender issues, drug abuseand alcohol are integrated in the models with an emphasis on delaying sexual activity..

The results that there is low condom use among adolescents should be of great concern to the government. HIV prevalence rate among adolescents is high, especially among the 20-24 year olds, young girls continue to have sex with older partners in exchange for gifts and money, and teenage pregnancy continues to be on the rise. It is not easy for the government stop adolescents in having sex, as studies show that 'once teenagers start having sex theydo not stop (Akwara *et al.*, 2003). Descriptive results in this study indicate that more young women reported having had sexual intercourse than men, and among those who reported being sexually experience, younger men indicated that they had had sex in the past 12 months compared towomen. It is however possible to advocate for consistent condom use among adolescents sepecially girls, which is being done through life skills in schools and

youth programmesthrough various media channels. While continuing education on the benefits of condom usein reducing risk of HIV, there is need to address cultural and religious factors and sensitivities surrounding condoms and young people in Malawi. A further recommendationis that local barriers to condom availability and use by both young men and women need tobe removed, e.g. condoms are mostly available in shops, spazas or tuckshops and socialplaces such as bars and hotels. It is not easy for young people to buy condoms at suchplaces, introducing them widely in schools could be a solution. Recent studies state thatamong young women who have had sexual intercourse, fewer than half say they could get acondom by themselves (UNAIDS, WHO, UNICEF, UNFPA, 2009). For all this to effective is need for political will, i.e. a strong leadership by governments is essential tomaking this happen, countries should put an effort to support the provision of 'youth-friendly' reproductive health information, education and services rather than largely dependon external donors.

Further recommendation on the MDICP longitudinal study is that the question on 'age atfirst sex' should be maintained in the next waves so that in future a causal relationship canbe established through longitudinal analysis, which this study has not been able toaccomplish, due to lack of data on the variable 'age at first sex' in most of the waves. Thisstudy also recommends that there should be a follow up analysis (for instance the MalawiDemographic and Health Survey for the years 2000, 2004 and 2010) to determine if age atfirst sex does influence HIV risk perceptions or sexual behaviours among young men andwomen, where a causal effect would be established.

References

- 1. Akwara P.A., Nyovani J.M., and Hinde A. 2003. Perception of Risk of HIV/Aids and Sexual Behaviour in Kenya.
- Anderson K.G., Beutel A.M., Maughan-Brown B., 2007. HIV Risk Perceptions andFirst Sexual Intercourse among Youth in Cape Town, South Africa; InternationalFamily Planning Perspectives 33 (3): 98-105.
- Anglewitcz P., Adams J., Obare F., Kohler H.P., Watkins S., 2009. The Malawi Diffusion and Ideational Change Project 2004-06: Data collection, data quality, and analysis of attrition; Demographic Research 20(21): 503-540.
- Anglewicz P.and Kohler H.P.,2005. Overestimating HIV Infection: TheConstruction and Accuracy of Subjective Probabilities of HIV Infection in Rural Malawi.
- Anglewicz, P., Bignami-Van Assche S., Chao Li-Wei, Chilongozi D., Hoffman I.,Kohler H.P., Martinson F., Onyango F., Reniers G., Smith K.P., 2004. HIV/STITesting in the 2004 Malawi Diffusion and Ideational Change Project: Lessons Learned.
- Balanchfiedl L., 2009. The United Nations Convention of theRights of the Child: Background and Policy Issues.
- 7. Bandura, A. 1997. Self-Efficacy: The Exercise of Control.
- 8. Bertrand J.T., & Anhang R., 2006. The effectiveness of mass media in changing HIV/AIDS-related behaviour among young people in developing countries.
- Bignami-Van Assche and G. Guiella. 2009. The Impact of HIVRisk Perception on Adolescents' Sexual Initiation: A Life-Table Analysis Using Panel Data from Rural Malawi.
- Bracher M., Santow G., and Watkins S.C., 2004. Assessing the Potential of CondomUse to Prevent the Spread of HIV: A Microsimulation Study. Studies in Family Planning 35(1): 48–64.

- 11. Biddlecom A.E., Munthali A., Singh S., Woog V. 2007. Adolescents' views of andpreferences for sexual and reproductive health services in Burkina Faso, Ghana, Malawi and Uganda. African Journal of Reproductive Health 11 (3).
- 12. Clark S., Poulin M., Hans-Peter K., 2009. Marital Aspirations, Sexual Behaviors, and HIV/AIDS in Rural Malawi; Journal of Marriage and Family 71: 396–416.
- 13. Clark S., 2004. Early Marriageand HIV Risks in Sub-Saharan Africa. Studies inFamily Planning; 35(3): 149–160.
- Coker AL, Richter D.L., Valois R.F., McKeown R.E, Garrison C.Z., Vincent M.L., 1994. Correlates and consequences of early initiation of sexual intercourse; Journal of School Health 64(9): 372-377.
- 15. Eaton, L., Flisher, A., & Aaro, L., 2003. Unsafe Sexual Behavior in South AfricanYouth; Social Science and Medicine 56(1): 149 165.
- Forste R.and Haas W.D., 2002. Transition of Adolescent Males To First SexualIntercourse: Anticipated or Delayed? Perspectives on Sexual and Reproductive Health 34(4):184-190.
- Gueye M., Castle S., Konate M.K., 2001. Timing of First Intercourse among MalianAdolescents, Implications for Contraceptive Use; International Family Planning Perspectives, 27(2):56-70.
- 18. Gerland P. 2004. Do Social Interactions Affect Individual HIV/AIDS Attitudes and Prevention Strategies in Rural Malawi?
- 19. Global Youth Partners (2009). Global Youth Coalition on HIV/AIDS. www.youthaidscoalition.org.
- 20. Guttmacher Institute. 2006. Adolescents in Malawi, Facts in Brief. Protecting the NextGeneration. <u>www.guttmacher.org</u>.
- Helleringer S. and Hans-Peter Kohler. 2005. Social networks, perceptions of risk, andchanging attitudes towards HIV/AIDS: New evidence from a longitudinal study usingfixed-effects analysis. Population Studies, 59 (3): 265-282.
- 22. Jewkes R, Vundule C, Maforah F, Jordaan E. 2001. Relationship Dynamics and Teenage Pregnancy in South Africa. Social Science & Medicine.; 52(5):733–744.
- 23. Kaestle C.E, HalpernC.T., Miller W.C., and Ford C.A., 2005. Young Age at First Sexual Intercourse and Sexually Transmitted Infections in Adolescents and Young Adults. American Journal of Epidemiology (161):774–780.
- 24. Kim C.C. 2008. Teen Sex: The Parent Factor.

- 25. Koenig, M.A., Zablotska I, Lutalo T, Nalugoda F, Wagman J, Gray R. 2004. Coerced First Intercourse and Reproductive Health Among Adolescent Women in Rakai, Uganda. International Family Planning Perspectives, 30(4):156–163
- 26. L'Engle L., Jackson C., Brown D.J., 2006.Early Adolescents' Susceptibility toInitiating Sexual Intercourse; Perspectives on Sexual and Reproductive Health 38(2):97-105.
- 27. Long J.S .1997. Regression Models for categorical and limited dependent variables.
- 28. Maharaj P., and Cleland J., 2005. Risk Perception and Condom Use among Married orCohabiting Couples in KwaZulu-Natal, South Africa; International Family Planning Perspectives 31(1): 24-29.
- 29. Maharaj P., and Munthree C., 2006. Coerced First Sexual Intercourse and Selected Reproductive Health Outcomes Among Young Women In Kwazulu-Natal, South Africa.Journal of Bioscience 39: 231-244.
- 30. McGrath N., Nyirenda M., Hosegood V., and Newell M.L. 2008. Age at First Sex in Rural South Africa.
- 31. Malawi Government. 2007. Malawi HIV and AIDS Monitoring and Evaluation Report.
- 32. Malawi Government, NAC, USAID, John Hopkins, Save the Children; Malawi BridgeProject, 2004. Exploring Community Beliefs, Attitudes, & Behaviours Related toHIV/AIDS.
- 33. Ministry of Health. 2009. Malawi Health Sector -Wide Approach (SWAp) Programme of Work (PoW) Framework.
- 34. Manning W.D., Longmore A.M, Giordano P.C., 2000. The Relationship Context ofContraceptive Use at First Intercourse; Family Planning Perspective 32: 04-110.
- 35. Manzini N., 2001. Sexual Initiation and Childbearing among Adolescent Girls inKwazulu Natal South Africa; Reproductive Health Matters 9 (27): 44-52.
- 36. Meekers D., 1994. Sexual Initiation and Premarital Childbearing in Sub SaharanAfrica; Population Studies 48(1): 47-64.
- Mensch B.S., Grant M. J., Blanc A.K., 2006. The Changing Context of SexualInitiation in Sub Saharan Africa; Population and Development Review 32 (4): 699-727.
- 38. Mensch, B.S., Hewett, P.C., Gregory, R., and Helleringer, S. 2008. Sexual Behaviorand STI/HIV Status among Adolescents in Rural Malawi: An Evaluation of

the Effectof Interview Mode on Reporting. Studies in Family Planning 39(4): 321-334.

- 39. Miangotar Y. 2004: Association between Age at First Intercourse, Sexual Behaviour and Sexually Transmitted Infections among Adolescents.
- 40. Ministry of Health, Malawi. 2009. Health Information Systems Assessment Report.
- 41. Mishra, Vinod, Agrawal P., Alva S., Yuan Gu, and Wang S. 2009. Changes in HIV-Related Knowledge and Behaviors in Sub-Saharan Africa. DHS Comparative Report 24.
- 42. Mkandawire P. 2009. Evaluating Condom Use among Sexually Active Youth (15-24 years) in Malawi.
- 43. Mott, F.L., Fondell M.M., Hu, P.N., Kowaleski-Jones, L., Menaghan E.G., (1996).Family Planning Perspectives, 28 (1)
- 44. Moore A.M., Awusabo-Asare A., Madise N., John-Langba J.,and Kumi-Kyereme A.,2007. Coerced First Sex among Adolescent Girls in Sub-Saharan Africa: Prevalenceand Context.
- 45. Munthali A.C., Chimbiri A., and Zulu E., 2004. Adolescent Sexual and ReproductiveHealth in Malawi; A Synthesis of Research Evidence, Occasional Report No. 15.
- 46. Munthali A., Zulu, E.M., Madise N., Moore A.M., Konyani S., Kaphuka J., andMaluwa-Banda D., 2006. Adolescent Sexual and Reproductive Health in Malawi:Results from the 2004 National Survey of Adolescents. Occasional Report No. 24.
- 47. National AIDS Commission. 2005. Malawi Behavioural Surveillance Survey.
- Prata N, Morris L, Elizio, Mazive, Farnaz, Vahidnia and Stehr.2006. Relationship between HIV Risk Perception and Condom Use. Evidence from a Population-BasedSurvey in Mozambique. International Family Planning Perspectives, 32(4):192–2003.
- 49. National Statistical Office. 2006. Malawi Demographic and Health Survey (2006).
- 50. National Statistical Office. 2011. Malawi Demographic and Health Survey (2010).
- 51. O'Donnell L., O'Donnell C., Stueve A., 2001. Early Sexual Initiation and SubsequentSex-Related Risks among Urban Minority Youth: The Reach for Health Study; FamilyPlanning Perspective 33:268-275.

- 52. Oyedokun A. and Odimegwu C. 2003. Young People's HIV/AIDS-Related Beliefs,Perception and Sexual Behaviours in South Africa: Analysis of Cape Area Panel Study.
- 53. Population Council Annual Report. 2008. From Evidence to Action. Assessing Need, Achieving Success.
- 54. Pettifor A, O'Brien K, Macphail C, Miller WC, Rees H. 2009. Early coital debut andassociated HIV risk factors among young women and men in South Africa.International Perspective on Sexual Reproductive Health. 35(2):82-90.
- 55. Poulin M. 2010.Reporting On First Sexual Experience: The Importance of Interviewer-Respondent Interaction. Demographic Research. 22 (11):237-288.
- 56. Republic Of Malawi. 2007. Malawi HIV And Aids Monitoring And Evaluation Report;Follow Up To The UN Declaration Of Commitment On HIV And AIDS.
- 57. Rutenberg N., Kaufman C.E., Macintyre K., Brown L., Karim A., 2003. Pregnant orPositive: Adolescent Childbearingand HIVRisk in KwaZulu Natal, South Africa.Reproductive Health Matters 11:122-133.
- 58. Sieving R., Eisenberg M.E., Pettingell S., Skay C., 2006. Friends Influence onAdolescents' First Sexual Intercourse; Perspectives on Sexual and Reproductive Health38(1):13-19.
- 59. Sheppard, Zoe, Madise, Nyovani and Hennink, Monique, 2001. Gender and risk of HIVin Ghana and Uganda.
- 60. Smith P.K., and Watkins S.C., 2005. Perceptions of Risk and Strategies for Prevention:Responses to HIV/AIDS in rural Malawi; Social Science & Medicine 60: 649–660.
- 61. Smith P.K., 2003. Why are they worried? Concern about HIV/AIDS in rural Malawi.Research Article, <u>www.demographic-research.org</u>.
- 62. Sneeringer and Stacy E. 2009. Fertility Transition in Sub-Saharan Africa: AComparative Analysis of Cohort Trends in 30 Countries.
- 63. UNICEF and National Statistical Office. 2006. Monitoring the Situation of Children and Women. 2005. Malawi MultipleIndicator Cluster Survey,
- 64. United Nations, 2008. The Millennium Development Goals Report.
- 65. United Nations 2009. The Millennium Development Goals Report.
- 66. United Nations. 2011. The Millenium Development Goals Reports
- 67. University of Cape Town, Médecins sans Frontières, Institute for Tropical Medicine.

- 68. Exploring HIV Risk Perception in a Context of Antiretroviral Treatment. Resultsfrom a Household Survey.
- 69. UNAIDS, WHO, UNICEF, UNFPA, 2009. Children and AIDS. Fourth StocktakingReport.
- 70. UNFPA, WHO, UNAIDS et al, 2001. IATT on HIV and Young People. HIVInterventions in the Health Sector for Young People.
- 71. UNFPA .2001. Document on HIV/AIDS: Resolution Adopted by the UN General Assembly, A/RES/S-26/2.
- 72. UNAIDS WHO, UNICEF, UNFPA London School of Tropical Hygiene.2008.Preventing Young People from HIV.
- 73. UNFPA. 2009. Preventing HIV/AIDS. Comprehensive Condom Programming: AStrategic Response to HIV and AIDS.
- 74. UNFPA. 2009. Our Voice, Our Future: Young People Report.
- 75. University of Malawi. 2008. Final Report Malawi Prevalence Study.
- 76. Watkins S. C., Zulu E.M., Kohler H.P., Behrman J.R., 2003. Introduction to SocialInteractions and HIV in Rural Africa; Demographic Research, Special Collection 1.
- 77. Weinreb, A., 2006: Description of 2004 MDICP Adolescent Sampling Process (Unpublished).
- 78. WHO. 2004. Risk and Protective Factors Affecting Adolescent Reproductive Health inDeveloping Countries; an analysis of adolescent sexual and reproductive healthliterature from around the world.
- 79. WHO. 2001. Access to Health Services for Young People for Preventing HIV andImproving Sexual and Reproductive Health.
- 80. Zaba, B., Pisani, E., Slaymaker E., and Boerma J.T., 2004. Age at first sex: Understanding Recent Trends In African Demographic Surveys. Sexually Transmitted Infections. <u>www.stijournal.com</u>.